# UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549 FORM 10-O

[X]	QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the Quarterly Period Ended September 30, 2018
[]	TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934
	For the transition period from to
	Commission File No. 1-13726

#### CHESAPEAKE ENERGY CORPORATION

(Exact name of registrant as specified in its charter)

Oklahoma 73-1395733

(State or other jurisdiction of incorporation or organization)

(I.R.S. Employer Identification No.)

6100 North Western Avenue, Oklahoma City, Oklahoma

73118

(Address of principal executive offices)

(Zip Code)

(405) 848-8000

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. YES [X] NO []

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). YES [X] NO []

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company" and "emerging growth company" in Rule 12b-2 of the Exchange Act.

Large Accelerated Filer [X] Accelerated Filer [] Non-accelerated Filer [] Smaller Reporting Company [] Emerging Growth Company []

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. []

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). YES [] NO [X] As of October 25, 2018, there were 913,710,098 shares of our \$0.01 par value common stock outstanding.

#### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES INDEX TO FORM 10-Q FOR THE QUARTER ENDED SEPTEMBER 30, 2018

	PART I. FINANCIAL INFORMATION	Page
Item 1.	Condensed Consolidated Financial Statements	J
	Condensed Consolidated Balance Sheets as of September 30, 2018 and December 31, 2017	<u>3</u>
	<u>Condensed Consolidated Statements of Operations</u> for the Three and Nine Months Ended September 30, 2018 and 2017	<u>5</u>
	<u>Condensed Consolidated Statements of Comprehensive Income</u> for the Three and Nine Months Ended September 30, 2018 and 2017	<u>6</u>
	Condensed Consolidated Statements of Cash Flows for the Nine Months Ended September 30, 2018 and 2017	<u>7</u>
	Condensed Consolidated Statements of Stockholders' Equity for the Nine Months Ended September 30, 2018 and 2017	<u>9</u>
	Notes to the Condensed Consolidated Financial Statements	
	Note 1. Basis of Presentation	<u>10</u>
	Note 2. Earnings per Share	<u>11</u>
	Note 3. Debt	<u>12</u>
	Note 4. Contingencies and Commitments	<u>14</u>
	Note 5. Other Liabilities	<u>15</u>
	Note 6. Income Taxes	<u>16</u>
	Note 7. Share-Based Compensation	<u>17</u>
	Note 8. Derivative and Hedging Activities	<u>20</u>
	Note 9. Oil and Natural Gas Property Transactions	<u>26</u>
	Note 10. Revenue Recognition	<u>28</u>
	Note 11. Investments	<u>30</u>
	Note 12. Impairments	<u>30</u>
	Note 13. Other Operating Expenses	<u>30</u>
	Note 14. Restructuring and Other Termination Costs	<u>30</u>
	Note 15. Fair Value Measurements	<u>31</u>
	Note 16. Condensed Consolidating Financial Information	<u>32</u>
	Note 17. Subsequent Events	<u>41</u>
Item 2.	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>42</u>
	Liquidity and Capital Resources	<u>45</u>
	Results of Operations for the Three and Nine Months Ended September 30, 2018 and 2017	<u>50</u>
Item 3.	Quantitative and Qualitative Disclosure About Market Risk	<u>58</u>
Item 4.	Controls and Procedures	<u>62</u>
	PART II. OTHER INFORMATION	
Item 1.	<u>Legal Proceedings</u>	<u>63</u>
Item 1A.	Risk Factors	<u>63</u>
Item 2.	<u>Unregistered Sales of Equity Securities and Use of Proceeds</u>	<u>66</u>
Item 3.	<u>Defaults Upon Senior Securities</u>	<u>66</u>
Item 4.	Mine Safety Disclosures	<u>66</u>
Item 5.	Other Information	<u>67</u>
Item 6.	<u>Exhibits</u>	<u>68</u>
<b>Signatures</b>		70

#### PART I. FINANCIAL INFORMATION

#### ITEM 1. Condensed Consolidated Financial Statements

### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited)

	Sep	otember 30, 2018	Decen	nber 31, 2017
		(\$ in n	nillions)	
CURRENT ASSETS:				
Cash and cash equivalents (\$1 and \$2 attributable to our VIE)	\$	4	\$	5
Accounts receivable, net		1,051		1,322
Short-term derivative assets		_		27
Other current assets		180		171
Total Current Assets		1,235		1,525
PROPERTY AND EQUIPMENT:				
Oil and natural gas properties, at cost based on full cost accounting:				
Proved oil and natural gas properties (\$488 and \$488 attributable to our VIE)		70,620		68,858
Unproved properties		3,198		3,484
Other property and equipment		1,812		1,986
Total Property and Equipment, at Cost		75,630		74,328
Less: accumulated depreciation, depletion and amortization ((\$463) and (\$461) attributable to our VIE)		(64,500)		(63,664)
Property and equipment held for sale, net		47		16
Total Property and Equipment, Net		11,177		10,680
LONG-TERM ASSETS:				
Other long-term assets		247		220
TOTAL ASSETS	\$	12,659	\$	12,425

### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS – (Continued) (Unaudited)

	Sep	tember 30, 2018	Decei	mber 31, 2017
	_	(\$ in m	nillions)	
CURRENT LIABILITIES:				
Accounts payable	\$	670	\$	654
Current maturities of long-term debt, net		432		52
Accrued interest		126		137
Short-term derivative liabilities		310		58
Other current liabilities (\$2 and \$3 attributable to our VIE)		1,438		1,455
Total Current Liabilities		2,976		2,356
LONG-TERM LIABILITIES:			'	
Long-term debt, net		9,380		9,921
Long-term derivative liabilities		28		4
Asset retirement obligations, net of current portion		154		162
Other long-term liabilities		160		354
Total Long-Term Liabilities		9,722		10,441
CONTINGENCIES AND COMMITMENTS (Note 4)			'	
EQUITY:				
Chesapeake Stockholders' Equity (Deficit):				
Preferred stock, \$0.01 par value, 20,000,000 shares authorized: 5,603,458 shares outstanding		1,671		1,671
Common stock, \$0.01 par value, 2,000,000,000 shares authorized: 913,691,662 and 908,732,809 shares issued		9		9
Additional paid-in capital		14,394		14,437
Accumulated deficit		(16,173)		(16,525)
Accumulated other comprehensive loss		(32)		(57)
Less: treasury stock, at cost; 3,307,953 and 2,240,394 common shares		(31)		(31)
Total Chesapeake Stockholders' Equity (Deficit)		(162)		(496)
Noncontrolling interests		123		124
Total Equity (Deficit)		(39)		(372)
TOTAL LIABILITIES AND EQUITY	\$	12,659	\$	12,425

## CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

		Three Months Ended September 30,			Nine Months September			
		2018		2017		2018		2017
		()	\$ in m	illions exc	ept po	er share da	ta)	
REVENUES:								
Oil, natural gas and NGL	\$	1,199	\$	979	\$	3,424	\$	3,727
Marketing		1,219		964		3,738		3,250
Total Revenues		2,418		1,943		7,162		6,977
OPERATING EXPENSES:								
Oil, natural gas and NGL production		132		151		417		426
Oil, natural gas and NGL gathering, processing and transportation		364		369		1,060		1,081
Production taxes		34		21		91		64
Marketing		1,238		978		3,798		3,333
General and administrative		66		54		229		189
Restructuring and other termination costs		_		_		38		_
Provision for legal contingencies, net		8		20		17		35
Oil, natural gas and NGL depreciation, depletion and amortization		274		228		813		627
Depreciation and amortization of other assets		17		20		54		62
Impairments		5		3		51		3
Other operating (income) expense		_		6		(1)		423
Net (gains) losses on sales of fixed assets		_		(1)		7		_
Total Operating Expenses	_	2,138		1,849		6,574		6,243
INCOME FROM OPERATIONS		280		94		588		734
OTHER INCOME (EXPENSE):							_	
Interest expense		(127)		(114)		(367)		(302
Gains on investments		_		_		139		
Gains (losses) on purchases or exchanges of debt		(68)		(1)		(68)		183
Other income		1		4		63		6
Total Other Expense		(194)		(111)		(233)		(113
INCOME (LOSS) BEFORE INCOME TAXES		86		(17)		355		621
Income tax expense (benefit)		1		_		(8)		2
NET INCOME (LOSS)		85		(17)		363		619
Net income attributable to noncontrolling interests		(1)		(1)		(3)		(3
NET INCOME (LOSS) ATTRIBUTABLE TO CHESAPEAKE		84		(18)	_	360		616
Preferred stock dividends		(23)		(23)		(69)		(62
Loss on exchange of preferred stock		_		_		_		(41
Earnings allocated to participating securities		(1)		_		(3)		(7
NET INCOME (LOSS) AVAILABLE TO COMMON STOCKHOLDERS	\$	60	\$	(41)	\$	288	\$	506
EARNINGS (LOSS) PER COMMON SHARE:	÷		<u> </u>	( /	Ť		Ť	
Basic	\$	0.07	\$	(0.05)	\$	0.32	\$	0.56
Diluted	\$	0.07	\$	(0.05)		0.32	\$	0.56
WEIGHTED AVERAGE COMMON AND COMMON EQUIVALENT SHARES OUTSTANDING (in millions):	Ψ	0.01	Ψ	(0.03)	Ψ	0.32	Ψ	0.50
Basic		910		909		909		908
Diluted		911		909		909		908

### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (Unaudited)

	Three Months Ended September 30,					Nine Mon Septen			
		2018		2017		2018	2017		
				(\$ in m	illior	1s)			
NET INCOME (LOSS)	\$	85	\$	(17)	\$	363	\$	619	
OTHER COMPREHENSIVE INCOME, NET OF INCOME TAX:									
Unrealized gains on derivative instruments <sup>(a)</sup>		_		_		_		4	
Reclassification of losses on settled derivative instruments <sup>(a)</sup>		8		8		25		25	
Other Comprehensive Income		8		8		25		29	
COMPREHENSIVE INCOME (LOSS)		93		(9)		388		648	
COMPREHENSIVE INCOME ATTRIBUTABLE TO NONCONTROLLING INTERESTS		(1)		(1)		(3)		(3)	
COMPREHENSIVE INCOME (LOSS) ATTRIBUTABLE TO CHESAPEAKE	\$	92	\$	(10)	\$	385	\$	645	

<sup>(</sup>a) Deferred tax activity incurred in other comprehensive income was offset by a valuation allowance.

## CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	 Nine Mor Septer		
	 2018		2017
	(\$ in n	nillio	ns)
CASH FLOWS FROM OPERATING ACTIVITIES:			
NET INCOME	\$ 363	\$	619
ADJUSTMENTS TO RECONCILE NET INCOME TO CASH PROVIDED BY (USED IN) OPERATING ACTIVITIES:			
Depreciation, depletion and amortization	867		689
Derivative (gains) losses, net	500		(452)
Cash payments on derivative settlements, net	(162)		(46)
Stock-based compensation	25		38
Net losses on sales of fixed assets	7		_
Impairments	51		3
Gains on investments	(139)		_
(Gains) losses on purchases or exchanges of debt	68		(185)
Other	(101)		(27)
Changes in assets and liabilities	116		(366)
Net Cash Provided By Operating Activities	 1,595		273
CASH FLOWS FROM INVESTING ACTIVITIES:			
Drilling and completion costs	(1,481)		(1,597)
Acquisitions of proved and unproved properties	(244)		(226)
Proceeds from divestitures of proved and unproved properties	395		1,193
Additions to other property and equipment	(11)		(12)
Proceeds from sales of other property and equipment	75		40
Proceeds from sales of investments	74		_
Net Cash Used In Investing Activities	 (1,192)		(602)
CASH FLOWS FROM FINANCING ACTIVITIES:			
Proceeds from revolving credit facility borrowings	9,095		4,775
Payments on revolving credit facility borrowings	(9,231)		(4,130)
Proceeds from issuance of senior notes, net	1,237		742
Extinguishment of other financing	(122)		_
Cash paid to purchase debt	(1,285)		(1,751)
Cash paid for preferred stock dividends	(69)		(160)
Distributions to noncontrolling interest owners	(4)		(7)
Other	(25)		(17)
Net Cash Used In Financing Activities	 (404)		(548)
Net decrease in cash and cash equivalents	(1)		(877)
Cash and cash equivalents, beginning of period	5		882
Cash and cash equivalents, end of period	\$ 4	\$	5

### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS – (Continued) (Unaudited)

Supplemental disclosures to the consolidated statements of cash flows are presented below:

	 Nine Months Ended September 30,			
	2018	2	2017	
	 (\$ in m	illions	)	
SUPPLEMENTAL CASH FLOW INFORMATION:				
Interest paid, net of capitalized interest	\$ 412	\$	342	
Income taxes paid, net of refunds received	\$ (3)	\$	(15)	
SUPPLEMENTAL DISCLOSURE OF SIGNIFICANT NON-CASH INVESTING AND FINANCING ACTIVITIES:				
Change in accrued drilling and completion costs	\$ 165	\$	134	
Change in accrued acquisitions of proved and unproved properties	\$ 1	\$	(1)	
Change in divested proved and unproved properties	\$ (5)	\$	(23)	

## CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF STOCKHOLDERS' EQUITY (Unaudited)

	Nine Mon Septen		
	 2018		2017
	 (\$ in m	illion	s)
PREFERRED STOCK:			
Balance, beginning of period	\$ 1,671	\$	1,771
Exchange/conversions of 0 and 236,048 shares of preferred stock for common stock	_		(100)
Balance, end of period	1,671		1,671
COMMON STOCK:			
Balance, beginning and end of period	9		9
ADDITIONAL PAID-IN CAPITAL:			
Balance, beginning of period	14,437		14,486
Stock-based compensation	26		43
Exchange of preferred stock for 0 and 9,965,835 shares of common stock	_		100
Equity component of contingent convertible notes repurchased, net of tax	_		(20)
Dividends on preferred stock	(69)		(160)
Balance, end of period	14,394		14,449
RETAINED EARNINGS (ACCUMULATED DEFICIT):			
Balance, beginning of period	(16,525)		(17,603)
Net income attributable to Chesapeake	360		616
Cumulative effect of accounting change	(8)		_
Balance, end of period	(16,173)		(16,987)
ACCUMULATED OTHER COMPREHENSIVE LOSS:			
Balance, beginning of period	(57)		(96)
Hedging activity	25		29
Balance, end of period	 (32)		(67)
TREASURY STOCK - COMMON:			
Balance, beginning of period	(31)		(27)
Purchase of 1,499,033 and 1,194,986 shares for company benefit plans	(4)		(7)
Release of 431,474 and 92,015 shares from company benefit plans	4		2
Balance, end of period	 (31)		(32)
TOTAL CHESAPEAKE STOCKHOLDERS' EQUITY (DEFICIT)	(162)		(957)
NONCONTROLLING INTERESTS:			
Balance, beginning of period	124		257
Net income attributable to noncontrolling interests	3		3
Distributions to noncontrolling interest owners	(4)		(7)
Balance, end of period	123		253
TOTAL EQUITY (DEFICIT)	\$ (39)	\$	(704)

#### 1. Basis of Presentation

Basis of Presentation

The accompanying condensed consolidated financial statements of Chesapeake were prepared in accordance with accounting principles generally accepted in the United States of America ("GAAP") and the rules and regulations of the SEC. Pursuant to such rules and regulations, certain disclosures have been condensed or omitted.

This Form 10-Q relates to the three and nine months ended September 30, 2018 (the "Current Quarter" and the "Current Period", respectively) and the three and nine months ended September 30, 2017 (the "Prior Quarter" and the "Prior Period", respectively). Our annual report on Form 10-K for the year ended December 31, 2017 ("2017 Form 10-K") should be read in conjunction with this Form 10-Q. The accompanying condensed consolidated financial statements reflect all normal recurring adjustments which, in the opinion of management, are necessary for a fair statement of our condensed consolidated financial statements and accompanying notes and include the accounts of our direct and indirect wholly owned subsidiaries and entities in which we have a controlling financial interest. Intercompany accounts and balances have been eliminated.

Recently Issued Accounting Standards

The Financial Accounting Standards Board (FASB) issued *Revenue from Contracts with Customers* (Topic 606) superseding virtually all existing revenue recognition guidance. We adopted this new standard in the first quarter of 2018 using the modified retrospective approach. We applied the new standard to all contracts that were not completed as of January 1, 2018 and reflected the aggregate effect of all modifications in determining and allocating the transaction price. See Note 10 for further details regarding our adoption of Topic 606.

In February 2018, the FASB issued Accounting Standards Update (ASU) 2018-02, *Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income*. The new standard allows for stranded tax effects resulting from tax reform legislation known as the Tax Cuts and Jobs Act of 2017 (the "Tax Act") previously recognized in accumulated other comprehensive income to be reclassified to retained earnings. For public business entities, the amendments are effective for annual periods, including interim periods within the annual periods, beginning after December 15, 2018. Early adoption is permitted in any interim or annual period, but we do not plan to early adopt. We are currently evaluating the impact of this standard on our consolidated financial statements and related disclosures.

In August 2017, the FASB issued ASU 2017-12, *Derivatives and Hedging (Topic 815)*, which makes significant changes to the current hedge accounting guidance. The new standard eliminates the requirement to separately measure and report hedge ineffectiveness and generally requires the entire change in the fair value of a hedging instrument to be presented in the same income statement line as the hedged item. The new standard also eases certain documentation and assessment requirements and modifies the accounting for components excluded from the assessment of hedge effectiveness. The new standard update is effective for annual and interim periods beginning after December 15, 2018, including interim periods within those annual periods. Early adoption is permitted, but we do not plan to early adopt. We are currently evaluating the impact of this standard on our consolidated financial statements and related disclosures.

In February 2016, the FASB issued ASU 2016-02, *Leases (Topic 842)*, which updated lease accounting guidance requiring lessees to recognize most leases, including operating leases, on the balance sheet as a right-of-use asset and lease liability for leases with terms in excess of 12 months. In January 2018, the FASB issued ASU 2018-01 permitting an entity to elect an optional transition practical expedient to not evaluate land easements that existed or expired before the adoption of Topic 842 and were not previously accounted for as leases. In July 2018, the FASB issued ASU 2018-11 to provide an additional transition practical expedient by allowing entities to initially apply the new leases standard at the adoption date and recognize a cumulative-effect adjustment to the opening balance of retained earnings in the period of adoption. We plan to elect both practical expedients. We plan to adopt the new standard on January 1, 2019 by recognizing a cumulative-effect adjustment to the opening balance of retained earnings. The standard will not apply to our leases of mineral rights. Using the revised framework, we have completed our assessment of lease categories that we believe will be affected by the new standard. We are continuing to assess the accounting treatment for these leases but do not expect the adoption to have significant impacts to our consolidated financial statements or related disclosures.

#### 2. Earnings Per Share

Basic earnings per share (EPS) is calculated using the weighted average number of common shares outstanding during the period and includes the effect of any participating securities as appropriate. Participating securities consist of unvested restricted stock issued to our employees and non-employee directors that provide dividend rights.

Diluted EPS is calculated assuming the issuance of common shares for all potentially dilutive securities, provided the effect is not antidilutive. For all periods presented, our contingent convertible senior notes did not have a dilutive effect and, therefore, were excluded from the calculation of diluted EPS. See Note 3 for further discussion of our convertible senior notes and contingent convertible senior notes.

A reconciliation of basic EPS and diluted EPS is as follows:

	Three Months Ended September 30,						nths Ended mber 30,	
		2018		2017	2018			2017
			(in mill	ions, exce	pt per	share dat	ta)	
Net income (loss) available to common stockholders	\$	60	\$	(41)	\$	288	\$	506
Effect of dilutive securities		_		_		_		_
Diluted income (loss) available to common stockholders	\$	60	\$	(41)	\$	288	\$	506
Weighted average common and common equivalent shares outstanding -								
basic		910		909		909		908
Effect of dilutive securities		1		_		_		_
Weighted average common and common equivalent shares outstanding - diluted		911		909		909		908
Net income per share attributable to Chesapeake:								
Basic	\$	0.07	\$	(0.05)	\$	0.32	\$	0.56
Diluted	\$	0.07	\$	(0.05)	\$	0.32	\$	0.56
Shares of common stock for the following securities were excluded from the calculation of diluted EPS as the effect was antidilutive:								
Common stock equivalent of our preferred stock outstanding		60		60		60		60
Common stock equivalent of our convertible senior notes outstanding		146		146		146		146
Participating securities		2		_		1		1

#### 3. Debt

Our long-term debt consisted of the following as of September 30, 2018 and December 31, 2017:

	September 30, 2018					Decembe	per 31, 2017		
		Principal Carrying Amount Amount			Principal Amount		Carrying Amount		
7.25% senior notes due 2018	\$	44	\$	44	\$	44	\$	44	
Floating rate senior notes due 2019		380		380		380		380	
6.625% senior notes due 2020		437		437		437		437	
6.875% senior notes due 2020		227		227		227		227	
6.125% senior notes due 2021		548		548		548		548	
5.375% senior notes due 2021		267		267		267		267	
4.875% senior notes due 2022		451		451		451		451	
8.00% senior secured second lien notes due 2022 <sup>(a)</sup>		1,416		1,823		1,416		1,895	
5.75% senior notes due 2023		338		338		338		338	
7.00% senior notes due 2024		850		850		_		_	
8.00% senior notes due 2025		1,300		1,290		1,300		1,290	
5.5% convertible senior notes due 2026(b)(c)		1,250		859		1,250		837	
7.5% senior notes due 2026		400		400		_		_	
8.00% senior notes due 2027		1,300		1,298		1,300		1,298	
2.25% contingent convertible senior notes due 2038 <sup>(b)</sup>	)	9		8		9		8	
Term loan due 2021		_		_		1,233		1,233	
Revolving credit facility		645		645		781		781	
Debt issuance costs		_		(54)		_		(63)	
Interest rate derivatives				1		<u> </u>		2	
Total debt, net		9,862		9,812		9,981		9,973	
Less current maturities of long-term debt, net <sup>(d)</sup>		(433)		(432)		(53)		(52)	
Total long-term debt, net	\$	9,429	\$	9,380	\$	9,928	\$	9,921	

- (a) On October 29, 2018, we delivered a notice of redemption to the trustee with respect to 100% of the aggregate principal amount of the outstanding senior secured second lien notes dues 2022.
- (b) We are required to account for the liability and equity components of our convertible debt instruments separately and to reflect interest expense through the first demand repurchase date, as applicable, at the interest rate of similar nonconvertible debt at the time of issuance. The applicable rates for our 2.25% Contingent Convertible Senior Notes due 2038 and our 5.5% Convertible Senior Notes due 2026 are 8.0% and 11.5%, respectively.
- (c) Prior to maturity under certain circumstances and at the holder's option, the notes are convertible. During the Current Quarter, the price of our common stock was below the threshold level for conversion and, as a result, the holders do not have the option to convert their notes in the fourth quarter of 2018.
- (d) As of September 30, 2018, net current maturities of long-term debt includes our 7.25% Senior Notes due December 2018, our Floating Rate Senior Notes due April 2019, and due to the holders' put option, our 2.25% Contingent Convertible Notes due December 2038.

#### Debt Issuances and Retirements

In the Current Quarter, we issued at par \$850 million of 7.00% Senior Notes due 2024 ("2024 notes") and \$400 million of 7.50% Senior Notes due 2026 ("2026 notes") pursuant to a public offering for net proceeds of \$1.230 billion. We used the net proceeds from the senior notes, together with cash on hand and borrowings under our revolving credit facility, to repay in full \$1.233 billion of borrowings under our secured term loan due 2021 for \$1.285 billion, which included a \$52 million call premium. We recorded a loss of approximately \$65 million associated with the repayment of the term loan, including the call premium and the write-off of \$13 million of associated deferred charges.

We may redeem some or all of the 2024 notes at any time prior to April 1, 2021 and some or all of the 2026 notes at any time prior to October 1, 2021, in each case at a price equal to 100% of the principal amount of the notes to be redeemed plus a "make-whole" premium. At any time prior to April 1, 2021, with respect to the 2024 notes, and October 1, 2021, with respect to the 2026 notes, we also may redeem up to 35% of the aggregate principal amount of each series of notes with an amount of cash not greater than the net cash proceeds of certain equity offerings at a specified redemption price. In addition, we may redeem some or all of the 2024 notes at any time on or after April 1, 2021 and some or all of the 2026 notes at any time on or after October 1, 2021, in each case at the redemption prices in accordance with the terms of the notes and the indenture and supplemental indenture governing the notes. These senior notes are unsecured obligations of Chesapeake and rank equally in right of payment with all of our other existing and future senior unsecured indebtedness and rank senior in right of payment to all of our future subordinated indebtedness. Our obligations under the senior notes are jointly and severally, fully and unconditionally guaranteed by certain of our direct and indirect wholly owned subsidiaries.

In the Prior Period, we retired \$1.609 billion principal amount of our outstanding senior notes, senior secured second lien notes and contingent convertible notes through purchases in the open market, tender offers or repayment upon maturity for \$1.751 billion. For the open market repurchases and tender offers, we recorded an aggregate net loss of approximately \$1 million in the Prior Quarter and an aggregate gain of approximately \$183 million in the Prior Period including \$260 million of premium associated with our 8.00% Senior Secured Second Lien Notes due 2022.

#### Revolving Credit Facility

On September 12, 2018, we amended and restated our credit agreement dated December 15, 2014. The amended and restated revolving credit facility matures in September 2023 and the aggregate initial commitment of the lenders and borrowing base under the facility is \$3.0 billion. The revolving credit facility provides for an accordion feature, pursuant to which the aggregate commitments thereunder may be increased to up to \$4.0 billion from time to time, subject to agreement of the participating lenders and certain other customary conditions. Borrowing base redeterminations will continue to occur semiannually and our next borrowing base redetermination is scheduled for the second quarter of 2019. As of September 30, 2018, we had outstanding borrowings of \$645 million under the revolving credit facility and had used \$182 million of the revolving credit facility for various letters of credit. We recorded a loss of \$3 million associated with certain deferred charges related to the revolving credit facility prior to this amendment.

Borrowings under the revolving credit facility bear interest at an alternative base rate (ABR) or LIBOR, at our election, plus an applicable margin ranging from 0.50%-2.00% per annum for ABR loans and 1.50%-3.00% per annum for LIBOR loans, depending on the percentage of the borrowing base then being utilized and whether our leverage ratio exceeds 4.00 to 1.

Our revolving credit facility is subject to various financial and other covenants. The terms of the revolving credit facility include covenants limiting, among other things, our ability to incur additional indebtedness, make investments or loans, incur liens, consummate mergers and similar fundamental changes, make restricted payments, make investments in unrestricted subsidiaries and enter into transactions with affiliates. Our revolving credit facility contains financial covenants that, after a transition period and the suspension of most of the covenants during the fourth quarter of 2018 as a result of the closing of the sale of certain of our Utica Shale properties pursuant to our purchase and sale agreement with EAP Ohio, LLC ("Encino"), requires the Company to maintain (i) a leverage ratio of not more than 5.50 to 1 through the fiscal quarter ending September 30, 2019, which threshold decreases over time to 4.00 to 1 for the fiscal quarter ending March 31, 2021 and each fiscal quarter thereafter, (ii) a secured leverage ratio of not more than 2.50 to 1 until the later of (x) the fiscal quarter ending March 31, 2021 or (y) the fiscal quarter in when the Company's leverage ratio does not exceed 4.00 to 1 and (iii) a fixed charge coverage ratio of not less than 2.00 to 1 through the fiscal quarter ending December 31, 2019; not less than 2.25 to 1 through the fiscal quarter ending June 30, 2020; and not less than 2.50 to 1 for the fiscal quarter ended September 30, 2020 and thereafter. For the Current Quarter, we

were subject to the financial covenants applicable prior to the amended and restated revolving credit facility in addition to maintaining a leverage ratio of not more than 5.50 to 1.

As of September 30, 2018, we were in compliance with all applicable financial covenants under the credit agreement and we were able to borrow up to the full availability under the revolving credit facility.

Fair Value of Debt

We estimate the fair value of our senior notes based on the market value of our publicly traded debt as determined based on the yield of our senior notes (Level 1). The fair value of all other debt is based on a market approach using estimates provided by an independent investment financial data services firm (Level 2). Fair value is compared to the carrying value, excluding the impact of interest rate derivatives, in the table below:

	Septemb	), 2018		Decembe	er 31, 2017		
	Carrying Amount		Estimated Fair Value	Carrying Amount			Estimated Fair Value
			_				
Short-term debt (Level 1)	\$ 432	\$	433	\$	52	\$	53
Long-term debt (Level 1)	\$ 3,495	\$	3,546	\$	2,633	\$	2,629
Long-term debt (Level 2)	\$ 5,884	\$	6,010	\$	7,286	\$	7,301

#### 4. Contingencies and Commitments

There have been no material developments in previously reported legal or environmental contingencies or commitments other than the items discussed below. For a discussion of commitments and contingencies, see "Contingencies and Commitments," Note 4 to the Consolidated Financial Statements in our 2017 Form 10-K.

#### Contingencies

Regulatory and Related Proceedings. We have previously disclosed receiving U.S. Postal Service and state subpoenas seeking information on our royalty payment practices. The U.S. Postal Service inquiry and all such outstanding state subpoenas have been resolved.

We have also previously disclosed defending lawsuits alleging various violations of the Sherman Antitrust Act and state antitrust laws. In 2016, putative class action lawsuits were filed in the U.S. District Court for the Western District of Oklahoma and in Oklahoma state courts, and an individual lawsuit was filed in the U.S. District Court of Kansas, in each case against us and other defendants. The lawsuits generally allege that, since 2007 and continuing through April 2013, the defendants conspired to rig bids and depress the market for the purchases of oil and natural gas leasehold interests and properties in the Anadarko Basin containing producing oil and natural gas wells. The lawsuits seek damages, attorney's fees, costs and interest, as well as enjoinment from adopting practices or plans that would restrain competition in a similar manner as alleged in the lawsuits. On April 12, 2018, we reached a tentative settlement to resolve substantially all Oklahoma civil class action antitrust cases for an immaterial amount.

On July 28, 2017, OOGC America LLC (OOGC) filed a demand for arbitration with the American Arbitration Association against Chesapeake Exploration, L.L.C., our wholly owned subsidiary, in connection with OOGC's purchase of certain oil and gas leases and other assets pursuant to a Purchase and Sale Agreement entered into on October 10, 2010. In connection with the sale, we also entered into a Development Agreement with OOGC, dated November 15, 2010 (the "Development Agreement"), which governs each of our rights and obligations with respect to the sale, including the transportation and marketing of oil and gas. OOGC's breach of contract, breach of agency and fiduciary duties and other claims generally allege, among other things, that we subjected OOGC to excessive rates for gathering and other services provided for under the Development Agreement and interfered with OOGC's right to audit the documents that supported those rates. OOGC seeks relief that may be material, including unspecified damages, attorneys' fees, costs and expenses, disgorgement and various declaratory judgments. We intend to vigorously defend these claims.

On July 24, 2018, Healthcare of Ontario Pension Plan (HOOPP) filed a demand for arbitration with the American Arbitration Association regarding HOOPP's purchase of our interest in Chaparral Energy, Inc. stock for \$215 million on January 5, 2014. HOOPP claims that the Company engaged in material misrepresentations and fraud, and that we violated the Exchange Act and Oklahoma Uniform Securities Act. HOOPP seeks either rescission or \$215 million in monetary damages, and in either case, interest, attorney's fees, disgorgement and punitive damages. We intend to vigorously defend these claims.

#### Commitments

Gathering, Processing and Transportation Agreements

We have contractual commitments with midstream service companies and pipeline carriers for future gathering, processing and transportation of oil, natural gas and NGL to move certain of our production to market. Working interest owners and royalty interest owners, where appropriate, will be responsible for their proportionate share of these costs. Commitments related to gathering, processing and transportation agreements are not recorded as obligations in the accompanying consolidated balance sheets; however, they are reflected in our estimates of proved reserves.

The aggregate undiscounted commitments under our gathering, processing and transportation agreements, excluding any reimbursement from working interest and royalty interest owners, credits for third-party volumes or future costs under cost-of-service agreements, are presented below:

	(\$ in millions)
2018	\$ 269
2019	1,048
2020	992
2021	900
2022	792
2023 – 2035	4,443
Total	\$ 8,444

In addition, we have entered into long-term agreements for certain natural gas gathering and related services within specified acreage dedication areas in exchange for cost-of-service based fees redetermined annually, or tiered fees based on volumes delivered relative to scheduled volumes. Future gathering fees may vary with the applicable agreement.

#### 5. Other Liabilities

Other current liabilities as of September 30, 2018 and December 31, 2017 are detailed below:

	September 30, 2018	Dec	cember 31, 2017
	(\$ in r		
Revenues and royalties due others	\$ 535	\$	612
Accrued drilling and production costs	313		216
Joint interest prepayments received	81		74
Accrued compensation and benefits	192		214
Other accrued taxes	123		43
Other	194		296
Total other current liabilities	\$ 1,438	\$	1,455

Other long-term liabilities as of September 30, 2018 and December 31, 2017 are detailed below:

	•	ember 30, 2018		nber 31, 017
	_	(\$ in millions)		
CHK Utica ORRI conveyance obligation <sup>(a)</sup>	\$	_	\$	156
Unrecognized tax benefits		53		101
Other		107		97
Total other long-term liabilities	\$	160	\$	354

<sup>(</sup>a) In the Current Period, we repurchased previously conveyed overriding royalty interests (ORRI) from the CHK Utica, L.L.C. investors and extinguished our obligation to convey future ORRIs to the CHK Utica, L.L.C. investors for combined consideration of \$199 million. The total CHK Utica ORRI conveyance obligation extinguished in the Current Period was \$183 million, of which, \$30 million was recorded in current liabilities and \$153 million was recorded in long-term liabilities. The fair value of the consideration allocated to the extinguishment of liability, \$122 million, was less than the carrying amount of the conveyance obligation and resulted in a gain of \$61 million recognized in other income on our condensed consolidated statement of operations. The fair value of the consideration allocated to the purchase of ORRIs on proved producing properties was \$77 million and recorded in proved oil and natural gas properties in our condensed consolidated balance sheet.

#### 6. Income Taxes

We estimate our annual effective tax rate for continuing operations in recording our quarterly income tax provision (or benefit) for the various jurisdictions in which we operate. The tax effects of statutory rate changes, significant unusual or infrequent items, and certain changes in the assessment of the realizability of deferred tax assets are excluded from the determination of our estimated annual effective tax rate as such items are recognized as discrete items in the quarter in which they occur.

For the Current Quarter, our estimated annual effective tax rate remains nominal as a result of having a full valuation allowance against our net deferred tax asset. Taking into account our projected operating results for the subsequent 2018 quarter, we project remaining in a net deferred tax asset position as of December 31, 2018. Based on all available positive and negative evidence, including estimates of future taxable income, we believe it is more-likely-than-not that these deferred tax assets will not be realized. A significant piece of objectively verifiable negative evidence evaluated is the cumulative loss incurred over the rolling three-year period ending September 30, 2018. Such evidence limits our ability to consider various forms of subjective positive evidence, such as our projections for future growth and earnings. A valuation allowance was recorded against substantially all of our net deferred tax asset as of December 31, 2017 and against all of our net deferred tax asset as of September 30, 2018.

We are subject to U.S. federal income tax as well as income and capital taxes in various state jurisdictions in which we operate. We recorded a \$1 million income tax expense in the Current Quarter and an \$8 million income tax benefit in the Current Period. The \$1 million expense in the Current Quarter was a result of discrete items related to additional state income tax expense for the settlement of a state income tax audit and the filing of amended state income tax returns. The \$8 million benefit in the Current Period was a result of discrete items consisting of a \$13 million reduction to the liability for state unrecognized tax benefits due to the expiration of applicable statutes of limitations which was partially offset by eliminating a deferred tax asset for alternative minimum tax carryforwards in the amount of \$3 million and recording additional state income tax expense of \$2 million relating primarily to the settlement of a state income tax audit and the filing of amended state income tax returns. A further reduction to the liability for state unrecognized tax benefits was also recorded against interest expense in the amount of \$4 million.

On December 22, 2017, the President of the United States signed into law the Tax Act, which substantially revised numerous areas of U.S. federal income tax law, including reducing the tax rate for corporations from a maximum rate of 35% to a flat rate of 21% and eliminating the corporate alternative minimum tax (AMT). The various estimates included in determining our tax provision as of December 31, 2017 remain provisional through the nine months ended September 30, 2018 and may be adjusted through subsequent events such as the filing of the 2017 consolidated federal income tax return and the issuance of additional guidance such as new Treasury Regulations. Moreover, we are still in the process of evaluating the full impact of the Tax Act both at the federal and state level.

#### 7. Share-Based Compensation

Our share-based compensation program consists of restricted stock, stock options, performance share units (PSUs) and cash restricted stock units (CRSUs) granted to employees and restricted stock granted to non-employee directors under our long term incentive plans. The restricted stock and stock options are equity-classified awards and the PSUs and CRSUs are liability-classified awards.

Equity-Classified Awards

Restricted Stock. We grant restricted stock units to employees and non-employee directors. A summary of the changes in unvested restricted stock during the Current Period is presented below:

	Shares of Unvested Restricted Stock	Ğr	ted Average ant Date lue Per Share
	(in thousands)		
Unvested restricted stock as of January 1, 2018	13,178	\$	6.37
Granted	5,776	\$	3.77
Vested	(5,782)	\$	7.67
Forfeited	(1,376)	\$	6.09
Unvested restricted stock as of September 30, 2018	11,796	\$	4.49

The aggregate intrinsic value of restricted stock that vested during the Current Period was approximately \$20 million based on the stock price at the time of vesting.

As of September 30, 2018, there was approximately \$38 million of total unrecognized compensation expense related to unvested restricted stock. The expense is expected to be recognized over a weighted average period of approximately 2.12 years.

Stock Options. In the Current Period and the Prior Period, we granted members of management stock options that vest ratably over a three-year period. Each stock option award has an exercise price equal to the closing price of our common stock on the grant date. Outstanding options expire seven years to ten years from the date of grant.

We utilize the Black-Scholes option pricing model to measure the fair value of stock options. The expected life of an option is determined using the simplified method. Volatility assumptions are estimated based on the average historical volatility of Chesapeake stock over the expected life of an option. The risk-free interest rate is based on the U.S. Treasury rate in effect at the time of the grant over the expected life of the option. The dividend yield is based on an annual dividend yield, taking into account our dividend policy, over the expected life of the option. We used the following weighted average assumptions to estimate the grant date fair value of the stock options granted in the Current Period:

Expected option life – years	6.0
Volatility	63.55%
Risk-free interest rate	2.72%
Dividend yield	—%

The following table provides information related to stock option activity in the Current Period:

	Number of Shares Underlying Options	Weighted Average Exercise Price Per Share		Average Av Exercise Price Contr		Weighted Average Contract Life in Years	Α	aggregate Intrinsic Value <sup>(a)</sup>
	(in thousands)				(\$ in millions)			
Outstanding as of January 1, 2018	16,285	\$	8.25	7.73	\$	1		
Granted	3,611	\$	3.01					
Exercised	_	\$	_		\$	_		
Expired	(602)	\$	13.83					
Forfeited	(1,067)	\$	5.45					
Outstanding as of September 30, 2018	18,227	\$	7.19	7.44	\$	8		
Exercisable as of September 30, 2018	8,250	\$	10.73	6.05	\$	2		

<sup>(</sup>a) The intrinsic value of a stock option is the amount by which the current market value or the market value upon exercise of the underlying stock exceeds the exercise price of the option.

As of September 30, 2018, there was \$16 million of total unrecognized compensation expense related to stock options. The expense is expected to be recognized over a weighted average period of approximately 1.74 years.

Restricted Stock and Stock Option Compensation. We recognized the following compensation costs related to restricted stock and stock options for the Current Quarter, the Prior Quarter, the Current Period and the Prior Period:

	Three Months Ended September 30,			Nine Months Ended September 30,				
	2018 2017			2018			2017	
				(\$ in n	nillions	)		
General and administrative expenses	\$	6	\$	9	\$	21	\$	20
Oil and natural gas properties		1		3		5		7
Oil, natural gas and NGL production expenses		1		2		4		7
Total restricted stock and stock option compensation	\$	8	\$	14	\$	30	\$	34

Liability-Classified Awards

Performance Share Units. We granted PSUs to senior management that vest ratably over a three-year performance period and are settled in cash. The ultimate amount earned is based on achievement of performance metrics established by the Compensation Committee of the Board of Directors. Compensation expense associated with PSU awards is recognized over the service period based on the graded-vesting method. The value of the PSU awards at the end of each reporting period is dependent upon our estimates of the underlying performance measures.

For PSUs granted in 2017 and 2016, performance metrics include a total shareholder return (TSR) component, which can range from 0% to 100% and an operational performance component based on finding and development costs, which can range from 0% to 100%, resulting in a maximum payout of 200%. The payout percentage for the 2016 and 2017 PSU awards is capped at 100% if our absolute TSR is less than zero. The PSUs are settled in cash on the third anniversary of the awards. We utilized a Monte Carlo simulation for the TSR performance measure and the following assumptions to determine the grant date fair value of the 2017 and 2016 PSU awards.

#### **Grant Date Assumptions**

Assumption	2017 Awards	2016 Awards
Volatility	80.65%	49.74%
Risk-free interest rate	1.54%	1.13%
Dividend yield for value of awards	<u> </u>	—%

#### **Reporting Period Assumptions**

Assumption	2017 Awards	2016 Awards
Volatility	51.65%	41.46%
Risk-free interest rate	2.65%	2.19%
Dividend yield for value of awards	—%	—%

As the above assumptions change, the PSU liabilities will be adjusted quarterly through the end of the performance period.

For PSUs granted in 2018, performance metrics include an operational performance component based on a ratio of cumulative earnings before interest expense, income taxes, and depreciation, depletion and amortization expense (EBITDA) to capital expenditures, for which payout can range from 0% to 200%. The vested PSUs are settled in cash on each of the three annual vesting dates. We used the closing price of our common stock on the grant date to determine the grant date fair value of the PSUs. The PSU liability will be adjusted quarterly, based on changes in our stock price and expected satisfaction of performance metrics, through the end of the performance period.

Cash Restricted Stock Units. In the Current Period, we granted CRSUs to employees that vest straight-line over a three-year period and are settled in cash on each of the three annual vesting dates. The ultimate amount earned is based on the closing price of our common stock on each of the vesting dates. We used the closing price of our common stock on the grant date to determine the grant date fair value of the CRSUs. The CRSU liability will be adjusted quarterly, based on changes in our stock price, through the end of the performance period. The CRSUs are subsequently adjusted, based on changes in our stock price through the end of each subsequent reporting period, through the end of each vesting period.

The following table presents a summary of our liability-classified awards:

		Grant Date Fair Value			Septembe	er 30, 20	18
	Units			Fai	r Value	Ves	sted Liability
		(\$ in m	illions)	(\$ in millions)			
2018 PSU Awards:							
Payable 2019, 2020 and 2021	3,992,358	\$	12	\$	18	\$	
2017 PSU Awards:							
Payable 2020	1,217,774	\$	8	\$	6	\$	4
2016 PSU Awards:							
Payable 2019	2,348,893	\$	10	\$	11	\$	10
2018 CRSU Awards:							
Payable 2019, 2020 and 2021	16,034,295	\$	48	\$	72	\$	_

#### 8. Derivative and Hedging Activities

We use derivative instruments to reduce our exposure to fluctuations in future commodity prices and to protect our expected operating cash flow against significant market movements or volatility. All of our oil, natural gas and NGL derivative instruments are net settled based on the difference between the fixed-price payment and the floating-price payment, resulting in a net amount due to or from the counterparty. None of our open oil, natural gas and NGL derivative instruments were designated for hedge accounting as of September 30, 2018 or December 31, 2017.

#### Oil, Natural Gas and NGL Derivatives

As of September 30, 2018 and December 31, 2017, our oil, natural gas and NGL derivative instruments consisted of the following types of instruments:

- Swaps: We receive a fixed price and pay a floating market price to the counterparty for the hedged commodity. In exchange for higher fixed prices on certain of our swap trades, we may sell call options and call swaptions.
- *Options*: We sell, and occasionally buy, call options in exchange for a premium. At the time of settlement, if the market price exceeds the fixed price of the call option, we pay the counterparty the excess on sold call options and we receive the excess on bought call options. If the market price settles below the fixed price of the call option, no payment is due from either party.
- Call Swaptions: We sell call swaptions to counterparties that allow the counterparty, on a specific date, to extend an existing fixed-price swap for a certain period of time.
- Collars: These instruments contain a fixed floor price (put) and ceiling price (call). If the market price exceeds the call strike price or
  falls below the put strike price, we receive the fixed price and pay the market price. If the market price is between the put and the call
  strike prices, no payments are due from either party. Three-way collars include the sale by us of an additional put option in exchange
  for a more favorable strike price on the call option. This eliminates the counterparty's downside exposure below the second put
  option strike price.
- Basis Protection Swaps: These instruments are arrangements that guarantee a fixed price differential to NYMEX from a specified delivery point. We receive the fixed price differential and pay the floating market price differential to the counterparty for the hedged commodity.

The estimated fair values of our oil, natural gas and NGL derivative instrument assets (liabilities) as of September 30, 2018 and December 31, 2017 are provided below:

	Septem	ber 30, 203	18	December 31, 2017			
	Notional Volume	Fa	air Value	Notional Volume	Fa	air Value	
		(\$ in	millions)		(\$ in millions)		
Oil (mmbbl):							
Fixed-price swaps	21	\$	(294)	21	\$	(151)	
Three-way collars	_		(8)	2		(10)	
Call swaptions	_		_	2		(13)	
Basis protection swaps	8		2	11		(9)	
Total oil	29		(300)	36		(183)	
Natural gas (bcf):							
Fixed-price swaps	400	(a)	(19)	532		149	
Three-way collars	88		3	_		_	
Collars	66		_	47		11	
Call options	60		_	110		(3)	
Basis protection swaps	44		(7)	65		(7)	
Total natural gas	658		(23)	754		150	
NGL (mmgal):							
Fixed-price swaps	57		(15)	33		(2)	
Total estimated fair value		\$	(338)		\$	(35)	

a) Includes 170 bcf related to trades executed in accordance with the purchase and sale agreement with Encino. These trades are reflected at fair market value as of September 30, 2018, with an offsetting receivable balance. The trades were novated to Encino upon closing of the purchase and sale agreement on October 29, 2018.

We have terminated certain commodity derivative contracts that were previously designated as cash flow hedges for which the original contract months are yet to occur. See further discussion below under *Effect of Derivative Instruments – Accumulated Other Comprehensive Income (Loss)*.

Effect of Derivative Instruments - Condensed Consolidated Balance Sheets

The following table presents the fair value and location of each classification of derivative instrument included in the condensed consolidated balance sheets as of September 30, 2018 and December 31, 2017 on a gross basis and after same-counterparty netting:

Balance Sheet Classification		Gross Fair Value		Amounts Netted in the Consolidated Balance Sheets		Net Fair Value Presented in the Consolidated Balance Sheet
				(\$ in millions)		
As of September 30, 2018						
Commodity Contracts:						
Short-term derivative asset	\$	14	\$	(14)	\$	_
Long-term derivative asset		3		(3)		_
Short-term derivative liability		(324)		14		(310)
Long-term derivative liability		(31)		3		(28)
Total derivatives	\$	(338)	\$		\$	(338)
As of December 31, 2017						
Commodity Contracts:						
· ·	ф	157	ф	(120)	ф	27
Short-term derivative asset	\$	157	\$	(130)	\$	27
Short-term derivative liability		(188)		130		(58)
Long-term derivative liability		(4)				(4)
Total derivatives	\$	(35)	\$		\$	(35)

Effect of Derivative Instruments – Condensed Consolidated Statements of Operations

The components of oil, natural gas and NGL revenues for the Current Quarter, the Prior Quarter, the Current Period and the Prior Period are presented below:

	Three Months Ended September 30,					nded 30,		
	2018		2017		2017 2			2017
Oil, natural gas and NGL revenues	\$	1,331	\$	1,049	\$	3,924	\$	3,275
Gains (losses) on undesignated oil, natural gas and NGL derivatives		(124)		(62)		(475)		477
Losses on terminated cash flow hedges		(8)		(8)		(25)		(25)
Total oil, natural gas and NGL revenues	\$	1,199	\$	979	\$	3,424	\$	3,727

Effect of Derivative Instruments - Accumulated Other Comprehensive Income (Loss)

A reconciliation of the changes in accumulated other comprehensive income (loss) in our consolidated statements of stockholders' equity related to our cash flow hedges is presented below:

	Three Months Ended September 30, 2018 2017								
	 2018				201	7			
	fore ax		fter Tax		efore Tax	After Tax			
			(\$ in m	illions	s)				
Balance, beginning of period	\$ (97)	\$	(40)	\$	(132)	(75)			
Losses reclassified to income	8		8		8	8			
Balance, end of period	\$ (89)	\$	(32)		(124)	(67)			

	Nine Months Ended September 30,								
		2018				20	017		
	Before Tax		After Tax					After Tax	
	(\$ in n					5)			
Balance, beginning of period	\$	(114)	\$	(57)	\$	(153)	\$	(96)	
Net change in fair value		_		_		4		4	
Losses reclassified to income		25		25		25		25	
Balance, end of period	\$	(89)	\$	(32)	\$	(124)	\$	(67)	

The accumulated other comprehensive loss as of September 30, 2018 represents the net deferred loss associated with commodity derivative contracts that were previously designated as cash flow hedges for which the original contract months are yet to occur. Remaining deferred gain or loss amounts will be recognized in earnings in the month for which the original contract months are to occur. As of September 30, 2018, we expect to transfer approximately \$35 million of net loss included in accumulated other comprehensive income to net income (loss) during the next 12 months. The remaining amounts will be transferred by December 31, 2022.

#### Credit Risk Considerations

Our derivative instruments expose us to our counterparties' credit risk. To mitigate this risk, we enter into derivative contracts only with counterparties that are highly rated or deemed by us to have acceptable credit strength and deemed by management to be competent and competitive market-makers, and we attempt to limit our exposure to non-performance by any single counterparty. As of September 30, 2018, our oil, natural gas and NGL derivative instruments were spread among 11 counterparties.

#### Hedging Arrangements

Certain of our hedging arrangements are with counterparties that are also lenders (or affiliates of lenders) under our revolving credit facility. The contracts entered into with these counterparties are secured by the same collateral that secures our revolving credit facility. In addition, we enter into bilateral hedging agreements with other counterparties. The counterparties' and our obligations under the bilateral hedging agreements must be secured by cash or letters of credit to the extent that any mark-to-market amounts owed to us or by us exceed defined thresholds. As of September 30, 2018, we posted \$14 million in letters of credit as collateral for our commodity derivatives. No cash was posted as collateral for our commodity derivatives.

#### Fair Value

The fair value of our derivatives is based on third-party pricing models which utilize inputs that are either readily available in the public market, such as oil, natural gas and NGL forward curves and discount rates, or can be corroborated from active markets or broker quotes. These values are compared to the values given by our counterparties for reasonableness. Since oil, natural gas and NGL swaps do not include optionality and therefore generally have no unobservable inputs, they are classified as Level 2. All other derivatives have some level of unobservable input, such as volatility curves, and are therefore classified as Level 3. Derivatives are also subject to the risk that either party to a contract will be unable to meet its obligations. We factor non-performance risk into the valuation of our derivatives using current published credit default swap rates. To date, this has not had a material impact on the values of our derivatives.

The following table provides information for financial assets (liabilities) measured at fair value on a recurring basis as of September 30, 2018 and December 31, 2017:

	_	Quoted Prices in Active Markets (Level 1)		Prices in Other Active Observabl Markets Inputs (Level 1) (Level 2)		Observable Unobservable Inputs Inputs		 Total Fair Value
As of September 30, 2018				<b>,</b>		,		
Derivative Assets (Liabilities):								
Commodity assets	\$	_	\$	7	\$	11	\$ 18	
Commodity liabilities		_		(340)		(16)	(356)	
Total derivatives	\$	_	\$	(333)	\$	(5)	\$ (338)	
As of December 31, 2017								
Derivative Assets (Liabilities):								
Commodity assets	\$	_	\$	_	\$	8	\$ 8	
Commodity liabilities		_		(20)		(23)	(43)	
Total derivatives	\$	_	\$	(20)	\$	(15)	\$ (35)	

A summary of the changes in the fair values of our financial assets (liabilities) classified as Level 3 during the Current Period and the Prior Period is presented below:

	Comm Deriva	•
	(\$ in mi	illions)
Balance, as of January 1, 2018	\$	(15)
Total gains (losses) (realized/unrealized):		
Included in earnings <sup>(a)</sup>		(3)
Total purchases, issuances, sales and settlements:		
Settlements		13
Balance, as of September 30, 2018	\$	(5)
Balance, as of January 1, 2017	\$	(10)
Total gains (losses) (realized/unrealized):		
Included in earnings <sup>(a)</sup>		1
Total purchases, issuances, sales and settlements:		
Settlements		1
Balance, as of September 30, 2017	\$	(8)

(a)

	Comm	odity Deri	ivatives
	2018		2017
	<b>(</b> \$	in millior	ns)
Total gains (losses) included in earnings for the period	\$	(3) \$	1
Change in unrealized gains (losses) related to assets still held at reporting date	\$	(3) \$	(7)

Qualitative and Quantitative Disclosures about Unobservable Inputs for Level 3 Fair Value Measurements

The significant unobservable inputs for Level 3 derivative contracts include market volatility. Changes in market volatility impacts the fair value measurement of our derivative contracts, which is based on an estimate derived from option models. For example, an increase or decrease in the forward prices and volatility of oil and natural gas prices decreases or increases the fair value of oil and natural gas derivatives. The following table presents quantitative information about Level 3 inputs used in the fair value measurement of our commodity derivative contracts at fair value as of September 30, 2018:

Instrument Type	Unobservable Input	Range	Weighted Average	Septem	ir Value ber 30, 2018 millions)
Oil trades	Oil price volatility curves	19.09% - 28.60%	24.97%	\$	(8)
Natural gas trades	Natural gas price volatility curves	15.60% - 62.08%	16.24%	\$	3

#### 9. Oil and Natural Gas Property Transactions

Under full cost accounting rules, we account for sales of oil and natural gas properties as adjustments to capitalized costs, with no recognition of gain or loss unless a sale involves a significant change in proved reserves and significantly alters the relationship between capitalized costs and proved reserves.

On October 29, 2018 we sold all of our approximately 1,500,000 gross (900,000 net) acres in Ohio, of which approximately 320,000 net acres are prospective for the Utica Shale with approximately 920 producing wells, along with related property and equipment (collectively, the "Designated Properties") for net proceeds of \$1.868 billion to Encino, with additional contingent payments to us of up to \$100 million comprised of \$50 million in consideration in each case if, on or prior to December 31, 2019, there is a period of twenty (20) trading days out of a period of thirty (30) consecutive trading days where (i) the average of the NYMEX natural gas strip prices for the months comprising the year 2022 equals or exceeds \$3.00/mmbtu as calculated pursuant to the purchase agreement, and (ii) the average of the NYMEX natural gas price strip prices for the months comprising the year 2023 equals or exceeds \$3.25/mmbtu as calculated pursuant to the purchase agreement.

We expect the sale of our Designated Properties to Encino to involve a significant change in proved reserves and to significantly alter the relationship between costs and proved reserves and therefore to result in the recognition of loss upon closing of that transaction. Under SEC rules for full cost companies, a transaction is deemed to be significant if the properties being sold represent 25% or more of the reserve quantities of the divesting company.

In the Current Period, we sold portions of our acreage, producing properties and other related property and equipment in the Mid-Continent, including our Mississippian Lime assets, for approximately \$491 million, subject to certain customary closing adjustments. Included in the sales were approximately 238,500 net acres and interests in approximately 3,200 wells. Also, in the Current Quarter and the Current Period, we received proceeds of approximately \$8 million and \$31 million, respectively, subject to customary closing adjustments, for the sale of other oil and natural gas properties covering various operating areas.

In the Prior Period, we sold portions of our acreage and producing properties in our Haynesville Shale operating area in northern Louisiana for approximately \$915 million, subject to certain customary closing adjustments. Included in the sales were approximately 119,500 net acres and interests in 576 wells that were producing approximately 80 mmcf of gas per day at the time of closing. Also in the Prior Quarter and the Prior Period, we received proceeds of approximately \$248 million and \$331 million, respectively, net of post-closing adjustments, for the sale of other oil and natural gas properties covering various operating areas.

#### Volumetric Production Payments

A VPP is a limited-term overriding royalty interest in oil and natural gas reserves that (i) entitles the purchaser to receive scheduled production volumes over a period of time from specific lease interests; (ii) is free and clear of all associated future production costs and capital expenditures; (iii) is non-recourse to the seller (i.e., the purchaser's only recourse is to the reserves acquired); (iv) transfers title of the reserves to the purchaser; and (v) allows the seller to retain all production beyond the specified volumes, if any, after the scheduled production volumes have been delivered. If contractually scheduled volumes exceed the actual volumes produced from the VPP wellbores that are attributable to the ORRI conveyed, either the shortfall will be made up from future production from these wellbores (or, at our option, from our retained interest in the wellbores) through an adjustment mechanism, or the initial term of the VPP will be extended until all scheduled volumes, to the extent produced, are delivered from the VPP wellbores to the VPP buyer. We retain drilling rights on the properties below currently producing intervals and outside of producing wellbores.

As the operator of the properties from which the VPP volumes have been sold, we bear the cost of producing the reserves attributable to these interests, which we include as a component of production expenses and production taxes in our consolidated statements of operations in the periods these costs are incurred. As with all non-expense-bearing royalty interests, volumes conveyed in a VPP transaction are excluded from our estimated proved reserves; however, the estimated production expenses and taxes associated with VPP volumes expected to be delivered in future periods are included as a reduction of the future net cash flows attributable to our proved reserves for purposes of determining our full cost ceiling test for impairment purposes and in determining our standardized measure. Our commitment to bear the costs on any future production of VPP volumes is not reflected as a liability on our balance sheet. Future costs will depend on the actual production volumes as well as the production costs and taxes in effect during the periods in which the production actually occurs, which could differ materially from our current and historical costs, and production may not occur at the times or in the quantities projected, or at all.

We have committed to purchase natural gas and liquids associated with our VPP transactions. Production purchased under these arrangements is based on market prices at the time of production, and the purchased natural gas and liquids are resold at market prices.

As of September 30, 2018, we had the following VPP outstanding:

					Volume Sold							
VPP						Natural						
#	Date of VPP	Location	Pr	oceeds	Oil	Gas	NGL	Total				
			(\$ in	millions)	(mmbbl)	(bcf)	(mmbbl)	(bcfe)				
9	May 2011	Mid-Continent	\$	853	1.7	138	4.8	177				

The volumes remaining to be delivered on behalf of our VPP buyers as of September 30, 2018 were as follows:

		•	Volume Remaining as of	) (mmbbl)	
VPP#	VPP # Term Remaining Oil		Natural Gas	NGL	Total
	(in months)	(in months) (mmbbl)		(mmbbl)	(bcfe)
9	29	0.3	25.8	0.7	31.4

#### 10. Revenue Recognition

The FASB issued *Revenue from Contracts with Customers* (Topic 606) superseding virtually all existing revenue recognition guidance. We adopted this new standard in the first quarter of 2018 using the modified retrospective approach. We applied the new standard to all contracts that were not completed as of January 1, 2018 and reflected the aggregate effect of all modifications in determining and allocating the transaction price. The cumulative effect of adoption of \$8 million did not have a material impact on our condensed consolidated financial statements. However, the adoption did result in certain purchase and sale contracts being recorded on a net basis, as an agent, rather than on a gross basis, as principal, due to management's evaluation under new considerations within Topic 606 that indicated we do not have control over the specified commodity in purchase and sale contracts with the same counterparty. Such presentation change did not have an impact on income (loss) from operations, earnings per share or cash flows.

In accordance with the new revenue standard requirements, the disclosure of the impact of adoption on our condensed consolidated statements of operations was as follows:

		Before adoption of ASC 606 Adjustments		As Reported	
Statement of Operations for the Three Months Ended September 30, 20	018				
Marketing revenues	\$	1,508	\$	(289)	\$ 1,219
Marketing operating expenses	\$	1,516	\$	(278)	\$ 1,238
Statement of Operations for the Nine Months Ended September 30, 203	L8				
Marketing revenues	\$	4,320	\$	(582)	\$ 3,738
Marketing operating expenses	\$	4,365	\$	(567)	\$ 3,798

Revenue from the sale of oil, natural gas and NGL is recognized upon the transfer of control of the products, which is typically when the products are delivered to customers. Revenue is recognized net of royalties due to third parties in an amount that reflects the consideration we expect to receive in exchange for those products.

Revenue from contracts with customers includes the sale of our oil, natural gas and NGL production (recorded as oil, natural gas and NGL revenues in the condensed consolidated statements of operations) as well as the sale of certain of our joint interest holders' production which we purchase under joint operating arrangements (recorded in marketing revenues in the condensed consolidated statements of operations). In connection with the marketing of these products, we obtain control of the oil, natural gas and NGL we purchase from other interest owners at defined delivery points and deliver the product to third parties, at which time revenues are recorded.

Payment terms and conditions vary by contract type, although terms generally include a requirement of payment within 30 days. There are no significant judgments that significantly affect the amount or timing of revenue from contracts with customers.

We also earn revenue from other sources, including from a variety of derivative and hedging activities to reduce our exposure to fluctuations in future commodity prices and to protect our expected operating cash flow against significant market movements or volatility, (recorded within oil, natural gas and NGL revenues in the condensed consolidated statements of operations) as well as a variety of oil, natural gas and NGL purchase and sale contracts with third parties for various commercial purposes, including credit risk mitigation and satisfaction of our pipeline delivery commitments (recorded within marketing revenues in the condensed consolidated statements of operations).

In circumstances where we act as an agent rather than a principal, our results of operations related to oil, natural gas and NGL marketing activities are presented on a net basis. These purchase and sales contracts were accounted for as derivatives under *Derivatives and Hedging* (Topic 815) and were not elected as normal purchase or normal sales. We considered the principal versus agent guidance in Topic 606 in determining whether the gains and losses on these derivatives should be reported on a gross or net basis.

The following table shows revenue disaggregated by operating area and product type, for the Current Quarter and the Current Period:

	Three Months Ended September 30, 2018								
	Oil	Natur	al Gas		NGL		Total		
		(\$ in millions)							
Marcellus	\$ _	\$	184	\$	_	\$	184		
Haynesville	_		195		_		195		
Eagle Ford	399		36		58		493		
Utica	59		131		76		266		
Mid-Continent	58		15		12		85		
Powder River Basin	78		17		13		108		
Revenue from contracts with customers	594		578		159		1,331		
Losses on oil, natural gas and NGL derivatives	(100)		(18)		(14)		(132)		
Oil, natural gas and NGL revenue	\$ 494	\$	560	\$	145	\$	1,199		
Marketing revenue from contracts with customers	\$ 707	\$	211	\$	112	\$	1,030		
Other marketing revenue	119		70		_		189		
Marketing revenue	\$ 826	\$	281	\$	112	\$	1,219		

	Nine Months Ended September 30, 2018							
		Oil	Natural Gas			NGL		Total
	(\$ in millions)							
Marcellus	\$	_	\$	646	\$	_	\$	646
Haynesville		2		603		_		605
Eagle Ford		1,148		120		143		1,411
Utica		179		350		189		718
Mid-Continent		196		63		42		301
Powder River Basin		173		40		30		243
Revenue from contracts with customers		1,698		1,822		404		3,924
Losses on oil, natural gas and NGL derivatives		(388)		(85)		(27)		(500)
Oil, natural gas and NGL revenue	\$	1,310	\$	1,737	\$	377	\$	3,424
			-					
Marketing revenue from contracts with customers	\$	2,125	\$	733	\$	324	\$	3,182
Other marketing revenue		381		175		_		556
Marketing revenue	\$	2,506	\$	908	\$	324	\$	3,738

#### Accounts Receivable

Our accounts receivable are primarily from purchasers of oil, natural gas and NGL and from exploration and production companies that own interests in properties we operate. This industry concentration could affect our overall exposure to credit risk, either positively or negatively, because our purchasers and joint working interest owners may be similarly affected by changes in economic, industry or other conditions. We monitor the creditworthiness of all our counterparties and we generally require letters of credit or parent guarantees for receivables from parties deemed to have sub-standard credit, unless the credit risk can otherwise be mitigated. We utilize an allowance method in accounting for bad debt based on historical trends in addition to specifically identifying receivables that we believe may be uncollectible. Accounts receivable as of September 30, 2018 and December 31, 2017 are detailed below:

	•	ember 30, 2018	December 31, 2017					
	(\$ in millions)							
Oil, natural gas and NGL sales	\$	829	\$	959				
Joint interest		160		209				
Other		78		184				
Allowance for doubtful accounts		(16)		(30)				
Total accounts receivable, net	\$	1,051	\$	1,322				

#### 11. Investments

In the Current Period, FTS International, Inc. (NYSE: FTSI) completed an initial public offering. Due to the offering, the ownership percentage of our equity method investment in FTSI decreased from approximately 29% to 24% and resulted in a gain of \$78 million. In addition, we sold approximately 4.3 million shares of FTSI in the offering for net proceeds of approximately \$74 million and recognized a gain of \$61 million decreasing our ownership percentage to approximately 20%. We continue to hold approximately 22.0 million shares in the publicly traded company.

#### 12. Impairments

In the Current Period, we have determined that certain of our other fixed assets will either be sold or disposed before the end of their useful lives indicating the carrying value may not be recoverable. As a result, we recognized an impairment loss of \$51 million in the Current Period for the difference between the carrying amount and fair value of the assets.

#### 13. Other Operating Expenses

In the Prior Period, we terminated future natural gas transportation commitments related to divested assets for cash payments of \$126 million. In the Prior Period, we paid \$290 million to assign an oil transportation agreement to a third party.

#### 14. Restructuring and Other Termination Costs

#### Workforce Reduction

On January 30, 2018, we underwent a reduction in workforce impacting approximately 13% of employees across all functions, primarily on our Oklahoma City campus. In connection with the reduction, we incurred a total charge in the Current Period of approximately \$38 million for one-time termination benefits. The following table summarizes our restructuring liabilities:

		er Current abilities
	(\$ in	millions)
Balance as of December 31, 2017	\$	_
Initial restructuring recognition on January 30, 2018		38
Termination benefits paid		(38)
Balance as of September 30, 2018	\$	

#### 15. Fair Value Measurements

Recurring Fair Value Measurements

Other Current Assets. Assets related to our deferred compensation plan are included in other current assets. The fair value of these assets is determined using quoted market prices, as they consist of exchange-traded securities.

Other Current Liabilities. Liabilities related to our deferred compensation plan are included in other current liabilities. The fair values of these liabilities are determined using quoted market prices, as the plan consists of exchange-traded mutual funds.

Financial Assets (Liabilities). The following table provides fair value measurement information for the above-noted financial assets (liabilities) measured at fair value on a recurring basis as of September 30, 2018 and December 31, 2017:

		Quoted Prices in Active Markets (Level 1)	Significant Other Observable Inputs (Level 2)		Significant Unobservable Inputs (Level 3)	Total Fair Value		
			(\$ in n	nillio	ons)			
As of September 30, 2018								
Financial Assets (Liabilities):								
Other current assets	\$	54	\$ _	\$	_	\$	54	
Other current liabilities		(54)	_		_		(54)	
Total	\$	_	\$ _	\$	_	\$	_	
As of December 31, 2017								
Financial Assets (Liabilities):								
Other current assets	\$	57	\$ _	\$	_	\$	57	
Other current liabilities		(60)	_				(60)	
Total	\$	(3)	\$ _	\$	_	\$	(3)	

See Note 3 for information regarding fair value measurement of our debt instruments. See Note 8 for information regarding fair value measurement of our derivatives.

TABLE OF CONTENTS

### CHESAPEAKE ENERGY CORPORATION AND SUBSIDIARIES NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS – (Continued) (Unaudited)

#### 16. Condensed Consolidating Financial Information

Chesapeake Energy Corporation is a holding company, owns no operating assets and has no significant operations independent of its subsidiaries. Our obligations under our outstanding senior notes, contingent convertible senior notes and revolving credit facility listed in Note 3 are fully and unconditionally guaranteed, jointly and severally, by certain of our 100% owned subsidiaries. Subsidiaries with noncontrolling interests, consolidated variable interest entities and certain de minimis subsidiaries are non-quarantors.

The tables below are condensed consolidating financial statements for Chesapeake Energy Corporation (parent) on a stand-alone, unconsolidated basis, and its combined guarantor and combined non-guarantor subsidiaries as of September 30, 2018 and December 31, 2017 and for the three and nine months ended September 30, 2018 and 2017. This financial information may not necessarily be indicative of our results of operations, cash flows or financial position had these subsidiaries operated as independent entities.

### CONDENSED CONSOLIDATING BALANCE SHEET AS OF SEPTEMBER 30, 2018 (\$ in millions)

	F	arent	Guarantor Subsidiaries	on-Guarantor Subsidiaries	Eliminations		c	Consolidated
CURRENT ASSETS:								
Cash and cash equivalents	\$	6	\$ 1	\$ 1	\$	(4)	\$	4
Other current assets		69	1,160	2		_		1,231
Intercompany receivable, net		8,003	29	176		(8,208)		
Total Current Assets		8,078	1,190	179		(8,212)		1,235
PROPERTY AND EQUIPMENT:			_	_		_		
Oil and natural gas properties at cost, based on full cost accounting, net		563	9,433	25		_		10,021
Other property and equipment, net		_	1,109	_		_		1,109
Property and equipment held for sale, net		_	47	_		_		47
Total Property and Equipment, Net		563	10,589	25		_		11,177
LONG-TERM ASSETS:								
Other long-term assets		29	218	_		_		247
Investments in subsidiaries and intercompany advances		1,193	79	_		(1,272)		_
TOTAL ASSETS	\$	9,863	\$ 12,076	\$ 204	\$	(9,484)	\$	12,659
CURRENT LIABILITIES:								
Current liabilities	\$	563	\$ 2,415	\$ 2	\$	(4)	\$	2,976
Intercompany payable, net		29	8,179	_		(8,208)		_
Total Current Liabilities		592	10,594	2		(8,212)		2,976
LONG-TERM LIABILITIES:	-							
Long-term debt, net		9,380	_	_		_		9,380
Other long-term liabilities		53	289	_		_		342
Total Long-Term Liabilities		9,433	289		-			9,722
EQUITY:								
Chesapeake stockholders' equity (deficit)		(162)	1,193	79		(1,272)		(162)
Noncontrolling interests		_	_	123		_		123
Total Equity (Deficit)		(162)	1,193	202		(1,272)		(39)
TOTAL LIABILITIES AND EQUITY	\$	9,863	\$ 12,076	\$ 204	\$	(9,484)	\$	12,659

### CONDENSED CONSOLIDATING BALANCE SHEET AS OF DECEMBER 31, 2017 (\$ in millions)

	 Parent		Guarantor Subsidiaries	Non-Guarantor Subsidiaries	Eliminations		onsolidated
CURRENT ASSETS:							
Cash and cash equivalents	\$ 5	\$	1	\$ 2	\$ (3)	\$	5
Other current assets	154		1,364	3	(1)		1,520
Intercompany receivable, net	8,697		436	_	(9,133)		_
Total Current Assets	8,856		1,801	5	(9,137)		1,525
PROPERTY AND EQUIPMENT:	 _						_
Oil and natural gas properties at cost, based on full cost accounting, net	435		8,888	27	_		9,350
Other property and equipment, net	_		1,314	_	_		1,314
Property and equipment held for sale, net	_		16	_	_		16
Total Property and Equipment, Net	435		10,218	27	_		10,680
LONG-TERM ASSETS:							
Other long-term assets	52		168	_	_		220
Investments in subsidiaries and intercompany advances	806		(146)	_	(660)		_
TOTAL ASSETS	\$ 10,149	\$	12,041	\$ 32	\$ (9,797)	\$	12,425
CURRENT LIABILITIES:							
Current liabilities	\$ 190	\$	2,168	\$ 2	\$ (4)	\$	2,356
Intercompany payable, net	433		8,648	52	(9,133)		_
Total Current Liabilities	623		10,816	54	 (9,137)		2,356
LONG-TERM LIABILITIES:							
Long-term debt, net	9,921		_	_	_		9,921
Other long-term liabilities	101		419	_	_		520
Total Long-Term Liabilities	10,022		419	_	 _		10,441
EQUITY:		,					
Chesapeake stockholders' equity (deficit)	(496)		806	(146)	(660)		(496)
Noncontrolling interests	_		_	124	_		124
Total Equity (Deficit)	(496)		806	(22)	 (660)		(372)
TOTAL LIABILITIES AND EQUITY	\$ 10,149	\$	12,041	\$ 32	\$ (9,797)	\$	12,425

#### CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS THREE MONTHS ENDED SEPTEMBER 30, 2018 (\$ in millions)

	Parent		Guarantor Parent Subsidiaries		Non- Guarantor Subsidiaries	Eliminations	Consolidated	
REVENUES:						-		
Oil, natural gas and NGL	\$	_	\$	1,194	\$ 5	\$ —	\$	1,199
Marketing		_		1,219	_	_		1,219
Total Revenues		_		2,413	5	_		2,418
OPERATING EXPENSES:								
Oil, natural gas and NGL production		_		132	_	_		132
Oil, natural gas and NGL gathering, processing and transportation		_		362	2	_		364
Production taxes		_		33	1	_		34
Marketing		_		1,238	_	_		1,238
General and administrative		_		66	_	_		66
Provision for legal contingencies, net		_		8	_	_		8
Oil, natural gas and NGL depreciation, depletion and amortization		_		274	_	_		274
Depreciation and amortization of other assets		_		17	_	_		17
Impairments				5				5
Total Operating Expenses		_		2,135	3			2,138
INCOME FROM OPERATIONS		_		278	2			280
OTHER INCOME (EXPENSE):								
Interest expense		(125)		(2)	_	_		(127)
Losses on purchases or exchanges of debt		(68)		_	_	_		(68)
Other income		_		1	_	_		1
Equity in net earnings of subsidiary		278		1		(279)		
Total Other Income (Expense)		85				(279)		(194)
INCOME BEFORE INCOME TAXES		85		278	2	(279)		86
INCOME TAX EXPENSE		1						1
NET INCOME		84		278	2	(279)		85
Net income attributable to noncontrolling interests		_			(1)	_		(1)
NET INCOME ATTRIBUTABLE TO CHESAPEAKE		84		278	1	(279)		84
Other comprehensive income		_		8				8
COMPREHENSIVE INCOME ATTRIBUTABLE TO CHESAPEAKE	\$	84	\$	286	\$ 1	\$ (279)	\$	92

### CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS THREE MONTHS ENDED SEPTEMBER 30, 2017 (\$ in millions)

	Parent	Guarantor Subsidiaries	Non- Guarantor Subsidiaries	Eliminations	Consolidated
REVENUES:					
Oil, natural gas and NGL	\$ —	\$ 974	\$ 5	\$ —	\$ 979
Marketing	_	964	_	_	964
Total Revenues	_	1,938	5	_	1,943
OPERATING EXPENSES:					
Oil, natural gas and NGL production		151	_	_	151
Oil, natural gas and NGL gathering, processing and transportation	_	367	2	_	369
Production taxes	_	20	1	_	21
Marketing	_	978	_	_	978
General and administrative	_	54	_	_	54
Provision for legal contingencies, net	_	20	_	_	20
Oil, natural gas and NGL depreciation, depletion and amortization	_	227	1	_	228
Depreciation and amortization of other assets	_	20	_	_	20
Impairments	_	3	_	_	3
Net gains on sales of fixed assets	_	(1)	_	_	(1)
Other operating expense	_	6	_	_	6
Total Operating Expenses		1,845	4		1,849
INCOME FROM OPERATIONS	_	93	1	_	94
OTHER INCOME (EXPENSE):					
Interest expense	(110)	(4)	_	_	(114)
Losses on purchases or exchanges of debt	(1)	_	_	_	(1)
Other income	1	3	_	_	4
Equity in net earnings of subsidiary	92			(92)	
Total Other Income (Expense)	(18)	(1)	_	(92)	(111)
INCOME (LOSS) BEFORE INCOME TAXES	(18)	92	1	(92)	(17)
INCOME TAX EXPENSE	_	_	_	_	
NET INCOME (LOSS)	(18)	92	1	(92)	(17)
Net income attributable to noncontrolling interests	_	_	(1)	_	(1)
NET INCOME (LOSS) ATTRIBUTABLE TO CHESAPEAKE	(18)	92		(92)	(18)
Other comprehensive income	_	8			8
COMPREHENSIVE INCOME (LOSS) ATTRIBUTABLE TO CHESAPEAKE	\$ (18)	\$ 100	\$ —	\$ (92)	\$ (10)

# CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS NINE MONTHS ENDED SEPTEMBER 30, 2018 (\$ in millions)

	Pa	rent	Guaran Subsidia		Non- Guarantor Subsidiaries	Eliminations	Cons	olidated
REVENUES:								
Oil, natural gas and NGL	\$	_	\$ 3	3,410	\$ 14	\$ —	\$	3,424
Marketing		_	3	3,738	_	_		3,738
Total Revenues			7	7,148	14			7,162
OPERATING EXPENSES:								
Oil, natural gas and NGL production		_		417	_	_		417
Oil, natural gas and NGL gathering, processing and transportation		_	1	L,055	5	_		1,060
Production taxes		_		90	1	_		91
Marketing		_	3	3,798	_	_		3,798
General and administrative		_		228	1	_		229
Restructuring and other termination costs		_		38	_	_		38
Provision for legal contingencies, net				17	_	_		17
Oil, natural gas and NGL depreciation, depletion and amortization		_		811	2	_		813
Depreciation and amortization of other assets		_		54	_	_		54
Impairments		_		51	_	_		51
Other operating income		_		(1)	_	_		(1)
Net losses on sales of fixed assets		_		7	_	_		7
Total Operating Expenses		_	6	6,565	9	_		6,574
INCOME FROM OPERATIONS		_		583	5			588
OTHER INCOME (EXPENSE):								
Interest expense		(365)		(2)	_	_		(367)
Gains on investments				139	_	_		139
Losses on exchanges or repurchases of debt		(68)		_	_	_		(68)
Other income		_		63	_	_		63
Equity in net earnings of subsidiary		785		2		(787)		_
Total Other Income (Expense)		352		202	_	(787)		(233)
INCOME BEFORE INCOME TAXES		352		785	5	(787)		355
INCOME TAX BENEFIT		(8)						(8)
NET INCOME		360		785	5	(787)		363
Net income attributable to noncontrolling interests		_		_	(3)	_		(3)
NET INCOME ATTRIBUTABLE TO CHESAPEAKE		360		785	2	(787)		360
Other comprehensive income		_		25	_	_		25
COMPREHENSIVE INCOME ATTRIBUTABLE TO CHESAPEAKE	\$	360	\$	810	\$ 2	\$ (787)	\$	385

# CONDENSED CONSOLIDATING STATEMENT OF OPERATIONS NINE MONTHS ENDED SEPTEMBER 30, 2017 (\$ in millions)

	Parent		Guara Subsidi		Non- Guarantor Subsidiaries	Eliminations	Consoli	dated
REVENUES:								
Oil, natural gas and NGL	\$	_	\$	3,710	\$ 17	\$ —	\$	3,727
Marketing		_		3,250	_	_		3,250
Total Revenues		_		6,960	17	_		6,977
OPERATING EXPENSES:								
Oil, natural gas and NGL production		_		426	_	_		426
Oil, natural gas and NGL gathering, processing and transportation		_		1,075	6	_		1,081
Production taxes		_		63	1	_		64
Marketing		_		3,333	_	_		3,333
General and administrative		3		185	1	_		189
Provision for legal contingencies, net		_		35	_	_		35
Oil, natural gas and NGL depreciation, depletion and amortization		_		624	3	_		627
Depreciation and amortization of other assets		_		62	_	_		62
Impairments				3	_	_		3
Other operating expense				423				423
Total Operating Expenses		3		6,229	11	_		6,243
INCOME (LOSS) FROM OPERATIONS		(3)		731	6			734
OTHER INCOME (EXPENSE):								
Interest expense		(300)		(2)	_	_		(302)
Gains on purchases or exchanges of debt		183			_	_		183
Other income		1		5	_	_		6
Equity in net earnings of subsidiary		737		3		(740)		
Total Other Income (Expense)		621		6		(740)		(113)
INCOME BEFORE INCOME TAXES		618		737	6	(740)		621
INCOME TAX EXPENSE		2						2
NET INCOME		616		737	6	(740)		619
Net income attributable to noncontrolling interests		_		_	(3)	_		(3)
NET INCOME ATTRIBUTABLE TO CHESAPEAKE	1	616		737	3	(740)		616
Other comprehensive income		_		29	_			29
COMPREHENSIVE INCOME ATTRIBUTABLE TO CHESAPEAKE	\$	616	\$	766	\$ 3	\$ (740)	\$	645

# CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS NINE MONTHS ENDED SEPTEMBER 30, 2018 (\$ in millions)

	Parent	Guarantor Subsidiaries	Non- Guarantor Subsidiaries	Eliminations	Consolidated
CASH FLOWS FROM OPERATING ACTIVITIES:					
Net Cash Provided By Operating Activities	\$ 86	\$ 1,512	\$ 7	\$ (10)	\$ 1,595
CASH FLOWS FROM INVESTING ACTIVITIES:					
Drilling and completion costs	_	(1,481)	_	_	(1,481)
Acquisitions of proved and unproved properties	_	(244)	_	_	(244)
Proceeds from divestitures of proved and unproved properties	_	395	_	_	395
Additions to other property and equipment	_	(11)	_	_	(11)
Other investing activities	_	149	_	_	149
Net Cash Used In Investing Activities		(1,192)	_		(1,192)
CASH FLOWS FROM FINANCING ACTIVITIES:					
Proceeds from revolving credit facility borrowings	9,095	_	_	_	9,095
Payments on revolving credit facility borrowings	(9,231)	_	_	_	(9,231)
Proceeds from issuance of senior notes, net	1,237	_	_	_	1,237
Cash paid to purchase debt	(1,285)	_	_	_	(1,285)
Cash paid for preferred stock dividends	(69)	_	_	_	(69)
Other financing activities	(2)	(127)	(9)	(13)	(151)
Intercompany advances, net	170	(193)	1	22	
Net Cash Used In Financing Activities	(85)	(320)	(8)	9	(404)
Net increase (decrease) in cash and cash equivalents	1	_	(1)	(1)	(1)
Cash and cash equivalents, beginning of period	5	1	2	(3)	5
Cash and cash equivalents, end of period	\$ 6	\$ 1	\$ 1	\$ (4)	\$ 4

# CONDENSED CONSOLIDATING STATEMENT OF CASH FLOWS NINE MONTHS ENDED SEPTEMBER 30, 2017 (\$ in millions)

	Parent	Guarantor Subsidiaries	Non- Guarantor Subsidiaries	Eliminations	Consolidated
CASH FLOWS FROM OPERATING ACTIVITIES:					
Net Cash Provided By Operating Activities	\$ 4	\$ 266	\$ 11	\$ (8)	\$ 273
CASH FLOWS FROM INVESTING ACTIVITIES:					
Drilling and completion costs	_	(1,597)	_	_	(1,597)
Acquisitions of proved and unproved properties	_	(226)	_	_	(226)
Proceeds from divestitures of proved and unproved properties	_	1,193	_	_	1,193
Additions to other property and equipment	_	(12)	_	_	(12)
Other investing activities	_	40			40
Net Cash Used In Investing Activities		(602)		_	(602)
CASH FLOWS FROM FINANCING ACTIVITIES:					
Proceeds from revolving credit facility borrowings	4,775	_	_	_	4,775
Payments on revolving credit facility borrowings	(4,130)	_	_	_	(4,130)
Proceeds from issuance of senior notes, net	742	_	_	_	742
Cash paid to purchase debt	(1,751)	_	_	_	(1,751)
Cash paid for preferred stock dividends	(160)	_	_	_	(160)
Other financing activities	(36)	(4)	(11)	27	(24)
Intercompany advances, net	(339)	339			
Net Cash Provided by (Used In) Financing Activities	(899)	335	(11)	27	(548)
Net increase (decrease) in cash and cash equivalents	(895)	(1)		19	(877)
Cash and cash equivalents, beginning of period	904	2	1	(25)	882
Cash and cash equivalents, end of period	\$ 9	\$ 1	\$ 1	\$ (6)	\$ 5

#### 17. Subsequent Events

On October 30, 2018, we jointly announced with WildHorse Resource Development Corporation ("WildHorse") that we have entered into a definitive agreement to acquire WildHorse, an oil and gas company with operations in the Eagle Ford Shale and Austin Chalk formations in southeast Texas, in a transaction valued at approximately \$3.977 billion, including the value of WildHorse's net debt of \$930 million as of June 30, 2018. At the election of each WildHorse common shareholder, the consideration will consist of either 5.989 shares of Chesapeake common stock or a combination of 5.336 shares of Chesapeake common stock and \$3.00 in cash for each share of WildHorse common stock. We intend to fund the cash portion of the consideration through borrowings under our revolving credit facility. The transaction has been unanimously approved by the Board of Directors of each company and is subject to shareholder approvals from both companies and customary closing conditions and regulatory approvals and is expected to close in the first half of 2019.

On October 29, 2018, we completed the sale of our Utica Shale assets in Ohio to Encino, pursuant to which Encino purchased all of our approximately 1,500,000 gross (900,000 net) acres in Ohio, of which approximately 320,000 net acres are prospective for the Utica Shale with approximately 920 producing wells, along with related property and equipment (collectively, the "Designated Properties") for net proceeds of \$1.868 billion in cash.

On October 29, 2018, we delivered a notice of redemption to the trustee for our 8.00% Senior Secured Second Lien Notes due 2022 to call for redemption \$1.416 billion aggregate principal amount of the outstanding notes, representing 100% of the aggregate principal amount of the outstanding notes. The notes will be redeemed at a redemption price of 100% of the principal amount thereof, plus the make-whole premium, as calculated in accordance with the indenture, plus accrued and unpaid interest. The settlement of the redemption is expected to occur approximately 30 days from the notice delivery date. The redemption is expected to be funded with proceeds from the sale of our Utica assets in Ohio.

#### ITEM 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

#### Introduction

The following discussion should be read together with the condensed consolidated financial statements included in Item 1 of Part I of this report and our 2017 Form 10-K.

We are an independent exploration and production company engaged in the acquisition, exploration and development of properties for the production of oil, natural gas and NGL from underground reservoirs. We own a large and geographically diverse portfolio of onshore U.S. unconventional natural gas and liquids assets, including interests in approximately 14,900 oil and natural gas wells. We have leading positions in the liquids-rich resource plays of the Eagle Ford Shale in South Texas, the stacked pay in the Powder River Basin in Wyoming and the Anadarko Basin in northwestern Oklahoma. Our natural gas resource plays are the Marcellus Shale in the northern Appalachian Basin in Pennsylvania and the Haynesville/Bossier Shales in northwestern Louisiana and East Texas.

Our strategy is to create shareholder value through the development of our significant resource plays. We continue to focus on reducing debt, increasing cash provided by operating activities, and improving margins through financial discipline and operating efficiencies. Our capital program is focused on investments that can improve our cash flow generating ability even in a challenging commodity price environment. Although we expect our forecasted capital expenditures in 2018 to be lower compared to 2017, we anticipate modest production growth from both our oil-producing and natural gas-producing assets, adjusted for asset sales. Our ability to reduce capital expenditures while still growing production is primarily the result of improved drilling and completion efficiencies and improved well performance. We continue to seek opportunities to reduce cash costs (production, gathering, processing and transportation, general and administrative and interest expenses) and improve our production volumes from existing wells.

We believe that our dedication to financial discipline, the flexibility and efficiency of our capital program and cost structure and our continued focus on safety and environmental stewardship will provide opportunities to create value for us and our shareholders.

In 2018, our focus is concentrated on three strategic priorities:

- reduce total debt by \$2 \$3 billion;
- increase net cash provided by operating activities to fund capital expenditures; and
- improve margins through financial discipline and operating efficiencies.

On October 29, 2018 we completed the sale of all of our assets in Ohio. This divestiture will result in our meeting or making significant progress toward all three of these priorities. The following discussion and analysis presents management's perspective of our business and material changes to our results of operations for the three and nine months ended September 30, 2018 compared to the three and nine months ended September 30, 2017 and in our financial condition and liquidity since December 31, 2017.

#### Overview

The transformation of Chesapeake over the past five years has been significant and our progress has continued in the Current Period. Our basic strategies have not changed through the price cycles of the past several years, and we believe our recent accomplishments and achievements in the Current Period have made our company stronger. Our progress has been guided by our strategies of financial discipline, pursuing profitable and efficient growth from our captured resources, leveraging technology and our operational expertise to unlock additional domestic resources and optimizing our portfolio through business development.

We have made significant progress towards achieving our strategic priorities in 2018. So far we have:

- entered into a definitive agreement to acquire WildHorse, an oil and gas company with operations in the Eagle Ford Shale and Austin Chalk formations in southeast Texas, in a transaction valued at approximately \$3.977 billion;
- sold our interests in the Utica Shale operating area located in Ohio for approximately \$1.868 billion, with an additional contingent
  payment to us of up to \$100 million based on future natural gas prices. We intend to use the net proceeds to reduce our
  indebtedness, including the redemption of our senior secured second lien notes, representing the remainder of our non-credit facility
  secured debt:
- retired our secured term loan borrowings and significantly extended our debt maturity profile by issuing at par \$850 million of 7.00%
   Senior Notes due 2024 and \$400 million of 7.50%
   Senior Notes due 2026 pursuant to a public offering for net proceeds of \$1.230 billion. We used the proceeds from these unsecured senior notes together with cash on hand and borrowings under our revolving credit facility, to repay in full \$1.233 billion of borrowings under our secured term loan due 2021;
- amended and restated our credit agreement dated December 15, 2014. The amended and restated revolving credit facility matures
  in September 2023 and the aggregate initial commitment of the lenders and borrowing base under the facility is \$3.0 billion. The
  revolving credit facility provides for an accordion feature, pursuant to which the aggregate commitments thereunder may be
  increased to up to \$4.0 billion from time to time, subject to agreement of the participating lenders and certain other customary
  conditions;
- repurchased the CHK Utica, L.L.C. investors' ORRI for \$199 million in an effort to remove financial and operational complexity and to improve our balance sheet;
- sold properties in the Mid-Continent, including our Mississippian Lime assets, for aggregate proceeds of approximately \$500 million;
- received net proceeds of approximately \$74 million from the sale of approximately 4.3 million shares of FTS International, Inc.
  (NYSE: FTSI). FTSI is a provider of hydraulic fracturing services in North America and a company in which Chesapeake has owned
  a significant stake since 2006. FTSI completed its initial public offering of common shares on February 6, 2018. We currently own
  approximately 22.0 million shares of FTSI; and
- reduced annual cash costs by approximately \$70 million through a reduction in workforce that better aligns our workforce to the needs of our business.

We continue to benefit from progress made over the last five years, including removing financial and operational complexity, significantly improving our balance sheet and addressing numerous legacy issues.

#### Financial Results

	Three Months Ended September 30,				Nine Months Ended September 30,					
		2018		2017	Change <sup>(b)</sup>	2018		2017		Change
					(\$ in m	nillie	ons)			
Net income (loss) available to common stockholders	\$	60	\$	(41)	n/m	\$	288	\$	506	(43)%
Net earnings (loss) per diluted common share	\$	0.07	\$	(0.05)	n/m	\$	0.32	\$	0.56	(43)%
Total production (mboe per day)		537		542	(1)%		540		532	2 %
Adjusted production <sup>(a)</sup> (mboe per day)		536		512	5 %		534		494	8 %
Average sales price (per boe)	\$	26.92	\$	21.06	28 %	\$	26.59	\$	22.53	18 %
Oil, natural gas and NGL production expenses	\$	132	\$	151	(13)%	\$	417	\$	426	(2)%
Oil, natural gas and NGL gathering, processing and transportation expenses	\$	364	\$	369	(1)%	\$	1,060	\$	1,081	(2)%
General and administrative expenses	\$	66	\$	54	22 %	\$	229	\$	189	21 %
				•	mber 30, 2018	ı	Decembe	er 31	L, 2017	Change
Total debt (principal amount)					\$9,862				\$9,981	(1)%

<sup>(</sup>a) Adjusted for assets sold.

<sup>(</sup>b) n/m - not meaningful.

#### **Liquidity and Capital Resources**

Liquidity Overview

Our ability to grow, make capital expenditures and service our debt depends primarily upon the prices we receive for the oil, natural gas and NGL we sell. Substantial expenditures are required to replace reserves, sustain production and fund our business plans. Historically, oil and natural gas prices have been volatile, and may be subject to wide fluctuations in the future. A decline in oil, natural gas and NGL prices could negatively affect the amount of cash we generate and have available for capital expenditures and debt service and could have a material impact on our financial position, results of operations, cash flows and on the quantities of reserves that we can economically produce or provide as collateral to our credit facility lenders. Other risks and uncertainties that could affect our liquidity include, but are not limited to, counterparty credit risk for our receivables, access to capital markets, regulatory risks and our ability to meet financial covenants in our financing agreements.

Based on our cash balance, forecasted cash flows from operating activities and availability under our revolving credit facility, we expect to be able to fund our planned capital expenditures, meet our debt service requirements and fund our other commitments and obligations for the next 12 months.

As of September 30, 2018, we had a cash balance of \$4 million compared to \$5 million as of December 31, 2017, and we had a net working capital deficit of \$1.741 billion as of September 30, 2018, compared to a net working capital deficit of \$831 million as of December 31, 2017. As of September 30, 2018, our working capital deficit includes \$432 million principal amount of debt due or that could be put to us in the next 12 months. As of September 30, 2018, we had \$2.173 billion of borrowing capacity available under our senior secured revolving credit facility, with outstanding borrowings of \$645 million and \$182 million utilized for various letters of credit. See Note 3 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of our debt obligations, including principal and carrying amounts of our notes.

Even though we have taken measures to mitigate the liquidity concerns facing us for the next 12 months as outlined above and in *Industry Outlook* in our 2017 Form 10-K, there can be no assurance that these measures will be sufficient for periods beyond the next 12 months. If needed, we may seek to access the capital markets or otherwise refinance a portion of our outstanding indebtedness to improve our liquidity. We closely monitor the amounts and timing of our sources and uses of funds, particularly as they affect our ability to maintain compliance with the financial covenants of our revolving credit facility. Furthermore, our ability to generate operating cash flow in the current commodity price environment, sell assets, access capital markets or take any other action to improve our liquidity and manage our debt is subject to the risks discussed above and elsewhere in our periodic reports and the other risks and uncertainties that exist in our industry, some of which we may not be able to anticipate at this time or control.

Year

2019

2019

2019

2019 2019

2020

**Swaps** 

Calls

Calls

Two-way collars

Three-way collars

Basis protection swaps

#### Derivative and Hedging Activities

Our results of operations and cash flows are impacted by changes in market prices for oil, natural gas and NGL. To mitigate a portion of our exposure to adverse market changes, we have entered into various derivative instruments. Our oil, natural gas and NGL derivative activities, when combined with our sales of oil, natural gas and NGL, allow us to better predict the total revenue we expect to receive.

We utilize various oil, natural gas and NGL derivative instruments to protect a portion of our cash flow against downside risk. As of October 26, 2018, we have downside price protection in the remainder of 2018, 2019 and 2020 through the following oil, natural gas and NGL derivative instruments:

#### Oil Derivatives(a)

**Notional Volume** 

325

54

88

37

22

22

**Average NYMEX Price** 

\$2.83

\$2.75/\$3.02

\$2.50/\$2.80/\$3.10

\$0.03

\$12.00

\$12.00

Type of Derivative Instrument

		(mmbbls)							
2018	Swaps	7	\$54.09						
2018	Three-way collars	1	\$39.15/\$47.00/\$55.00						
2018	Basis protection swaps	4	\$3.52						
2019	Swaps	14	\$59.44						
2019	Basis protection swaps	7	\$6.01						
2020	Swaps	3	\$69.47						
Natural Gas Derivatives <sup>(a)</sup>									
Year	Type of Derivative Instrument	Notional Volume	Average NYMEX Price						
	<del></del>	(bcf)							
2018	Swaps	120	\$3.00						
2018	Two-way collars	12	\$3.00/\$3.25						
2018	Calls	17	\$6.27						
		_	(+)						
2018	Basis protection swaps	6	(\$0.77)						

#### NGL Derivatives(a)

Year	Type of Derivative Instrument	Notional Volume	Average NYMEX Price
		(mmgal)	
2018	Butane swaps	1	\$0.88
2018	Butane % of WTI swaps	1	70.5% of WTI
2018	Propane swaps	15	\$0.79
2018	Ethane swaps	23	\$0.29
2018	Isobutane swaps	4	\$0.92
2018	Natural gasoline	12	\$1.42

<sup>(</sup>a) Includes amounts settled in October 2018 and excludes derivatives novated to Encino on October 29, 2018.

See Note 8 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of derivatives and hedging activities.

#### Debt

We are committed to decreasing the amount of debt outstanding by \$2-3 billion in 2018. To accomplish this objective, we intend to use the anticipated net proceeds from the pending sale of our Utica interests, allocate our capital expenditures to the highest-return projects, deploy leading drilling and completion technology throughout our portfolio to profitably and efficiently grow, and divest additional assets to strengthen our cost structure and our portfolio. We are seeking to reduce cash costs (production, gathering, processing and transportation, general and administrative and interest expenses), improve our production volumes from existing wells, and achieve additional operating and capital efficiencies with a focus on growing our oil volumes.

We may continue to use a combination of cash, borrowings and issuances of our common stock or other securities and the proceeds from asset sales to retire our outstanding debt and/or preferred stock through privately negotiated transactions, open market repurchases, redemptions, tender offers or otherwise, but we are under no obligation to do so.

#### Revolving Credit Facility

In the Current Quarter, we amended and restated our senior secured revolving credit facility which is currently subject to a \$3.0 billion borrowing base and matures in September 2023. As of September 30, 2018, we had \$2.173 billion of borrowing capacity available under our revolving credit facility. Our next borrowing base redetermination is scheduled for the second quarter of 2019. As of September 30, 2018, we had outstanding borrowings of \$645 million under the revolving credit facility and had used \$182 million of the revolving credit facility for various letters of credit. Borrowings under the facility bear interest at a variable rate. See Note 3 of the notes to our condensed consolidated financial statements included in Item 1 of this report for further discussion of the terms of the revolving credit facility. As of September 30, 2018, we were in compliance with all applicable financial covenants under the credit agreement. Our total leverage ratio was approximately 3.82 to 1.00, our first lien secured leverage ratio was approximately 0.38 to 1.00.

#### Contractual Obligations and Off-Balance Sheet Arrangements

From time to time, we enter into arrangements and transactions that can give rise to contractual obligations and off-balance sheet commitments. As of September 30, 2018, these arrangements and transactions included (i) operating lease agreements, (ii) a volumetric production payment (VPP) (to purchase production and pay related production expenses and taxes in the future), (iii) open purchase commitments, (iv) open delivery commitments, (v) open drilling commitments, (vi) undrawn letters of credit, (vii) open gathering and transportation commitments, and (viii) various other commitments we enter into in the ordinary course of business that could result in a future cash obligation. See Notes 4 and 9 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of commitments and VPPs, respectively.

#### Capital Expenditures

Our 2018 capital expenditures program, while planned to be lower than our 2017 program, is expected to generate greater capital efficiency as we focus on expanding our margins by investing in the highest-return projects. We have significant control and flexibility over the timing and execution of our development plan, enabling us to reduce our capital spending as needed. Our forecasted 2018 capital expenditures, inclusive of capitalized interest, are \$2.2 – \$2.5 billion compared to our 2017 capital spending level of \$2.5 billion. Management continues to review operational plans for 2018 and beyond, which could result in changes to projected capital expenditures and projected revenues from sales of oil, natural gas and NGL.

#### Credit Risk

Some of our counterparties have requested or required us to post collateral as financial assurance of our performance under certain contractual arrangements, such as gathering, processing, transportation and hedging agreements. As of October 26, 2018, we have received requests and posted approximately \$222 million of collateral related to certain of our marketing and other contracts. We may be requested or required by other counterparties to post additional collateral in an aggregate amount of approximately \$441 million, which may be in the form of additional letters of credit, cash or other acceptable collateral. However, we have substantial long-term business relationships with each of these counterparties, and we may be able to mitigate any collateral requests through ongoing business arrangements and by offsetting amounts that the counterparty owes us. Any posting of collateral consisting of cash or letters of credit reduces availability under our revolving credit facility and negatively impacts our liquidity.

#### Sources of Funds

The following table presents the sources of our cash and cash equivalents for the Current Period and the Prior Period. See Note 9 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of divestitures of oil and natural gas assets.

	Nine Months Ended September 30,				
	 2018 201				
	 (\$ in n	(\$ in millions)			
Cash provided by operating activities	\$ 1,595	\$	273		
Proceeds from divestitures of proved and unproved properties, net	395		1,193		
Proceeds from issuance of senior notes, net	1,237		742		
Proceeds from issuance of credit facility borrowings, net	_		645		
Proceeds from sales of other property and equipment, net	75		40		
Proceeds from sales of investments	74		_		
Total sources of cash and cash equivalents	\$ 3,376	\$	2,893		

#### Cash Flow from Operating Activities

Cash provided by operating activities was \$1.595 billion in the Current Period compared to cash used by operating activities of \$273 million in the Prior Period. The increase in the Current Period is primarily due to the result of higher prices for the oil and NGL we sold and higher volumes of oil and natural gas sold. Changes in cash flow from operations are largely due to the same factors that affect our net income, excluding various non-cash items, such as depreciation, depletion and amortization, certain impairments, gains or losses on sales of fixed assets, deferred income taxes and mark-to-market changes in our open derivative instruments. See further discussion below under *Results of Operations*.

#### Uses of Funds

The following table presents the uses of our cash and cash equivalents for the Current Period and the Prior Period:

	Nine Mon Septer		
	 2018		2017
	 (\$ in m	nillion	ıs)
Oil and Natural Gas Expenditures:			
Drilling and completion costs	\$ 1,481	\$	1,597
Acquisitions of proved and unproved properties	123		87
Interest capitalized on unproved leasehold	121		139
Total oil and natural gas expenditures	1,725		1,823
Other Uses of Cash and Cash Equivalents:			
Payments on revolving credit facility borrowings, net	136		_
Extinguishment of other financing	122		_
Cash paid to purchase debt	1,285		1,751
Additions to other property and equipment	11		12
Dividends paid	69		160
Other	 29		24
Total other uses of cash and cash equivalents	1,652		1,947
Total uses of cash and cash equivalents	\$ 3,377	\$	3,770

#### **Drilling and Completion Costs**

Our drilling and completion costs decreased in the Current Period compared to the Prior Period primarily as a result of lower rig and completion activity. During the Current Period, our average operated rig count was 17 rigs compared to an average operated rig count of 18 rigs in the Prior Period and we completed 242 operated wells in the Current Period compared to 326 in the Prior Period.

#### Extinguishment of Other Financing

In the Current Period, we repurchased previously conveyed overriding royalty interests (ORRIs) from the CHK Utica, L.L.C. investors and extinguished our obligation to convey future ORRIs to the investors for combined consideration of \$199 million. The cash paid was bifurcated between extinguishment of the obligation and acquisition of the ORRI. See Note 5 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of the transaction.

#### Repurchase and Repayment of Debt

In the Current Quarter, we used \$1.285 billion of cash from the issuance of senior notes together with cash on hand and borrowings under our revolving credit facility to repay in full \$1.233 billion principal amount of borrowings under our secured term loan due 2021 plus a call premium of \$52 million. In the Prior Period, we used \$1.751 billion of cash from debt issuances to repurchase \$1.609 billion principal amount of debt.

#### Dividends

We paid dividends of \$69 million on our preferred stock during the Current Period and we paid dividends of \$160 million on our preferred stock in the Prior Period, including \$92 million of dividends in arrears that had been suspended throughout 2016. We eliminated common stock dividends in the 2015 third guarter and do not anticipate paying any common stock dividends in the foreseeable future.

# **Results of Operations**

Oil, Natural Gas and NGL Production and Average Sales Prices

Three Months Ended September 30, 203
--------------------------------------

		Three months Ended September 30, 2010							
	0	Oil		l Gas	NGL		Total		
	mbbl per day	\$/bbl	mmcf per day	\$/mcf	mbbl per day	\$/bbl	mboe per day	%	\$/boe
Marcellus			812	2.46			135	25	14.74
Haynesville	_	_	769	2.74	_	_	128	24	16.44
Eagle Ford	58	74.40	122	3.26	22	28.95	100	19	53.43
Utica	10	67.09	488	2.92	28	29.39	119	22	24.33
Mid-Continent	9	69.41	66	2.50	4	29.40	25	5	37.68
Powder River Basin	12	69.23	73	2.50	5	27.89	29	5	39.79
Retained assets <sup>(a)</sup>	89	72.39	2,330	2.69	59	29.10	536	100	26.92
Divested assets	_	_	2	2.02	_	_	1	_	19.17
Total	89	72.39	2,332	2.69	59	29.09	537	100%	26.92

### Three Months Ended September 30, 2017

					•	,				
	0	il	Natura	l Gas	NC	GL	Total			
	mbbl per day	\$/bbl	mmcf per day	\$/mcf	mbbl per day	\$/bbl	mboe per day	%	\$/boe	
Marcellus	_		748	1.96			125	23	11.76	
Haynesville	_	_	804	2.77	_	_	134	25	16.63	
Eagle Ford	52	49.08	136	3.25	18	23.07	92	17	36.91	
Utica	12	44.18	475	2.76	28	20.30	120	22	20.21	
Mid-Continent	10	46.98	69	2.54	6	22.18	27	5	28.03	
Powder River Basin	5	47.12	35	2.91	3	26.77	14	2	31.01	
Retained assets <sup>(a)</sup>	79	47.96	2,267	2.52	55	21.70	512	94	20.94	
Divested assets	7	47.71	115	2.47	4	23.63	30	6	23.25	
Total	86	47.94	2,382	2.52	59	21.83	542	100%	21.06	

### Nine Months Ended September 30, 2018

	0	il	Natura	l Gas	NC	GL.									
	mbbl per day	\$/bbl	mmcf per day	\$/mcf	mbbl per day	\$/bbl	mboe per day	%	\$/boe						
Marcellus			829	2.85			138	26	17.13						
Haynesville	_	_	811	2.72	_	_	135	25	16.34						
Eagle Ford	60	70.33	135	3.26	19	26.90	102	19	50.79						
Utica	10	63.39	446	2.88	26	26.65	111	20	23.78						
Mid-Continent	9	66.08	67	2.51	5	26.63	25	5	36.02						
Powder River Basin	10	67.01	59	2.48	4	27.86	23	4	38.31						
Retained assets <sup>(a)</sup>	89	68.73	2,347	2.82	54	26.82	534	99	26.53						
Divested assets	2	63.37	22	2.77	1	29.62	6	1	31.47						
Total	91	68.63	2,369	2.82	55	26.87	540	100%	26.59						

Nine Months Ended September 30, 2017

	0	il	Natura	l Gas	NC	GL.		Total	
	mbbl per day	\$/bbl	mmcf per day	\$/mcf	mbbl per day	\$/bbl	mboe per day	%	\$/boe
Marcellus			796	2.52			132	25	15.14
Haynesville	_	_	736	2.90	_	_	123	23	17.39
Eagle Ford	55	49.42	140	3.36	18	21.27	96	18	37.22
Utica	9	44.01	410	3.12	26	20.87	104	19	21.54
Mid-Continent	8	47.43	69	2.85	5	21.02	25	5	27.65
Powder River Basin	6	48.12	34	3.06	3	24.52	14	3	31.58
Retained assets <sup>(a)</sup>	78	48.48	2,185	2.83	52	21.20	494	93	22.43
Divested assets	8	49.00	154	2.75	5	22.31	38	7	23.77
Total	86	48.53	2,339	2.83	57	21.28	532	100%	22.53

(a) Includes assets retained as of September 30, 2018.

Oil, Natural Gas and NGL Sales

	Three Months Ended September 30,						nded 0,			
	2018 2017			2017	Change		2018	2017		Change
					(\$ in m	illic	ons)			
Oil	\$	594	\$	379	57%	\$	1,698	\$	1,140	49%
Natural gas		578		553	5%		1,822		1,807	1%
NGL		159		117	36%		404		328	23%
Oil, natural gas and NGL sales	\$	1,331	\$	1,049	27%	\$	3,924	\$	3,275	20%

The increase in the price received per boe in the Current Quarter resulted in a \$291 million increase in revenues, and decreased sales volumes resulted in a \$9 million decrease in revenues, for a total net increase in revenues of \$282 million. The increase in the price received per boe in the Current Period resulted in a \$599 million increase in revenues, and increased sales volumes resulted in a \$50 million increase in revenues, for a total net increase in revenues of \$649 million.

A change in oil, natural gas and NGL prices has a significant impact on our revenues and cash flows. Assuming our Current Quarter production levels and without considering the effect of derivatives, an increase or decrease of \$1.00 per barrel of oil sold would have resulted in an increase or decrease in Current Quarter revenues and cash flows from operations of approximately \$8 million, an increase or decrease of \$0.10 per mcf of natural gas sold would have resulted in an increase or decrease in Current Quarter revenues and cash flows from operations of approximately \$21 million and an increase or decrease of \$1.00 per barrel of NGL sold would have resulted in an increase or decrease in Current Quarter revenues and cash flows from operations of approximately \$5 million. Assuming our Current Period production levels and without considering the effect of derivatives, an increase or decrease of \$1.00 per barrel of oil sold would have resulted in an increase or decrease in Current Period revenues and cash flows from operations of approximately \$24 million, an increase or decrease of \$0.10 per mcf of natural gas sold would have resulted in an increase or decrease in Current Period revenues and cash flows from operations of approximately \$65 million and \$64 million, respectively, and an increase or decrease of \$1.00 per barrel of NGL sold would have resulted in an increase or decrease in Current Period revenues and cash flows from operations of approximately \$15 million.

#### Oil, Natural Gas and NGL Derivatives

	Three Mon Septen	ths Ended nber 30,		Nine Mon Septen	 
	2018	2017		2018	2017
		(\$ in m	nillic	ons)	
Oil derivatives – realized gains (losses)	\$ (112)	\$ 35	\$	(273)	\$ 79
Oil derivatives – unrealized gains (losses)	12	(96)		(115)	45
Total gains (losses) on oil derivatives	(100)	(61)		(388)	124
				_	
Natural gas derivatives – realized gains (losses)	(1)	(1)		83	(53)
Natural gas derivatives – unrealized gains (losses)	(17)	(3)		(168)	384
Total gains (losses) on natural gas derivatives	(18)	(4)		(85)	 331
NGL derivatives – realized gains (losses)	(10)	(3)		(14)	(1)
NGL derivatives – unrealized gains (losses)	(4)	(2)		(13)	(2)
Total gains (losses) on NGL derivatives	(14)	(5)		(27)	(3)
Total gains (losses) on oil, natural gas and NGL derivatives	\$ (132)	\$ (70)	\$	(500)	\$ 452

See Note 8 of the notes to our condensed consolidated financial statements included in Item 1 of this report for a discussion of our derivative activity.

#### Marketing Revenues and Expenses

In connection with the marketing of our production, we take title to the oil, natural gas and NGL we purchase from other working interest owners at defined delivery points and deliver the product to third parties, at which time revenues are recorded. In circumstances where we act as a principal rather than an agent, revenue is presented on a gross basis. Marketing revenues primarily consist of marketing services, including commodity price structuring, securing and negotiating gathering, hauling, processing and transportation services, contract administration and nomination services for Chesapeake and other interest owners in Chesapeake-operated wells.

	 Three Months Ended September 30,						nded 0,	
	2018		2017	Change		2018	2017	Change
				(\$ in m	illio	ns)		
Marketing revenues	\$ 1,219	\$	964	26 %	\$	3,738	\$ 3,250	15%
Marketing expenses	1,238		978	27 %		3,798	3,333	14%
Marketing gross margin	\$ (19)	\$	(14)	(36)%	\$	(60)	\$ (83)	28%

Marketing revenues and expenses increased in the Current Quarter and the Current Period primarily as a result of increased oil, natural gas and NGL prices received in our marketing operations.

#### Oil, Natural Gas and NGL Production Expenses

	Three Months Ended September 30,						Ni		Months Er otember 3	
		2018		2017	Change		2018		2017	Change
					(\$ in m	illic	ons)			
Marcellus	\$	8	\$	7	14 %	\$	23	\$	17	35 %
Haynesville		15		16	(6)%		45		38	18 %
Eagle Ford		41		51	(20)%		141		142	(1)%
Utica		11		11	— %		33		30	10 %
Mid-Continent		29		29	— %		78		82	(5)%
Powder River Basin		12		7	71 %		35		21	67 %
Retained Assets <sup>(a)</sup>		116		121	(4)%		355		330	8 %
Divested Assets		_		18	(100)%		14		60	(77)%
Total		116		139	(17)%		369		390	(5)%
Ad valorem tax		16		12	33 %		48		36	33 %
	_		_							
Total oil, natural gas and NGL production expenses	\$	132	\$	151	(13)%	\$	417	\$	426	(2)%
					(\$ pe	r ho	ne)			
Marcellus	\$	0.62	\$	0.58	7 %		0.61	\$	0.46	33 %
Haynesville	\$	1.28	\$	1.27	1 %	\$	1.21	\$	1.12	8 %
Eagle Ford	\$	4.52	\$	6.10	(26)%		5.08	\$	5.41	(6)%
Utica	\$	0.98	\$	0.98	— %	\$	1.10	\$	1.06	4 %
Mid-Continent	\$	12.95	\$	11.44	13 %	\$	11.37	\$	11.98	(5)%
Powder River Basin	\$	4.39	\$	5.89	(25)%	\$	5.47	\$	5.56	(2)%
Retained Assets <sup>(a)</sup>	\$	2.36	\$	2.57	(8)%	_	2.43	\$	2.44	— %
Divested Assets	\$	_	\$	6.47	(100)%	\$	8.19	\$	5.83	40 %
Total	\$	2.36	\$	2.79	(15)%	_	2.50	\$	2.68	(7)%
	÷		<u> </u>		( -)			· <u> </u>		( )
Ad valorem tax	\$	0.32	\$	0.24	33 %	\$	0.33	\$	0.25	32 %
								_		
Total oil, natural gas and NGL production expenses per boe	\$	2.68	\$	3.03	(12)%	\$	2.83	\$	2.93	(3)%

<sup>(</sup>a) Includes assets retained as of September 30, 2018.

The absolute and per unit decrease in the Current Quarter was the result of decreased workover activity in Eagle Ford and the sale of certain oil and natural gas properties in 2017 and 2018. The absolute and per unit decrease in the Current Period was the result of the sale of certain oil and natural gas properties in 2017 and 2018.

Production expenses in the Current Quarter, the Prior Quarter, the Current Period and the Prior Period included approximately \$4 million, \$5 million, \$12 million and \$15 million associated with VPP production volumes, respectively. We anticipate a continued decrease in production expenses associated with VPP production volumes as the contractually scheduled volumes under our remaining VPP agreement decrease and operating efficiencies generally improve.

Oil, Natural Gas, and NGL Gathering, Processing and Transportation Expenses

	Three Months Ended September 30,					Nine Mor Septer				
		2018		2017		2018		2017		
			(\$ iı	n millions,	exce	ept per unit	<b>:</b> )			
Oil, natural gas and NGL gathering, processing and transportation expenses	\$	364	\$	369	\$	1,060	\$	1,081		
Oil (\$ per bbl)	\$	3.83	\$	4.33	\$	3.75	\$	3.96		
Natural gas (\$ per mcf)	\$	1.33	\$	1.34	\$	1.30	\$	1.36		
NGL (\$ per bbl)	\$	8.59	\$	7.40	\$	8.62	\$	7.90		
Total (\$ per boe)	\$	7.36	\$	7.40	\$	7.18	\$	7.43		

The absolute and per unit decrease in oil, natural gas and NGL gathering, processing and transportation expenses was primarily due to lower gathering fees associated with restructured midstream contracts, lower volume commitments on downstream pipelines and certain 2017 and 2018 divestitures.

#### **Production Taxes**

	Three Months Ended September 30,						nded 0,		
	2018		2017	Change		2018		2017	Change
			(;	\$ in millions,	exce	pt per u	ınit)		
Production taxes	\$ 34	\$	21	62%	\$	91	\$	64	42%
Production taxes per boe	\$ 0.69	\$	0.43	60%	\$	0.62	\$	0.44	41%

The absolute and per unit increase in production taxes was primarily due to higher prices received for our oil, natural gas and NGL production.

#### General and Administrative Expenses

	 Three Months Ended September 30,						nded 0,				
	2018 2017 Change					2018		2017	Change		
	 (\$ in millions, except per unit)										
Gross overhead	\$ 175	\$	187	(6)%	\$	564	\$	595	(5)%		
Allocated to production expenses	(36)		(45)	(20)%		(112)		(135)	(17)%		
Allocated to marketing expenses	(5)		(7)	(29)%		(16)		(23)	(30)%		
Capitalized	(31)		(35)	(11)%		(93)		(103)	(10)%		
Reimbursed from third parties	(37)		(46)	(20)%		(114)		(145)	(21)%		
General and administrative expenses, net	\$ 66	\$	54	22 %	\$	229	\$	189	21 %		
General and administrative expenses, net per boe	\$ 1.34	\$	1.08	24 %	\$	1.55	\$	1.30	19 %		

Gross overhead decreased primarily due to our reduction in workforce. The absolute and per unit net expense increase was primarily due to less overhead allocated to production expenses, marketing expenses and capitalized general and administrative costs, as well as less overhead billed to third party working interest owners, due to certain divestitures in 2017 and 2018.

#### Restructuring and Other Termination Costs

On January 30, 2018, we underwent a reduction in workforce impacting approximately 13% of employees across all functions, primarily on our Oklahoma City campus. In connection with the reduction, we incurred a total charge of approximately \$38 million in the Current Period for one-time termination benefits. The charge consisted of \$33 million in salary expense and \$5 million of other termination benefits.

#### Oil, Natural Gas and NGL Depreciation, Depletion and Amortization

	Three Months Ended September 30,					Nine Months Ended September 30,				
	2018 2017 Change					2018	2	2017	Change	
			(\$	in millions,	exce	pt per ι	ınit)			
Oil, natural gas and NGL depreciation, depletion and amortization	\$ 274	\$	228	20%	\$	813	\$	627	30%	
Oil, natural gas and NGL depreciation, depletion and amortization per boe	\$ 5.54	\$	4.57	21%	\$	5.51	\$	4.31	28%	

The absolute and per unit increase in the Current Quarter and the Current Period is primarily the result of a higher depletion rate per boe coupled with an increase in production. The depletion rate per boe is a function of capitalized costs, future development costs, and the related underlying reserves in the periods presented. The increase in depletion rate per boe primarily reflects a downward revision in proved reserve estimates in the fourth quarter of 2017 due to an updated development plan in the Eagle Ford aligning up-spacing, our activity schedule and well performance. The downward revision in proved reserves was partially offset by the effect of upward price revisions as a result of improved trailing 12-month oil, natural gas and NGL prices, net of differentials.

#### Depreciation and Amortization of Other Assets

	Thi	lonths E tember 3			Ni	Ended 30,			
	2018		2017	Change		2018		2017	Change
			(9	in millions,	exce	ept per ι	ınit)		
Depreciation and amortization of other assets	\$ 17	\$	20	(15)%	\$	54	\$	62	(13)%
Depreciation and amortization of other assets per boe	\$ 0.35	\$	0.41	(15)%	\$	0.37	\$	0.43	(14)%

The absolute and per unit decrease in the Current Quarter and the Current Period was primarily the result of the sale of certain other assets.

#### **Impairments**

We determined that certain of our other fixed assets will either be sold or disposed before the end of their useful lives indicating the carrying value may not be recoverable. As a result, we recognized impairment losses of \$5 million, \$3 million, \$51 million and \$3 million in the Current Quarter, Prior Quarter, Current Period and the Prior Period, respectively, for the difference between the carrying amount and fair value of the assets.

#### Other Operating (Income) Expense

			ee Mor Septer	-			Nine Months Ended September 30,				
	201	18	20:	17	Change		2018		2017	Change	
					(\$ in m	illio	ns)				
Other operating (income) expense	\$	_	\$	6	(100)%	\$	(1)	\$	423	(100)%	

In the Prior Period, we terminated future natural gas gathering transportation commitments related to divested assets for cash payments of \$126 million. In the Prior Period, we also paid \$290 million to assign an oil transportation agreement to a third party.

#### Interest Expense

	Three Mor Septer	 			nths Ended mber 30,		
	 2018	2017		2018		2017	
		(\$ in m	nillio	ns)			
Interest expense on senior notes	\$ 146	\$ 135	\$	434	\$	407	
Interest expense on term loan	29	34		87		98	
Amortization of loan discount, issuance costs and other	7	13		17		28	
Amortization of premium	(24)	(29)		(72)		(112)	
Interest expense on revolving credit facility	11	11		29		28	
Realized gains on interest rate derivatives	(1)	(1)		(2)		(3)	
Unrealized losses on interest rate derivatives	1	_		2		3	
Capitalized interest	(42)	(49)		(128)		(147)	
Total interest expense	\$ 127	\$ 114	\$	367	\$	302	
Interest expense per boe <sup>(a)</sup>	\$ 2.56	\$ 2.26	\$	2.48	\$	2.05	
Average senior notes borrowings	\$ 8,021	\$ 7,632	\$	7,985	\$	7,640	
Average credit facilities borrowings	\$ 642	\$ 631	\$	540	\$	330	
Average term loan borrowings	\$ 1,179	\$ 1,500	\$	1,215	\$	1,500	

<sup>(</sup>a) Includes the effects of realized (gains) losses from interest rate derivatives, excludes the effects of unrealized (gains) losses from interest rate derivatives and is shown net of amounts capitalized.

The increase in interest expense is primarily due to the increase in the average outstanding principal amount of senior notes and a decrease in amortization of premium and capitalized interest. The decrease in amortization of premium is due to the decrease in the average outstanding principal amount of our senior secured second lien notes. The decrease in capitalized interest is a result of lower average balances of unproved oil and natural gas properties, the primary asset on which interest is capitalized. See <a href="Note 3">Note 3</a> of the notes to our condensed consolidated financial statements included in Item 1 of this report for a discussion of our debt refinancing.

#### Gains on Investments

In the Current Period, we recognized \$139 million of gains related to our equity investment in FTSI, including the sale of a portion of that investment. See Note 11 of the notes to our condensed consolidated financial statements included in Item 1 of this report for further discussion.

### Losses on Purchases or Exchanges of Debt

In the Current Quarter, we used the proceeds from the issuance of senior notes together with cash on hand and borrowings under our revolving credit facility, to repay in full \$1.233 billion of borrowings under our secured term loan due 2021 for \$1.285 billion including a call premium of \$52 million. We recorded a loss of approximately \$65 million associated with the repayment of the term loan, including the call premium and write-off of \$13 million of associated deferred charges. Also in the Current Quarter, we recorded a loss of \$3 million associated with certain deferred charges related to the revolving credit facility prior to its amendment. See Note 3 of the notes to our condensed consolidated financial statements included in Item 1 of this report for a discussion of our debt refinancing.

In the Prior Quarter, we repurchased \$5 million principal amount of our outstanding senior notes and contingent convertible senior notes for \$6 million. We recorded an aggregate loss of approximately \$1 million associated with the transaction.

In the Prior Period, we retired \$1.609 billion principal amount of our outstanding senior notes, senior secured second lien notes and contingent convertible notes through purchases in the open market, tender offers or repayment upon maturity for \$1.751 billion, which included the maturity of our 6.25% Euro-denominated Senior Notes due 2017 and the corresponding cross currency swap. We recorded an aggregate net gain of approximately \$183 million associated with the repurchases and tender offers.

#### Other Income (Expense)

In the Current Period, we extinguished our obligation to convey future ORRIs to the CHK Utica L.L.C. investors and recognized a \$61 million gain included in other income on our condensed consolidated statement of operations. See Note 5 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for a discussion of the transaction.

#### Income Tax Expense (Benefit)

We recorded a \$1 million income tax expense in the Current Quarter and an \$8 million income tax benefit in the Current Period and recorded a nominal amount of income tax benefit in the Prior Quarter and \$2 million of income tax expense in the Prior Period. Our effective income tax rate was 1.2% for the Current Quarter and (2.3%) for the Current Period compared to 0.0% and 0.3% for the Prior Quarter and for the Prior Period, respectively. Our effective tax rate can fluctuate as a result of the impact of discrete items, state income taxes and permanent differences. For the Current Quarter, our estimated annual effective tax rate remains nominal as a result of having a full valuation allowance against our net deferred tax asset. See <a href="Note 6">Note 6</a> of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for a discussion of income tax expense.

### **Forward-Looking Statements**

This report includes "forward-looking statements" within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934 (the "Exchange Act"). Forward-looking statements include our current expectations or forecasts of future events, including matters relating to the acquisition of WildHorse, our ability to meet debt service requirements and the other items discussed in the Introduction to Item 2 of this report. In this context, forward-looking statements often address our expected future business, financial performance and financial condition, and often contain words such as "expect," "could," "may," "anticipate," "intend," "plan," "ability," "believe," "seek," "see," "will," "would," "estimate," "forecast," "target," "guidance," "outlook," "opportunity" or "strategy."

Although we believe the expectations and forecasts reflected in our forward-looking statements are reasonable, they are inherently subject to numerous risks and uncertainties, most of which are difficult to predict and many of which are beyond our control. No assurance can be given that such forward-looking statements will be correct or achieved or that the assumptions are accurate or will not change over time. Particular uncertainties that could cause our actual results to be materially different than those expressed in our forward-looking statements include:

- · the volatility of oil, natural gas and NGL prices;
- uncertainties inherent in estimating quantities of oil, natural gas and NGL reserves and projecting future rates of production and the amount and timing of development expenditures;
- · our ability to replace reserves and sustain production;
- drilling and operating risks and resulting liabilities;
- our ability to generate profits or achieve targeted results in drilling and well operations;
- the limitations our level of indebtedness may have on our financial flexibility;
- · our inability to access the capital markets on favorable terms;
- · the availability of cash flows from operations and other funds to finance reserve replacement costs or satisfy our debt obligations;

- adverse developments or losses from pending or future litigation and regulatory proceedings, including royalty claims;
- effects of environmental protection laws and regulation on our business;
- terrorist activities and/or cyber-attacks adversely impacting our operations;
- · effects and risks of acquisitions and dispositions, including the WildHorse Merger;
- effects of purchase price adjustments and indemnity obligations;
- the need to obtain certain consents and approvals and satisfy certain conditions to closing of the Utica transaction, which may not be completed in the anticipated time frame or at all;
- · failure to consummate the WildHorse acquisition; and
- other factors that are described under Risk Factors in Part II, Item 1A of this report and Item 1A of our 2017 Form 10-K.

We caution you not to place undue reliance on the forward-looking statements contained in this report, which speak only as of the filing date, and we undertake no obligation to update this information. We urge you to carefully review and consider the disclosures in this report and our other filings with the SEC that attempt to advise interested parties of the risks and factors that may affect our business.

#### ITEM 3. Quantitative and Qualitative Disclosures About Market Risk

#### Oil, Natural Gas and NGL Derivatives

Our results of operations and cash flows are impacted by changes in market prices for oil, natural gas and NGL. To mitigate a portion of our exposure to adverse price changes, we have entered into various derivative instruments. Our oil, natural gas and NGL derivative activities, when combined with our sales of oil, natural gas and NGL, allow us to predict with greater certainty the revenue we will receive. We believe our derivative instruments continue to be highly effective in achieving our risk management objectives.

Our general strategy for protecting short-term cash flow and attempting to mitigate exposure to adverse oil, natural gas and NGL price changes is to hedge into strengthening oil, natural gas and NGL futures markets when prices reach levels that management believes are unsustainable for the long term, have material downside risk in the short term or provide reasonable rates of return on our invested capital. Information we consider in forming an opinion about future prices includes general economic conditions, industrial output levels and expectations, producer breakeven cost structures, liquefied natural gas trends, oil and natural gas storage inventory levels, industry decline rates for base production and weather trends. Executive management is involved in our risk management activities and the Board of Directors reviews our derivative program at quarterly board meetings. We believe we have sufficient internal controls to prevent unauthorized trading.

We use derivative instruments to achieve our risk management objectives, including swaps, collars and options. All of these are described in more detail below. We typically use swaps and collars for a large portion of the oil and natural gas price risk we hedge. We have also sold calls, taking advantage of premiums associated with market price volatility.

We determine the notional volume potentially subject to derivative contracts by reviewing our overall estimated future production levels, which are derived from extensive examination of existing producing reserve estimates and estimates of likely production from new drilling. Production forecasts are updated at least monthly and adjusted if necessary to actual results and activity levels. We do not enter into derivative contracts for volumes in excess of our share of forecasted production, and if production estimates were lowered for future periods and derivative instruments are already executed for some volume above the new production forecasts, the positions would be reversed. The actual fixed price on our derivative instruments is derived from the reference NYMEX price, as reflected in current NYMEX trading. The pricing dates of our derivative contracts follow NYMEX futures. All of our commodity derivative instruments are net settled based on the difference between the fixed price as stated in the contract and the floating-price, resulting in a net amount due to or from the counterparty.

We review our derivative positions continuously and if future market conditions change and prices are at levels we believe could jeopardize the effectiveness of a position, we will mitigate this risk by either negotiating a cash settlement with our counterparty, restructuring the position or entering into a new trade that effectively reverses the current position. The factors we consider in closing or restructuring a position before the settlement date are identical to those we review when deciding to enter into the original derivative position. Gains or losses related to closed positions will be recognized in the month specified in the original contract.

We have determined the fair value of our derivative instruments utilizing established index prices, volatility curves and discount factors. These estimates are compared to counterparty valuations for reasonableness. Derivative transactions are also subject to the risk that counterparties will be unable to meet their obligations. This non-performance risk is considered in the valuation of our derivative instruments, but to date has not had a material impact on the values of our derivatives. Future risk related to counterparties not being able to meet their obligations has been partially mitigated under our commodity hedging arrangements that require counterparties to post collateral if their obligations to us are in excess of defined thresholds. The values we report in our financial statements are as of a point in time and subsequently change as these estimates are revised to reflect actual results, changes in market conditions and other factors. See Note 8 of the notes to our condensed consolidated financial statements included in Item 1 of Part I of this report for further discussion of the fair value measurements associated with our derivatives.

As of September 30, 2018, our oil, natural gas and NGL derivative instruments consisted of the following types of instruments:

- Swaps: We receive a fixed price and pay a floating market price to the counterparty for the hedged commodity. In exchange for higher fixed prices on certain of our swap trades, we may sell call options and call swaptions.
- *Options*: We sell, and occasionally buy, call options in exchange for a premium. At the time of settlement, if the market price exceeds the fixed price of the call option, we pay the counterparty the excess on sold call options, and we receive the excess on bought call options. If the market price settles below the fixed price of the call option, no payment is due from either party.
- Call Swaptions: We sell call swaptions to counterparties that allow the counterparty, on a specific date, to extend an existing fixed-price swap for a certain period of time
- Collars: These instruments contain a fixed floor price (put) and ceiling price (call). If the market price exceeds the call strike price or
  falls below the put strike price, we receive the fixed price and pay the market price. If the market price is between the put and the call
  strike prices, no payments are due from either party. Three-way collars include the sale by us of an additional put option in exchange
  for a more favorable strike price on the call option. This eliminates the counterparty's downside exposure below the second put
  option strike price.
- Basis Protection Swaps: These instruments are arrangements that guarantee a fixed price differential to NYMEX from a specified delivery point. We receive the fixed price differential and pay the floating market price differential to the counterparty for the hedged commodity.

As of September 30, 2018, we had the following open oil, natural gas and NGL derivative instruments:

			Weighted Average Price									Fair Value	
		Volume		Fixed		Call		Put	I	Differential		Asset (Liability)	
	_	(mmbbl)					(\$ pe	er bbl)			(	(\$ in millions)	
Oil:													
Swaps:													
Short-term		18	\$	57.43	\$	_	\$	_	\$	_	\$	(265)	
Long-term		3	\$	59.96	\$	_	\$	_	\$	_		(29)	
Three-Way Collars:													
Short-term		_	\$	_	\$	55.00		39.15/47.00	\$	_		(8)	
Basis Protection Sv	vaps:												
Short-term		7	\$	_	\$	_	\$	_	\$	4.75		_	
Long-term		1	\$	_	\$	_	\$	_	\$	6.20		2	
	Total Oil											(300)	
	_	(bcf)					(\$ pe	er mcf)					
Natural Gas:													
Swaps <sup>(a)</sup> :													
Short-term <sup>(b)</sup>		372	\$	2.79	\$		\$	_	\$	_		(19)	
Long-term		28	\$	2.77	\$	_	\$	_	\$	_		_	
Three-Way Collars:													
Short-term		66	\$	_	\$	3.10		2.50/2.80		_		3	
Long-term		22	\$	_	\$	3.10		2.50/2.80		_			
Collars:													
Short-term		57	\$	_	\$	3.09	\$	2.80	\$	_			
Long-term		9	\$	_	\$	2.91	\$	2.75	\$	_		_	
Call Options (sold):													
Short-term		33	\$	_	\$	9.12	\$	_	\$	_		_	
Long-term		27	\$	_	\$	12.00	\$	_	\$	_			
Basis Protection Sv	vaps:												
Short-term		38	\$	_	\$	_	\$	_	\$	(0.02)		(7)	
Long-term		6	\$	_	\$	_	\$	_	\$	(0.39)		_	
	Total Natural Gas	3										(23)	

			Fair Value						
	Volume	 Fixed	Call Put Differential		Asset (Liability)				
	(mmgal)				(\$ pe	r gal)			
NGL:									
Propane Swaps:									
Short-term	16	\$ 0.79	\$	_	\$		_	\$ _	(4)
Butane Swaps:									
Short-term	1	\$ 0.88	\$	_	\$		_	\$ _	(1)
Short-term % of WTI	1	70.50%	\$	_	\$		_	\$ _	_
Isobutane Swaps:									
Short-term	4	\$ 0.92	\$	_	\$		_	\$ _	(1)
Ethane Swaps:									
Short-term	23	\$ 0.29	\$	_	\$		_	\$ _	(6)
Natural Gasoline Swaps:									
Short-term	12	\$ 1.42	\$	_	\$		_	\$ _	(3)
Total NGL									(15)
			Tota	l Estimate	d Fair	Value			\$ (338)

<sup>(</sup>a) This amount includes a sold option to enhance the swap price at an average price of \$3.40/mcf covering 11 bcf, included in the sold call options.

In addition to the open derivative positions disclosed above, as of September 30, 2018, we had \$63 million of net derivative losses related to settled contracts for future periods that will be recorded within oil, natural gas and NGL revenues as realized gains (losses) on derivatives once they are transferred from either accumulated other comprehensive income or unrealized gains (losses) on derivatives in the month specified in the original contract as noted below:

	September 30, 2018
	 (\$ in millions)
Short-term	\$ (24)
Long-term	(39)
Total	\$ (63)

The table below reconciles the changes in fair value of our oil and natural gas derivatives during the Current Period. Of the \$338 million fair value liability as of September 30, 2018, a \$310 million liability relates to contracts maturing in the next 12 months and a \$28 million liability relates to contracts maturing after 12 months. All open derivative instruments as of September 30, 2018 are expected to mature by December 31, 2020.

	S	eptember 30, 2018
	(\$	\$ in millions)
Fair value of contracts outstanding, as of January 1, 2018	\$	(35)
Change in fair value of contracts		(117)
Contracts realized or otherwise settled		(186)
Fair value of contracts outstanding, as of September 30, 2018	\$	(338)

<sup>(</sup>b) Includes 170 bcf related to trades executed in accordance with the purchase and sale agreement with Encino. These trades are reflected at fair market value as of September 30, 2018, with an offsetting receivable balance. The trades were novated to Encino upon closing of the purchase and sale agreement on October 29, 2018.

#### Interest Rate Risk

The table below presents principal cash flows and related weighted average interest rates by expected maturity dates, using the earliest demand repurchase date for contingent convertible senior notes.

			Years (	of M	laturity				
	 2018	2019	2020		2021		2022	Thereafter	Total
				(:	\$ in millio	ns)			
Liabilities:									
Debt – fixed rate	\$ 53	\$ _	\$ 664	\$	815	\$	1,867	\$ 5,438	\$ 8,837
Average interest rate	6.42%	%	6.71%		5.88%		7.25%	7.09%	6.98%
Debt – variable rate	\$ _	\$ 380	\$ _	\$	_	\$	_	\$ 645	\$ 1,025
Average interest rate	%	5.59%	%		%		%	4.21%	4.72%

Changes in interest rates affect the amount of interest we earn on our cash, cash equivalents and short-term investments and the interest rate we pay on borrowings under our revolving credit facility and our floating rate senior notes. All of our other indebtedness is fixed rate and, therefore, does not expose us to the risk of fluctuations in earnings or cash flow due to changes in market interest rates. However, changes in interest rates do affect the fair value of our fixed-rate debt.

As of September 30, 2018, we had \$5 million of net gains related to settled interest rate derivative contracts that will be recorded within interest expense as realized gains or losses once they are transferred from our senior note liability or within interest expense as unrealized gains or losses over the remaining six-year term of our related senior notes.

Realized and unrealized (gains) or losses from interest rate derivative transactions are reflected as adjustments to interest expense on the consolidated statements of operations.

#### ITEM 4. Controls and Procedures

#### Evaluation of Disclosure Controls and Procedures

We maintain disclosure controls and procedures designed to ensure that information required to be disclosed in reports we file or submit under the Securities Exchange Act of 1934, as amended (the "Exchange Act"), is recorded, processed, summarized and reported within the time periods specified in SEC rules and forms, and that such information is accumulated and communicated to management, including our principal executive and principal financial officers, as appropriate, to allow timely decisions regarding required disclosure.

As of the end of the period covered by this report, we carried out an evaluation, under the supervision and with the participation of management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures pursuant to Exchange Act Rule 13a-15(b). Based on that evaluation, our Chief Executive Officer and Chief Financial Officer concluded as of September 30, 2018 that our disclosure controls and procedures were effective.

### Changes in Internal Control Over Financial Reporting

There were no changes in our internal control over financial reporting that occurred during the quarter ended September 30, 2018 that have materially affected, or are reasonably likely to materially affect, our internal control over financial reporting.

#### PART II. OTHER INFORMATION

#### ITEM 1. Legal Proceedings

There have been no material developments in previously reported legal or environmental proceedings. For a description of certain legal and regulatory proceedings affecting us, see "Contingencies and Commitments," Note 4 to the Consolidated Financial Statements included in Item 1 of Part 1 of this report and Item 3 in our 2017 Form 10-K and in Note 4 to the Consolidated Financial Statements in our Form 10-Q for the quarterly periods ended March 31, 2018 and June 30, 2018.

#### ITEM 1A. Risk Factors

Our business has many risks. Factors that could materially adversely affect our business, financial condition, operating results or liquidity and the trading price of our common stock, preferred stock or senior notes are described under "Risk Factors" in Item 1A of our 2017 Form 10-K. This information should be considered carefully, together with other information in this report and other reports and materials we file with the SEC.

Our acquisition of WildHorse is subject to conditions, including certain conditions that may not be satisfied, or completed on a timely basis, if at all. Failure to complete the acquisition of WildHorse could have a material and adverse effect on us.

Completion of our acquisition of WildHorse is subject to a number of conditions set forth in our merger agreement with WildHorse, including the approval by our shareholders of the issuance of shares of our common stock as acquisition consideration and approval by WildHorse stockholders of the adoption of the merger agreement, which make the completion and timing of the completion of the transactions uncertain. Also, either we or WildHorse may terminate the merger agreement if the WildHorse Merger has not been consummated by May 31, 2019 (or, at either party's discretion, if any governmental entity having jurisdiction over either us or WildHorse has issued any order, decree, ruling or injunction or taken any other action permanently restraining, enjoining or otherwise prohibiting the consummation of the WildHorse Merger).

If the transactions contemplated by the merger agreement are not completed, our ongoing business may be adversely affected and, without realizing any of the benefits of having completed the transactions, we will be subject to a number of risks, including the following:

- we will be required to pay our costs relating to the transactions, such as legal, accounting, financial advisory and printing fees, whether or not the transactions are completed;
- time and resources committed by our management to matters relating to the transactions could otherwise have been devoted to pursuing other beneficial opportunities, and our ongoing business and financial results may be adversely affected;
- the market price of our common stock could decline to the extent that the current market price reflects a market assumption that the transactions will be completed;
- being required to pay a termination fee or expense reimbursement fee of \$120 million or \$35 million, respectively, under certain circumstances provided in the merger agreement;
- the manner in which customers, vendors, business partners and other third parties perceive us may be negatively impacted, which in turn could affect our marketing operations or our ability to compete for new business or obtain renewals in the marketplace more broadly;
- · we may experience negative reactions from employees; and
- if the merger agreement is terminated and our board of directors seeks another acquisition, our shareholders cannot be certain that we will be able to find a party willing to enter into a transaction as attractive to us as the acquisition of WildHorse.

The WildHorse Merger may not be accretive, and may be dilutive, to our earnings per share, which may negatively affect the market price of shares on our common stock.

Because shares of our common stock will be issued in the WildHorse Merger, it is possible that the WildHorse Merger may be dilutive to our earnings per share, which could negatively affect the market price of shares of our common stock.

In connection with the completion of the WildHorse Merger, the two largest shareholders of WildHorse, who together own approximately 70% of WildHorse common stock, have agreed to accept consideration of \$3.00 per share of cash and 5.336 shares of Chesapeake common stock in exchange for each share of WildHorse common stock. The other WildHorse shareholders will have a choice of receiving a combination of \$3.00 per share in cash and 5.336 shares of Chesapeake common stock, or 5.989 shares of Chesapeake common stock, for each share of WildHorse common stock. As a result, based on the number of issued and outstanding shares of WildHorse common stock as of October 29, 2018, we will issue up to 745 million shares of our common stock. The issuance of these new shares of

our common stock could have the effect of depressing the market price of shares of our common stock, through dilution of earnings per share or otherwise. Any dilution of, or delay of any accretion to, our earnings per share could cause the price of shares of our common stock to decline or increase at a reduced rate.

#### Our stockholders will be diluted by the WildHorse Merger.

The WildHorse Merger will dilute the ownership position of our current stockholders. As a result of issuances in connection with the WildHorse Merger, current stockholders of ours and WildHorse stockholders are expected to hold approximately 55% and 45%, respectively, of the combined company's outstanding common stock immediately following completion of the WildHorse Merger, depending on the consideration elected.

The market price of shares of our common stock may decline in the future as a result of the sale of shares of our common stock held by former WildHorse stockholders or current stockholders.

Following their receipt of shares of our common stock as acquisition consideration in the WildHorse Merger, former WildHorse stockholders may, following 60 and 180 day lock-up periods for certain primary former stockholders, seek to sell the shares of our common stock delivered to them following the consummation of the WildHorse Merger. Other shareholders may also seek to sell shares of our common stock held by them following, or in anticipation of, completion of the WildHorse Merger. In addition, we have granted certain stockholders of WildHorse registration rights with respect to the shares of our common stock they receive in the WildHorse Merger. These sales (or the perception that these sales may occur), coupled with the increase in the outstanding number of shares of our common stock, may affect the market for, and the market price of, our common stock in an adverse manner.

We and WildHorse will be subject to business uncertainties while the WildHorse Merger is pending, which could adversely affect our respective businesses.

Uncertainty about the effect of the WildHorse Merger on employees and customers may have an adverse effect on us and WildHorse. These uncertainties may impair our and WildHorse's ability to attract, retain and motivate key personnel until the WildHorse Merger is completed and for a period of time thereafter and could cause customers and others that deal with us and WildHorse to seek to change their existing business relationships with us and WildHorse, respectively. Employee retention at WildHorse may be particularly challenging during the pendency of the WildHorse Merger, as employees may experience uncertainty about their roles with us following the WildHorse Merger. In addition, the merger agreement restricts us and WildHorse from entering into certain corporate transactions and taking other specified actions without the consent of the other party, and generally requires each party to continue its operations in the ordinary course of business, until completion of the WildHorse Merger. These restrictions may prevent us and WildHorse from pursuing attractive business opportunities that may arise prior to the completion of the WildHorse Merger.

#### We have substantial indebtedness, and following the WildHorse Merger, will continue to have substantial indebtedness.

At September 30, 2018, we had approximately \$9.812 billion of indebtedness and at June 30, 2018, WildHorse had approximately \$930 million of outstanding indebtedness. We continue to review the treatment of our and WildHorse's existing indebtedness, and we may seek to repay, refinance, repurchase, redeem, exchange or otherwise terminatea portion of our or WildHorse's existing indebtedness prior to, in connection with or following the completion of the WildHorse Merger. If we do seek to refinance our or WildHorse's existing indebtedness, there can be no guarantee that we would be able to execute the refinancing on favorable terms or at all.

Any increase in our level of indebtedness could have adverse effects on our financial condition and results of operations, including:

- imposing additional cash requirements on us in order to support interest payments, which reduces the amount we have available to fund our operations and other business activities;
- increasing the risk that we may default on our debt obligations;
- increasing our vulnerability to adverse changes in general economic and industry conditions, economic downturns and adverse developments in our business;
- limiting our ability to sell assets, engage in strategic transactions or obtain additional financing for working capital, capital expenditures, general corporate and other purposes;

- limiting our flexibility in planning for or reacting to changes in our business and the industry in which we operate; and
- increasing our exposure to a rise in interest rates, which will generate greater interest expense to the extent we do not have applicable interest rate fluctuation hedges.

# Even if we and WildHorse complete the WildHorse Merger, we may fail to realize all of the anticipated benefits of the WildHorse Merger.

The success of the WildHorse Merger will depend, in part, on our ability to realize the anticipated benefits and cost savings from combining our and WildHorse's businesses, including operational and other synergies that we believe the combined company will achieve. The anticipated benefits and cost savings of the WildHorse Merger may not be realized fully or at all, may take longer to realize than expected or could have other adverse effects that we do not currently foresee. Some of the assumptions that we have made, such as the achievement of operating synergies, may not be realized. The integration process may, for us and WildHorse, result in the loss of key employees, the disruption of ongoing businesses or inconsistencies in standards, controls, procedures and policies. There could be potential unknown liabilities and unforeseen expenses associated with the WildHorse Merger that were not discovered in the course of performing due diligence. The integration will require significant time and focus from management following the acquisition.

#### We and WildHorse will incur substantial transaction fees and costs in connection with the WildHorse Merger.

We and WildHorse expect to incur a number of non-recurring transaction-related costs associated with completing the WildHorse Merger, combining the operations of the two organizations and achieving desired synergies. These fees and costs will be substantial. Non-recurring transaction costs include, but are not limited to, fees paid to legal, financial and accounting advisors, filing fees and printing costs. Additional unanticipated costs may be incurred in the integration of WildHorse's business with our business. There can be no assurance that the elimination of certain duplicative costs, as well as the realization of other efficiencies related to the integration of the two businesses, will offset the incremental transaction-related costs over time. Thus, any net benefit of the WildHorse Merger may not be achieved in the near term, the long term or at all.

#### The exchange ratio is fixed and will not be adjusted in the event of any change in either our or WildHorse's stock price.

At the effective time, each share of WildHorse common stock outstanding immediately prior to the effective time will be converted into the right to receive 5.989 shares of our common stock or 5.336 shares of our common stock and \$3.00, at the election of the holder. This exchange ratio will not be adjusted for changes in the market price of either our common stock or WildHorse common stock between the date of signing the merger agreement and completion of the WildHorse Merger. Changes in the price of our common stock prior to the WildHorse Merger will affect the value of our common stock that WildHorse common stockholders will receive on the date of the WildHorse Merger.

The prices of our common stock and WildHorse common stock at the closing of the WildHorse Merger may vary from their prices on the date the merger agreement was executed and on the date of each special meeting. As a result, the value represented by the exchange ratio will also vary, and you will not know or be able to calculate the market value of the merger consideration you will receive upon completion of the WildHorse Merger.

# We and WildHorse may be targets of securities class action and derivative lawsuits which could result in substantial costs and may delay or prevent the WildHorse Merger from being completed.

Securities class action lawsuits and derivative lawsuits are often brought against public companies that have entered into merger agreements. Even if the lawsuits are without merit, defending against these claims can result in substantial costs and divert management time and resources. An adverse judgment could result in monetary damages, which could have a negative impact on our and WildHorse's respective liquidity and financial condition. Additionally, if a plaintiff is successful in obtaining an injunction prohibiting completion of the WildHorse Merger, then that injunction may delay or prevent the WildHorse Merger from being completed, which may adversely affect our and WildHorse's respective business, financial position and results of operations.

The issuance of our common stock to shareholders of WildHorse as well as other stock transactions can lead to an ownership change under Section 382 of the Internal Revenue Code.

Our ability to utilize U.S. net operating loss carryforwards to reduce future taxable income is subject to various limitations under the Internal Revenue Code of 1986, as amended (the "Code"). The utilization of such carryforwards may be limited under Section 382 of the Code upon the occurrence of ownership changes resulting from issuances of our stock or the sale or exchange of our stock by certain shareholders if, as a result, there is a cumulative change of more than 50% in the beneficial ownership of our stock during any three-year period. For this purpose, "stock" includes certain preferred stock. In the event of such an ownership change, Section 382 of the Code imposes an annual limitation on the amount of our loss carryforwards that can be used to offset taxable income. The limitation is generally equal to the product of (a) the fair market value of our equity multiplied by (b) the long-term tax-exempt rate in effect for the month in which an ownership change occurs. In addition, if we are in a net unrealized built-in gain position at the time of an ownership change, then the limitation is increased if there are recognized built-in gains during any post-change year, but only to the extent of any net unrealized built-in gains inherent in the assets sold. If we are in a net unrealized built-in loss position at the time of an ownership change, then the limitation may apply to tax attributes other than just loss carryforwards, such as depreciable basis. Some states impose similar limitations on tax attribute utilization upon experiencing an ownership change. We do not believe we have a Section 382 limitation on the ability to utilize our U.S. loss carryforwards as of September 30, 2018. Further, we do not expect an ownership change to occur as a result of the WildHorse Merger based on information known today. However, issuances, sales and/or exchanges of our stock (including, potentially, relatively small transactions and transactions beyond our control) occurring after September 30, 2018, taken together with prior transactions with respect to our stock and the WildHorse Merger, could trigger an ownership change under Section 382 of the Code and therefore a limitation on our ability to utilize our U.S. loss carryforwards. Any such limitation could cause some of such loss carryforwards to expire before we would be able to utilize them to reduce taxable income in future periods, possibly resulting in a substantial income tax expense or write down of our tax assets or both.

#### ITEM 2. Unregistered Sales of Equity Securities and Use of Proceeds

The following table presents information about repurchases of our common stock during the quarter ended September 30, 2018:

Period	Total Number of Shares Purchased <sup>(a)</sup>	verage Price Paid Per Share <sup>(a)</sup>	Total Number of Shares Purchased as Part of Publicly Announced Plans or Programs	Maximum Approximate Dollar Value of Shares That May Yet Be Purchased Under the Plans or Programs <sup>(b)</sup>
				(\$ in millions)
July 1, 2018 through July 31, 2018	9,376	\$ 5.32	_	\$ 1,000
August 1, 2018 through August 31, 2018	_	\$ _	_	\$ 1,000
September 1, 2018 through September 30, 2018	_	\$ _	_	\$ 1,000
Total	9,376	\$ _		

a) Includes shares of common stock purchased on behalf of our deferred compensation plan.

### ITEM 3. Defaults Upon Senior Securities

None.

#### ITEM 4. Mine Safety Disclosures

Not applicable.

<sup>(</sup>b) In December 2014, our Board of Directors authorized the repurchase of up to \$1 billion of our common stock from time to time. The repurchase program does not have an expiration date. As of September 30, 2018, there have been no repurchases under the program.

## ITEM 5. Other Information

Not applicable.

#### ITEM 6. Exhibits

The exhibits listed below in the Index of Exhibits are filed, furnished or incorporated by reference pursuant to the requirements of Item 601 of Regulation S-K.

### **INDEX OF EXHIBITS**

Exhibit Number	Exhibit Description	Form	SEC File Number	Exhibit	Filing Date	Filed or Furnished Herewith
2.1	Purchase and Sale Agreement by and among certain subsidiaries of Chesapeake Energy Corporation and EAP Ohio, LLC dated July 26, 2018.					Х
3.1.1	Chesapeake's Restated Certificate of Incorporation.	10-Q	001-13726	3.1.1	8/3/2017	
3.1.2	Certificate of Designation of 5% Cumulative Convertible Preferred Stock (Series 2005B), as amended.	10-Q	001-13726	3.1.4	11/10/2008	
3.1.3	Certificate of Designation of 4.5% Cumulative Convertible Preferred Stock, as amended.	10-Q	001-13726	3.1.6	8/11/2008	
3.1.4	<u>Certificate of Designation of 5.75% Cumulative Non-Voting Convertible Preferred Stock (Series A).</u>	8-K	001-13726	3.2	5/20/2010	
3.1.5	<u>Certificate of Designation of 5.75% Cumulative Non-Voting Convertible Preferred Stock, as amended.</u>	10-Q	001-13726	3.1.5	8/9/2010	
3.2	Chesapeake's Amended and Restated Bylaws.	8-K	001-13726	3.2	6/19/2014	
4.1	Indenture dated as of April 24, 2014, among Chesapeake Energy Corporation, the subsidiary guarantors named therein and Deutsche Bank Trust Company Americas, as trustee.	8-K	001-13726	4.1	4/29/2014	
4.2	Form of Eighth Supplemental Indenture, dated as of September 27, 2018.	8-K	001-13726	4.2	9/27/2018	
4.3	Form of Ninth Supplemental Indenture, dated as of September 27, 2018.	8-K	001-13726	4.3	9/27/2018	
4.4	Form of 7.00% Senior Note due 2024 (included as Exhibit A to Exhibit 4.2.	8-K	001-13726	4.4	9/27/2018	
4.5	Form of 7.50% Senior Note due 2026 (included as Exhibit A to Exhibit 4.3.	8-K	001-13726	4.5	9/27/2018	
10.1	Amended and Restated Credit Agreement, dated as of September 12, 2018, by and among: (i) the Company, as borrower; (ii) MUFG Union Bank N.A., as the administrative agent, a swingline lender and a letter of credit issuer; (iii) Wells Fargo Bank, National Association, as co-syndication agent, a swingline lender and a letter of credit issuer; (iv) JPMorgan Chase Bank, N.A., as co-syndication agent, a swingline lender and a letter of credit issuer; and (v) certain other lenders and letter of credit issuers named therein.	8-K	001-13726	10.1	9/12/2018	
31.1	Robert D. Lawler, President and Chief Executive Officer, Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.					Χ
31.2	<u>Domenic J. Dell'Osso, Jr., Executive Vice President and Chief Financial Officer, Certification pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.</u>					Х
32.1	Robert D. Lawler, President and Chief Executive Officer, Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.					Х
32.2	<u>Domenic J. Dell'Osso, Jr., Executive Vice President and Chief Financial Officer, Certification pursuant to Section 906 of the Sarbanes-Oxley Act of 2002.</u>					Х

101 INS	XBRL Instance Document.	X
101 SCH	XBRL Taxonomy Extension Schema Document.	X
101 CAL	XBRL Taxonomy Extension Calculation Linkbase Document.	Χ
101 DEF	XBRL Taxonomy Extension Definition Linkbase Document.	Χ
101 LAB	XBRL Taxonomy Extension Labels Linkbase Document.	X
101 PRE	XBRL Taxonomy Extension Presentation Linkbase Document.	X

### **Signatures**

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

#### **CHESAPEAKE ENERGY CORPORATION**

Date: October 30, 2018 By: <u>/s/ ROBERT D. LAWLER</u>

Robert D. Lawler

President and Chief Executive Officer

Date: October 30, 2018 By: <u>/s/ DOMENIC J. DELL'OSSO, JR.</u>

Domenic J. Dell'Osso, Jr. Executive Vice President and Chief Financial Officer

#### PURCHASE AND SALE AGREEMENT

by and among

CHESAPEAKE EXPLORATION, L.L.C.;
CHESAPEAKE APPALACHIA, L.L.C.;
CHK UTICA, L.L.C.;
MC MINERAL COMPANY, L.L.C.;
CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C.;
CHESAPEAKE ENERGY MARKETING, L.L.C.;
CHESAPEAKE ROYALTY, L.L.C.; and
MIDCON COMPRESSION, L.L.C.

(collectively, the "Sellers")

and

**EAP OHIO, LLC** 

(the "Buyer")

**JULY 26, 2018** 

## **TABLE OF CONTENTS**

			Page
1.	Definitio	ns and References.	1
	1.1	Definitions.	1
	1.2	References.	27
2.	Purchase and Sale; Purchase Price.		
	2.1	Title and Environmental Defects.	28
	2.2	Defect Notice.	30
	2.3	Special Warranty.	37
	2.4	Preferential Purchase Rights; Required Consents.	37
	2.5	Gas Imbalances.	39
	2.6	Certain Upward Adjustments.	40
	2.7	Certain Downward Adjustments.	41
	2.8	Closing Date Estimates.	42
	2.9	Final Accounting.	42
	2.10	Payments.	43
	2.11	Tax Allocation.	43
	2.12	Casualty Loss.	44
	2.13	Additional Consideration.	44
3.	Sellers' R	Representations and Warranties.	45
	3.1	Organization; Good Standing, Etc.	45
	3.2	Authorization.	45
	3.3	No Breach.	45
	3.4	Litigation.	46
	3.5	Taxes.	46
	3.6	Permits.	47
	3.7	Compliance with Laws.	47
	3.8	Material Contracts.	47
	3.9	Environmental and Safety Matters.	49
	3.10	Broker's or Finder's Fees.	49
	3.11	Bankruptcy.	50
	3.12	PPRs and Required Consents.	50
	3.13	Gas Balancing.	50
	3.14	Hedges	50
	3.15	Current Commitments.	50
	3.16	Unleased AR	50
	3.17	Leases	51
	3.18	Wells; Equipment.	51
	3.19	Operations; Take or Pay.	51
	3.20	Suspended Funds.	51
	3.21	Payout Balances.	51
	3.22	Governmental Consent.	52
	3.23	Letters of Credit, Bonds and Guarantees.	52
	3.24	Absence of Certain Changes.	52
4.	Buyer's F	Representations and Warranties.	52

	4.1	Organization and Standing.	52		
	4.2	Powers.			
	4.3	No Restriction.	52		
	4.4	Authorization.	52		
	4.5	Non-Contravention.	53		
	4.6	Governmental Consent.	53		
	4.7	Litigation, Etc.	53		
	4.8	Broker's or Finder's Fees.	53		
	4.9	Bankruptcy.	53		
	4.10	Qualifications.			
	4.11	Funding; Investment.			
5.	Covenants.				
	5.1	Access to Information.	55		
	5.2	Conduct of Business.	57		
	5.3	Lease Extensions.	61		
	5.4	Non-Operated Properties.	61		
	5.5	Revenues.	61		
	5.6	Tax Matters.	62		
	5.7	Suspended Funds.	63		
	5.8	Operatorship.	64		
	5.9	Limitations on Representations and Warranties.	64		
	5.10	0 NORM, WASTES AND OTHER SUBSTANCES.			
	5.11	Governmental Approvals.	67		
	5.12	CFIUS.	68		
	5.13	Midstream Agreements.	69		
	5.14	Hedges and Novation Agreements.	71		
	5.15	Total Development Agreement.	73		
	5.16	Completion Contracts.	73		
	5.17	Financing Cooperation.	73		
	5.18	Drilling Rig Contracts.	75		
	5.19	FCC Licenses.	75		
	5.20	Production True-Up Payments.	76		
	5.21	Environmental Records.	77		
6.	Buyer's C	onditions Precedent.	77		
7.	Seller's C	ller's Conditions Precedent.			
8.	The Closi	ng.	80		
	8.1	Buyer's Deliveries.	81		
	8.2	Sellers' Deliveries.	82		
	8.3	Post-Closing Adjustments.	83		
	8.4	Post-Closing Deliveries.	83		
9.	Press Rel	eases.	83		
10.	Indemnification.				
	10.1	Assumed Obligations.	83		
	10.2	Sellers' Indemnification.	84		
	10.3	Buyer's Indemnification.	84		

	10.4	Indemnification Procedure.	84
	10.5	Defense.	85
	10.6	Certain Limitations of Indemnity Obligations.	85
	10.7	EXTENT OF INDEMNIFICATION.	88
11.	Preservation of Books and Records; Financial Statements.		
	11.1	Preservation of Books and Records.	88
	11.2	Financial Statements.	89
12.	Termina	tion.	91
	12.1	Right to Terminate.	91
	12.2	Effect of Termination.	92
	12.3	Distribution of Deposit Upon Termination.	93
	12.4	Buyer Indemnification Support.	95
13.	Miscellaneous.		95
	13.1	Time.	95
	13.2	Notices.	95
	13.3	Survival.	96
	13.4	Cooperation.	97
	13.5	No Third Party Beneficiaries.	97
	13.6	Cumulative Remedies.	97
	13.7	CHOICE OF LAW; VENUE.	98
	13.8	JURISDICTION.	98
	13.9	Entire Agreement.	99
	13.10	Assignment.	99
	13.11	Amendment.	99
	13.12	Severability.	100
	13.13	Attorney Fees.	100
	13.14	Counterparts; Electronic Transmission.	100
	13.15	LIMITATION ON DAMAGES.	100
	13.16	WAIVER OF JURY TRIAL.	100
	13.17	Mutuality.	101
	13.18	Schedules.	101
	13.19	Possible Exchange.	101
	13.20	Confidentiality.	101
	13.21	Sale or Use Tax, Recording Fees, and Similar Taxes and Fees.	102
	13.22	Expenses.	102
	13.23	Change of Name.	103
	13.24	Replacement of Bonds, Letters of Credit and Guarantees.	103
	13.25	CONSPICUOUS.	103
	13.26	Specific Performance.	103
	13.27	Seller Liability.	103
	13.28	No Recourse.	104
	13 20	Waiver of Claims Against Debt Financing Sources	104

## **EXHIBITS AND SCHEDULES**

# Exhibits:

Exhibits:			
	Exhibit A-1	_	Real Property Interests and Wells
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit A-2	_	Mineral Interests and Royalties
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit A-3	_	Surface Assets
	Exhibit A-4	_	Vehicles and ATVs
	Exhibit A-5	_	Excluded Assets
	Exhibit A-6	_	MidCon Assets
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit B-1	_	Form of Assignment, Bill o Sale and Conveyance
	Exhibit B-2	_	Form of Mineral Deed
	Exhibit B-3	_	Form of Surface Deed
	Exhibit B-4	_	Form of Certificate of Non-Foreign Status
	Exhibit B-5	_	Form of Transition Services Agreement
	Exhibit B-6	_	Form of Core Access Agreement
	Exhibit B-7	_	Form of CHK Parent Guaranty
	Exhibit B-8	_	Form of Oil Purchase Agreement
	Exhibit B-9	_	Form of Gas Purchase Agreement
	Exhibit B-10	_	Form of NGLs Purchase Agreement
	Exhibit C	_	Contracts
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit D	_	Midstream Agreements Subject to Certain Consents
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit E	_	Midstream Agreements
		Part 1	<ul> <li>Total AMI Properties</li> </ul>
		Part 2	<ul> <li>Non-AMI Properties</li> </ul>
	Exhibit F	_	Well Log
Schedules	<u>s</u> :		
	Schedule 1.1(a)		Persons with Knowledge
	Schedule 1.1(b)		CAA Retained Liability
	Schedule 2.4.4		Total Pref Right
	Schedule 3.3		No Breach
	Schedule 3.4		Litigation
	Schedule 3.5		Taxes
	Schedule 3.8		Contracts

Schedule 3.9		<b>Environmental and Safety Matters</b>
Schedule 3.12		PPRs and Required Consents
Schedule 3.13		Gas Balancing
Schedule 3.15		AFEs
Schedule 3.16		Unleased AR
Schedule 3.18		Wells
Schedule 3.20		Suspended Funds
Schedule 3.21		Payout Balances
Schedule 3.23		Letters of Credit, Bonds, Guarantees
Schedule 5.2		Conduct of Business
Schedule 5.3		Lease Extensions
Schedule 5.14.	1	Natural Gas Hedges
	Part 1	<ul> <li>Total AMI Natural Gas Hedges</li> </ul>
	Part 2	<ul> <li>Non-AMI Natural Gas Hedges</li> </ul>
Schedule 5.20		True-Up Quantity
Schedule 13.24	ļ.	Letters of Credit, Bonds, Guarantees

### PURCHASE AND SALE AGREEMENT

### PURCHASE AND SALE AGREEMENT

THIS PURCHASE AND SALE AGREEMENT (this "Agreement") is entered into this 26th day of July 2018 (the "Execution Date"), by and among CHESAPEAKE EXPLORATION, L.L.C., an Oklahoma limited liability company ("CELLC"), CHESAPEAKE APPALACHIA, L.L.C., an Oklahoma limited liability company ("CHK App"), CHK UTICA, L.L.C., a Delaware limited liability company ("CHK Utica"), MC MINERAL COMPANY, L.L.C., an Oklahoma limited liability company ("MCMC"), CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C., an Oklahoma limited liability company ("CLDC"), CHESAPEAKE ENERGY MARKETING, L.L.C., an Oklahoma limited liability company ("CHK Royalty"), MIDCON COMPRESSION, L.L.C., an Oklahoma limited liability company ("MidCon" and together with CELLC, CHK App, CHK Utica, MCMC, CLDC, CEM and CHK Royalty, each, a "Seller" and, collectively, the "Sellers") and EAP OHIO, LLC, a Delaware limited liability company (the "Buyer"). The Buyer and the Sellers may be referred to herein collectively as the "Parties", or individually as a "Party".

#### **BACKGROUND:**

- A. The Sellers desire to sell and the Buyer desires to purchase all of the Sellers' right, title and interest in and to the Properties (as hereinafter defined).
  - B. The purchase and sale of the Properties will be consummated on the terms and conditions set forth in this Agreement.

NOW, THEREFORE, in consideration of the mutual covenants herein contained and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

# ARTICLE 1 DEFINITIONS AND REFERENCES

- 1. <u>Definitions and References</u>. Capitalized terms used throughout this Agreement including the Recitals above and not defined in <u>Section 1.1</u> below shall have the meaning ascribed to them elsewhere in this Agreement.
  - 1.1 <u>Definitions</u>. Each of the following terms has the meaning given in this <u>Section 1.1</u> or in the Section referred to below:
    - "2022 Additional Consideration" has the meaning specified in Section 2.13.
    - "2022 NYMEX Price" means the arithmetic average of the twelve (12) monthly NYMEX Gas Contract Prices for the year 2022.
    - "2023 Additional Consideration" has the meaning specified in Section 2.13.
    - "2023 NYMEX Price" means the arithmetic average of the twelve (12) monthly NYMEX Gas Contract Prices for the year 2023.

- "Accounting Referee" means a nationally recognized accounting firm mutually agreed upon by the Parties, <u>provided</u> that prior to the appointment of any Accounting Referee, each Party will certify in writing to the other Party that such Accounting Referee has neither performed any work for such Party or its Affiliates nor been an officer, director or employee of such Party or any of its Affiliates within the preceding two (2) year period.
- "AFEs" has the meaning specified in Section 3.15.
- "Affiliate" means, with respect to any Person, any other Person that directly or indirectly (through one or more intermediaries or otherwise) Controls, is Controlled by, or is under common Control with such Person; provided, that any (a) representations or warranties herein of the Buyer that refer to any of the Buyer's "Affiliates" shall not be deemed or otherwise considered to include Canadian Pension Plan Investment Board or its other portfolio companies, or its or their subsidiaries, sponsors, or partners, and (b) covenants or agreements herein obligating Buyer to perform any action will not be deemed or otherwise considered to extend to any such Person (except in the case of Sections 2.11, 5.11, and 13.4, which shall apply to Canadian Pension Plan Investment Board).
- "**Agreement**" has the meaning specified in the introductory paragraph and includes the Exhibits and Schedules attached hereto.
- "Allocated Value" means, with respect to each Well or Real Property Interest for which value is allocated on Exhibit A-1, the portion of the Purchase Price attributable to such Well or Real Property Interest as set forth on Exhibit A-1.
- "Assessment" has the meaning specified in <u>Section 5.1.1</u>.
- "Asset Taxes" means production, severance, sales, use, occupation, ad valorem property, personal property or similar Taxes, and shall include any interest, penalties or additions to tax attributable to any such Tax, including penalties for the failure to file any Tax Return or report, but not income, franchise or commercial activity Taxes or Transfer Taxes.
- "Assignment" means the Assignment, Bill of Sale and Conveyance in substantially the same form attached hereto as <u>Exhibit B-1</u> with respect to the Properties other than the Surface Assets and the Mineral Interests.
- "Assumed Obligations" means, other than the Retained Liabilities, all Liabilities of every kind and character of the Sellers with respect to the Properties or to the ownership, use, operation or other disposition thereof, whether or not attributable to periods before or after the Effective Time, including those arising out of or relating to (i) Liabilities and duties to properly plug and abandon or re-plug or re-abandon or remove wells, flowlines, gathering lines or other facilities, equipment or other personal property or fixtures located on or within the geographical boundaries of the Lands or otherwise comprising part of the Properties, and (ii) except as specifically provided in Section 2.2.13, all Liabilities to restore the

surface of the Real Property Interests or the Lands and any other obligations relating to the failure of the Real Property Interests, Lands or Wells to comply with Environmental Laws, including any and all obligations to bring the Properties into compliance with applicable Environmental Laws (including conducting any remediation activities that may be required on or otherwise in connection with activities on the Properties), regardless of whether such obligations or conditions or events giving rise to such obligations arose, occurred or accrued before or after the Effective Time; provided, however, that to the extent a Retained Liability ceases to be a Retained Liability it shall thereafter be an Assumed Obligation. Notwithstanding anything to the contrary, the definition of "Assumed Obligations" shall be without prejudice to the express rights and remedies of the Buyer herein, including those with respect to Title Defects, Environmental Defects and the Sellers' indemnification obligations hereunder.

"Average Henry Hub Determination Price" has the meaning specified in Section 5.20.3.

"Basket" has the meaning specified in Section 10.6.1A.

"Benefit Notice" has the meaning specified in Section 2.2.11.

"Business Day" means any day other than Saturday or Sunday or a day on which banking institutions in Oklahoma City, Oklahoma are authorized by Law to close.

"Buyer" has the meaning specified in the introductory paragraph.

"Buyer Hedging Indemnities" has the meaning specified in Section 5.14.4.

"Buyer Indemnified Parties" has the meaning specified in Section 10.2.

"Buyer Pref Termination Right" has the meaning specified in Schedule 2.4.4.

"Buyer True-Up Payments" has the meaning specified in Section 5.20.1A.

"Casualty" means volcanic eruptions, acts of God, terrorist acts, fire, explosion, earthquake, wind storm, flood, drought, condemnation, the exercise of eminent domain, confiscation or seizure or other casualty, but excluding any loss, damage or reduction in value as a result of depreciation, ordinary wear and tear and any change in condition of the Properties for production of Hydrocarbons through normal depletion (including the watering out of any well, collapsed casing or sand infiltration of any well).

"Casualty Loss" has the meaning specified in Section 2.12.

"Casualty Loss Amount" has the meaning specified in Section 2.12.

"CFIUS" has the meaning specified in Section 5.12.1.

"CFIUS Approval" shall mean Buyer and Sellers shall have received written notice from CFIUS stating that: (a) CFIUS has concluded that the transaction is not a "covered transaction" and not subject to review under applicable Law; or (b) the review of the transaction contemplated by this Agreement under Section 721 of the U.S. Defense Production Act of 1950 has been concluded, and there are no unresolved national security concerns with respect to the transaction contemplated by this Agreement; or (c) CFIUS has sent a report to the President of the United States requesting the President's decision on the CFIUS notice submitted by Buyer and Sellers and either (i) the period under the Defense Production Act of 1950 during which the President may announce his decision to take action to suspend, prohibit or place any limitations on the transactions contemplated hereby has expired without any such action being threatened, announced or taken or (ii) the President has announced a decision not to take any action to suspend, prohibit or place any limitations on the transactions contemplated hereby.

**"CHK Parent Guaranty"** means a parent guaranty from Chesapeake Energy Corporation in substantially the form of Exhibit B-7.

"Closing" means the closing and consummation of the transactions contemplated by this Agreement.

"Closing Date" has the meaning specified in Section 8.

"Closing Statement" has the meaning specified in Section 2.8.

"Code" means the Internal Revenue Code of 1986, as amended, and any successor statute thereof.

"Completion Contract Counterparty" means Halliburton Energy Services Inc. or BJ Services, as applicable.

"Compressor Assets" has the meaning specified in the definition of Properties.

"Confidentiality Agreement" means that certain confidentiality agreement dated May 15, 2017 executed by Chesapeake Energy Corporation and Encino Energy, LLC, as amended.

"Consultant" has the meaning specified in Section 2.2.12.

"Contracts" has the meaning specified in the definition of Properties.

"Control" means the possession, directly or indirectly, of the power, directly or indirectly, to direct or cause the direction of the management or policies of the controlled Person, whether through the ownership of equity interests in or voting rights attributable to the equity interests in such Person, by contract or agency, by the general partner of a Person that is a partnership, or otherwise; and "Controls" and "Controlled" have meanings correlative thereto.

"Core Access Agreement" means a core access agreement in substantially the same form attached hereto as <u>Exhibit</u> <u>B-6</u>.

"Curable Defect" has the meaning specified in Section 2.1.3.

"Cure Period" has the meaning specified in Section 2.1.2.

"Currently Producing Zone" means for a given Well, the zone or formation containing the perforated interval(s) from which such Well is currently (or, in the case of shut-in Wells, is capable of) producing oil or gas, as applicable, as reported for such Well to the applicable Governmental Authority governing such Well on the Effective Time and the Closing Date (and any dates between), or such other zone or formation from which the Well is being (or, in the case of shut-in Wells, is capable of being) produced as can otherwise be reasonably determined within the Records of the Sellers on the Effective Time and the Closing Date (and any dates between).

"Customary Post-Closing Consents" has the meaning specified in the definition of Permitted Encumbrances.

"Debt Commitment Letter" shall mean the executed commitment letters (together with all of its respective exhibits, schedules, annexes, supplements and amendments thereto) from the applicable Debt Financing Sources, as amended, supplemented or replaced in compliance with this Agreement, pursuant to which such Debt Financing Sources have committed, subject only to the financing conditions set forth therein, to provide the Buyer the portion of the Debt Financing set forth therein.

**"Debt Financing"** shall mean the debt financing set forth in the Debt Commitment Letter, pursuant to the terms of the Debt Commitment Letter, including the offering or private placement of debt securities contemplated by the Debt Commitment Letter and any related engagement letter.

"Debt Financing Sources" shall mean the lending parties to the Debt Commitment Letter.

"**Defect Notice**" has the meaning specified in <u>Section 2.2</u>.

"**Defect Notice Date**" has the meaning specified in <u>Section 2.2</u>.

"**Defensible Title**" means that title of Sellers (and, from and after Closing, the Buyer) to the Wells or Real Property Interests, as applicable, that, subject to Permitted Encumbrances, as of Effective Time and the Closing Date (and all dates between):

(a) entitles the Sellers (and, from and after Closing, the Buyer) to receive, throughout the productive life of a Well or Real Property Interest, as applicable, not less than the Net Revenue Interest set forth in <a href="Exhibit A-1">Exhibit A-1</a>

in and to all Hydrocarbons produced and saved or sold from or allocated to the Target Formation of such Well or Real Property Interest, as applicable, except for (i) decreases in connection with any operation performed after the Execution Date in which the owner of such Well or Real Property Interest, as applicable, is a non-consenting owner or are taken on or after the Closing Date, including decreases resulting from any reversion of interest to co-owners with respect to which such co-owners, after the Execution Date, may elect to be a non-consenting co-owner, in each case, to the extent not in violation of Section 5.2, (ii) decreases resulting from the establishment of pools or units after the Execution Date to the extent not in violation of Section 5.2, (iii) decreases required to allow other working interest owners, pipelines or plants to make up past underproduction, and (iv) decreases resulting from reversionary interests, carried interests, horizontal or vertical severances, the effluxion of time or other matters or changes in interest, in each case, to the extent specifically shown in Exhibit A-1;

- (b) obligates the Sellers (and, from and after Closing, the Buyer) to bear, throughout the productive life of a Well (and the plugging, abandonment and salvage thereof) not greater than the Working Interest in such Well set forth in <a href="Exhibit A-1">Exhibit A-1</a> with respect to the Target Formation, except (i) increases in such Working Interest resulting from matters specifically shown in <a href="Exhibit A-1">Exhibit A-1</a>, (ii) increases resulting from contribution requirements with respect to defaulting parties under applicable operating, unit, pooling, pre-pooling or similar agreements, in each case, to the extent occurring after the Execution Date not in violation of <a href="Section 5.2">Section 5.2</a>, and (iii) increases that are accompanied by at least a proportionate increase in the Net Revenue Interest for such Well;
- (c) with respect to any Real Property Interest, entitles the Sellers (and, from and after Closing, Buyer), throughout the productive life of a Real Property Interest, to the Net Acres set forth on <u>Exhibit A-1</u> for such Real Property Interest with respect to the Target Formation; and
- (d) is free and clear of all Liens.

"Deposit" has the meaning specified in Section 2.

"Determination Date" has the meaning specified in Section 5.20.1.

"Disputed Title/Enviro Matter" has the meaning specified in Section 2.1.3.

"Dollar" means the United States of America dollar.

**"Drilling Rig Contract 1"** means that certain International Association of Drilling Contractors Drilling Bid Proposal and Daywork Drilling Contract – U.S. between Chesapeake Operating, L.L.C. and Drilling Rig Contract Counterparty, dated May 10, 2018 and covering Rig 571.

"**Drilling Rig Contract 2**" means that certain Daywork Drilling Contract by and between Chesapeake Operating, Inc. and the Drilling Rig Contract Counterparty, for Contractor's Rig 580, dated June 25, 2014, as amended.

"Drilling Rig Contracts" means, collectively, Drilling Rig Contract 1 and Drilling Rig Contract 2.

"Drilling Rig Contract Counterparty" means Patterson-UTI Drilling Company, LLC.

"Easements" has the meaning specified in the definition of Properties.

"Effective Time" means 7:00 a.m. Central Time, January 1, 2018.

"Environmental Defect" means, as to a Real Property Interest or Well, the violation of or liability under applicable Environmental Laws in effect as of the Execution Date; provided that the term "Environmental Defect" does not include the failure to meet good or desirable operating practices or standards that may be employed or adopted by other oil or gas well operators or recommended by a Governmental Authority but which are not subject to legal requirements under Environmental Laws, and <u>provided further</u> that the term "Environmental Defect" excludes any of the foregoing liabilities to the extent caused by or relating to NORM or otherwise disclosed in any Schedule. For the avoidance of doubt, (a) the fact that a pipe is temporarily not in use shall not form the basis of an Environmental Defect, (b) the fact that a Well should be temporarily plugged or abandoned or permanently plugged and abandoned, or facts or conditions based on the quantity of Hydrocarbon production from a Real Property Interest or Well, shall, in each case, not form the basis of an Environmental Defect, (c) except with respect to equipment (i) that causes or has caused contamination of soil, surface water or groundwater or (ii) the use or condition of which is a violation of Environmental Law, the physical condition of any surface or subsurface production equipment, including water or oil tanks, separators or other ancillary equipment, shall not form the basis of an Environmental Defect (so long as, in the case of clauses (a), (b) and (c), the existence of the matter(s) mentioned therein in and of themselves do not constitute a violation of or liability under applicable Environmental Laws in effect as of the Execution Date) and (d) no Retained Liability shall be the basis for an Environmental Defect.

"Environmental Defect Amount" has the meaning specified in Section 2.2.10.

"Environmental Defect Deductible" means an amount equal to two percent (2.0%) of the unadjusted Purchase Price.

"Environmental Law" means any Law relating to the environment, natural resources, health and safety, Hazardous Substances, industrial hygiene, or environmental conditions on, under, or about any of the Properties, including soil, groundwater, surface water, soil gas, and indoor and ambient air conditions, or the reporting, investigation or remediation of environmental contamination and

includes the Comprehensive Environmental Response, Compensation and Liability Act, including the Superfund Amendments and Reauthorization Act of 1986, 42 U.S.C. § 9601 et seq.; the Resource Conservation and Recovery Act, including the Hazardous and Solid Waste Amendments Act of 1984, 42 U.S.C. § 6901 et seq.; the Federal Water Pollution Control Act, 33 U.S.C. § 1251 et seq.; the Clean Air Act, 42 U.S.C. § 7401 et seq.; the Hazardous Materials Transportation Act, 49 U.S.C. § 1471 et seq.; the Toxic Substances Control Act, 15 U.S.C. §§ 2601 through 2629; the Oil Pollution Act, 33 U.S.C. § 2701 et seq.; the Emergency Planning and Community Right to Know Act, 42 U.S.C. § 11001 et seq.; the Safe Drinking Water Act, 42 U.S.C. §§ 300f through 300j; the Rivers and Harbors Act of 1899, 33 U.S.C. § 401 et seq.; and the Occupational Safety and Health Act, 29 U.S.C. § 651 et seq.; as any of the foregoing may be amended as of the Execution Date, as well as common law.

**"Environmental Records"** means all notices, demands, requests for information, and other correspondence transmitted between U.S. EPA and Sellers related to the Properties, and all records and documents created, compiled, or produced by or at the request of Sellers, in respect of the December 6, 2017, Request for Information from U.S. EPA to Chesapeake Energy Corporation under Section 114(a) of the Clean Air Act.

"Escrow Account" has the meaning specified in Section 2.

"Escrow Agent" means CitiBank, N.A., or its successor under the Escrow Agreement.

**"Escrow Agreement"** means the escrow agreement entered into as of the Execution Date by the Sellers, the Buyer, and the Escrow Agent.

"Exchange Act" has the meaning specified in <u>Section 11.2.1</u>.

#### "Excluded Assets" means:

- (a) the Excluded Information;
- (b) the items expressly identified on Exhibit A-5;
- (c) any existing or future refund of costs, Taxes or expenses borne by the Sellers, their Affiliates or each of their respective predecessors in title attributable to the period prior to the Effective Time;
- (d) any and all proceeds from production and from the settlements of contract disputes with purchasers of Hydrocarbons or byproducts from the Real Property Interests, including settlement of take-or-pay disputes, insofar as said proceeds are attributable to periods of time prior to the Effective Time;
- (e) all rights of the Sellers under Contracts attributable to periods before the Effective Time;

- (f) all rights and interests of the Sellers (i) under any bond, letter of credit or guarantee or (ii) relating to existing claims and causes of action that may be asserted against a Third Party;
- (g) all Hydrocarbons produced from the Properties with respect to all periods prior to the Effective Time and all proceeds from the disposition thereof other than Stored CEM NGLs, Stored CEM Oil, and inventory for which an adjustment is made to the Purchase Price under <u>Section 2.6.1</u>;
- (h) all audit rights (including rights to receive costs and revenues in connection therewith, in each case to the extent the Sellers are responsible for such costs under this Agreement) arising under any of the Contracts or otherwise with respect to the Properties, in each case, for any period prior to the Effective Time or to any of the Excluded Assets, except for any Gas Imbalances;
- (i) all trade credits, account receivables, note receivables, take-or-pay amounts receivable and other receivables attributable to the Properties (other than Stored CEM NGLs, Stored CEM Oil, and inventory for which an adjustment is made under <u>Section 2.6.1</u>), in each case, with respect to any period of time prior to the Effective Time, as determined in accordance with GAAP;
- (j) the Excluded IT Equipment;
- (k) all office supplies, furniture and related personal effects located off the Properties or only temporarily located on the Properties;
- (l) all artwork and personal effects whether located on or among the Properties;
- (m) Properties retained by the Sellers hereunder or excluded from the Properties at Closing pursuant to <u>Section 2.2</u>, <u>Section 2.4</u>, <u>Section 2.12</u>, <u>Section 5.1.1</u>, <u>Section 5.12</u>, or <u>Schedule 2.4.4</u>, subject to the terms of each such Section or Schedule;
- (n) all claims of the Sellers or any of their Affiliates for refunds of or loss carry forwards with respect to (i) any Asset Taxes allocated to a Seller pursuant to <u>Section 5.6</u>, (ii) income or franchise Taxes or (iii) any Taxes attributable to the Excluded Assets;
- (o) all rights and interests of the Sellers under any policy or agreement of (i) insurance or (ii) indemnity (including any rights, claims or causes of action of a Seller against Third Parties under any influence or hold harmless agreements and any indemnities received in connection with a Seller's prior acquisition of any of the Properties);

- (p) all drilling rigs and related equipment, work over rigs and related equipment, tools and other equipment, in each case, brought onto a well site temporarily for purposes of drilling, reworking or maintaining a well;
- (q) the Total Development Agreement (not including, for the avoidance of doubt, any agreements entered into pursuant thereto, including any joint operating agreements); and
- (r) except as set forth on <u>Exhibit A-6</u>, all assets of MidCon;

<u>provided</u> that, in the case of <u>clauses (e)</u>, (<u>f</u>), (<u>h</u>), and (<u>o</u>)(<u>ii</u>), the items described therein shall only constitute "Excluded Assets" to the extent that such items do not constitute Assumed Obligations for which Buyer has an indemnification obligation or otherwise constitute an indemnification obligation of the Buyer hereunder.

"Excluded G&A Expenses" means any general and administrative expenses (including overhead, salaries, benefits, legal and professional costs) incurred by Sellers or their Affiliates.

"Excluded Information" means (a) any data or licenses whose change in ownership or transfer is subject to payment of a fee or other consideration by an agreement with a Third Party or by applicable Law (which Buyer has not agreed in writing to pay or exchange to or with such Third Party); (b) any contracts, books, records, files, maps, information, data, software, licenses (including radio licenses), or similar information (or copies thereof) to the extent not relating to the Properties; (c) any records or information relating to the offer, negotiation or sale of the Properties, including bids received from and records of negotiations with Buyer or Third Parties; (d) any original or copied records, information, data, software and licenses permitted to be retained by Sellers pursuant to the terms hereof; (e) any of Sellers' (or their Affiliates') reserve, engineering, economic or marketing evaluations or interpretations, interpretive maps, or correspondence related thereto; (f) master service contracts, software or licenses used for both the Properties and other assets of the Sellers (or their Affiliates); (g) all corporate, limited liability company, partnership or financial, tax and legal records of Sellers relating to each Seller's business generally (other than title records, records related to Asset Taxes or except for copies of records subject to this clause (g) to the extent substantially related to any of the Assumed Obligations and not otherwise excluded by clauses (a), (e), (i), (j), (k) or (l) hereof); (h) the Sellers' area-wide bonds, permits and licenses used in the conduct of Sellers' business generally; (i) any environmental studies, evaluations, reports or interpretations thereof that were generated by or for the Sellers' (or their Affiliates') solely for internal use; (j) all "virtual courthouses" or physical or virtual "data rooms" of the Sellers; (k) each Seller's exclusive use arrangements with title abstract facilities; (l) all legal records and files constituting work product of, and attorney-client communications with, the Sellers' legal counsel that are protected by an attorney-client privilege (excluding the title documents); (m) all of the Sellers' (and their

Affiliates') proprietary computer software, patents, trade secrets, copyrights, names, trademarks, logos and other similar intellectual property; (n) any contracts, books, records, files, maps, information, data, software, permits, or licenses (including radio licenses), or similar information (or copies thereof), to the extent relating to any of the other Excluded Assets; (o) all G&G Data to the extent and only to the extent that it may not be assigned to the Buyer without payment of a fee or other penalty (unless the Buyer has agreed in writing to pay such fee or accept such penalty); and (p) any interpretations or analyses of G&G Data.

**"Excluded IT Equipment"** means all computers, mobile devices, servers, network equipment, computer peripherals, and external hard drives as well as email and software licenses.

**Excluded Property Costs**" means all (i) costs and expenses of Sellers related to: (a) claims, investigations, administrative proceedings, arbitration or litigation directly or indirectly arising out of or resulting from the Retained Liabilities (or any matters that would have constituted Retained Liabilities but were settled, compromised, or otherwise resolved prior to the Execution Date); (b) the remediation or cure (or attempted remediation or cure) of any Title Defect or Environmental Defect; (c) any costs associated with Sellers' obligations arising under Section 2.4 or any type of costs described in Section 13.22 of the Sellers; (d) any costs incurred with respect to an indemnification obligation of Sellers hereunder; or (e) except to the extent permitted by Sections 5.2 or 5.3, lease bonuses, lease acquisition costs, and similar payments incurred after the Execution Date and prior to the Closing Date; and (ii) Excluded G&A Expenses.

"Execution Date" has the meaning specified in the introductory paragraph.

"Expiration Date" has the meaning specified in Section 13.3.

"FCC" shall mean the Federal Communications Commission.

"FCC Assignment Applications" has the meaning set forth in Section 5.19.1.

"FERC" means the Federal Energy Regulatory Commission.

"FERC Contracts" has the meaning specified in Section 5.13.1.

"Final Statement" has the meaning specified in Section 2.9.

"**Financial Statements**" has the meaning specified in Section 11.2.1.

"**Financing**" means, to the extent obtained by the Buyer, any debt or equity financing (or combination thereof) issued, incurred or intended to be issued or incurred to be used in whole or in part to fund all or any portion of the Purchase Price.

"FTP Site" means that certain File Transfer Protocol site located at https://myftp.chk.com with the Id: ftp\_Utica\_2018.

"Fundamental Representations" means Section 3.1, 3.2, 3.3, 3.10, 3.11.

"G&G Data" means geological or geophysical information including seismic data, studies, core and fluid samples and information and any other similar geological or geophysical information, in each case, related to the Properties, but excluding any interpretations and analyses of such information.

"GAAP" means generally accepted accounting principles, consistently applied, as recognized by the U.S. Financial Accounting Standards Board (or any generally recognized successor). The requisite that such principles be consistently applied means that the accounting principles in a current period are comparable in all material respects to those applied in preceding periods.

"Gas Imbalances" means any gas production, pipeline, storage, processing, transportation or other imbalance or unsatisfied through-put obligations attributable to Hydrocarbons produced from the respective Seller's interest in such Seller's Wells or any interests pooled or unitized therewith.

"Gas Purchase Agreement" means a gas purchase agreement in substantially the form of Exhibit A-9.

"Governmental Authority" means any national, state, county or municipal government or division thereof, domestic or foreign, any governmental, regulatory, or administrative agency, board, bureau, commission, court, department or other instrumentality of any such government, or other authority exercising or entitled to exercise any administrative, executive, judicial, legislative, police, regulatory or taxing authority or power, or any court or governmental tribunal or arbitrator in any case that has jurisdiction over any of the Parties or any of their respective properties or assets.

"Hazardous Substance" means any waste, chemical, material or other substance that is listed, defined, designated or classified as hazardous, radioactive or toxic or a pollutant or a contaminant under any Environmental Law, or otherwise regulated under any Environmental Law, including petroleum and all derivatives thereof, asbestos or asbestos-containing materials in any form or condition, and polychlorinated biphenyls, but excluding NORM.

"Hedges" means any future derivative, swap, collar, put, call, cap, option or other contract that is intended to benefit from, relate to, or reduce or eliminate the risk of fluctuations in interest rates, basis risk or the price of commodities, including Hydrocarbons.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

- "HSR Approval" means the expiration or termination of any applicable waiting periods under the HSR Act.
- "Hydrocarbons" means oil, condensate, gas, casinghead gas and other liquid or gaseous hydrocarbons produced in association therewith.
- **"Hydrocarbons Purchase Agreement"** means, as applicable, the Gas Purchase Agreement, Oil Purchase Agreement and NGLs Purchase Agreement.
- "Individual Claim" has the meaning specified in Section 10.6.1.
- "Individual Defect Threshold" has the meaning specified in Section 2.2.1.
- "**Knowledge**" of a fact or matter means, with respect to a Seller, the actual knowledge (after reasonable due inquiry of those employees reporting directly to such individuals) of the individuals of such Seller listed in Schedule 1.1(a).
- "Lands" means the lands covered by the Real Property Interests or Units.
- "Lease Expiration Defect" has the meaning set forth in the definition of Title Defect.
- **"Laws"** means any and all applicable laws, statutes, ordinances, permits, decrees, writs, injunctions, orders, codes, judgments, rules or regulations (including Environmental Laws) which are promulgated, issued or enacted by a Governmental Authority having jurisdiction.
- "Liabilities" means any and all claims, causes of action, payments, charges, judgments, assessments, liabilities, losses, damages, penalties, fines, fees, interest obligations, deficiencies, debts, obligations, costs and expenses, and other liabilities (whether absolute, accrued, contingent, fixed or otherwise, or whether known or unknown, or due or to become due or otherwise), including amounts paid in settlement, interest, court costs, costs of investigators, and, to the extent reasonable, any attorneys', accountants', financial advisors' or other experts' fees or other expenses incurred in connection therewith.
- **"Lien"** means any lien, mortgage, security interest, pledge, charge, encumbrance or other arrangements substantially equivalent thereto, but does not include any production payment, net profits interest, overriding royalty interest or similar interest to the extent any such interest does not reduce the Sellers' Net Revenue Interest below that shown on Exhibit A-1.
- "Material Adverse Effect" means any result, consequence, condition or matter which, individually or in the aggregate, (a) materially adversely affects, or would be reasonably likely to be material adverse to, the Properties or the operations, rights, results of operations or the value of the Properties taken as a whole, or (b) materially impairs, prevents or delays a Seller's ability to perform its obligations under this Agreement or to consummate the transactions contemplated by this

Agreement; provided, however, that, in any event, the following shall not be deemed to constitute, create or cause a Material Adverse Effect: any changes, circumstances or effects (i) that affect generally the oil and gas industry, such as fluctuations in the price of oil and gas, or that result from international, national, regional, state or local economic conditions, from general developments or from other general economic conditions, facts or circumstances, including changes in Tax policy or other fiscal conditions, that are not subject to the control of the relevant Party, (ii) that result from the public announcement of any of the transactions contemplated in this Agreement, (iii) that result from conditions or events resulting from an outbreak or escalation of hostilities (whether nationally or internationally), or the occurrence of any other calamity or crisis (whether nationally or internationally), including the occurrence of one or more terrorist attacks or acts of God or natural disasters, or (iv) that result from acts or failures to act of any Governmental Authorities and changes in Laws from and after the Execution Date unless caused by the actions or inactions of Seller; provided, further, any result, consequence, condition or matter referred to in the immediately preceding clauses (i), (iii) or (iv) will be taken into account for purposes of determining whether there has been a Material Adverse Effect to the extent such result, consequence, condition or matter adversely affects the Properties in a disproportionately adverse manner relative to other companies operating in the oil and gas industry in the geographic region in which the Properties are located. Notwithstanding anything to the contrary in this Agreement, in determining whether a Material Adverse Effect exists, any indemnification by the Sellers or Buyers under this Agreement or any insurance, claim right of contribution, other indemnity or other rights, shall not be taken into consideration or account.

"Material Contracts" has the meaning specified in Section 3.8.

"Midstream Agreements" has the meaning specified in Section 5.13.1.

"Midstream Assignments" has the meaning specified in Section 5.13.1.

"Midstream Pass-Through Arrangement" has the meaning specified in Section 5.13.2.

"Mineral Deed" means the Mineral Deed which shall contain a limited warranty of title under applicable Laws of the State of Ohio with respect to the Mineral Interests, in substantially the same form attached hereto as Exhibit B-2.

"Mineral Interests" has the meaning specified in the definition of Properties.

"Mitigation Standard" has the meaning specified in Section 5.12.1.

"Natural Gas Hedges" has the meaning specified in Section 5.14.1.

"**Net Acres**" means, as calculated separately with respect to each Real Property Interest, the product of (a) the number of gross acres of land covered by such Real

Property Interest multiplied by (b) the lessor's undivided interest in the Hydrocarbons in the lands covered by such Real Property Interest, multiplied by (c) Sellers' undivided interests in such Real Property Interest (<u>provided</u> that if items (b) and (c) vary as to different areas of lands covered by any such Real Property Interests, then a separate calculation shall be performed with respect to each such area). All references herein to the Net Acres set forth on <u>Exhibit A-1</u> refer to the column labeled "Seller Net Acres" on <u>Exhibit A-1</u>.

"Net Revenue Interest" (or "NRI") means, with respect to any Well or Real Property Interest, the decimal interest in and to all production of Hydrocarbons produced and saved or sold from or allocated to the relevant Well or Real Property Interest after giving effect to all valid lessors' royalties, overriding royalties, nonparticipating royalties, net profits interests, production payments, carried interests, reversionary interests, and other burdens upon, measurable or payable out of production therefrom.

"**NGLs Purchase Agreement**" means a NGLs purchase agreement in substantially the form attached hereto as Exhibit B-10.

"Non-Total Pref Properties" has the meaning specified in Schedule 2.4.4.

"Non-Total Pref Properties Election" has the meaning specified in Schedule 2.4.4.

"Nonconsented Interest" has the meaning specified in Section 2.4.2.

"**NORM**" has the meaning specified in <u>Section 5.10</u>.

"Novation Agreements" shall mean, collectively, each ISDA Novation Agreement by and among a Natural Gas Hedge counterparty, the Sellers and the Buyer pursuant to which the Sellers will novate the Natural Gas Hedges to the Buyer, in a form agreed to by each of the parties thereto.

"NYMEX Gas Contract Price" means the New York Mercantile Exchange natural gas strip price for a given calendar month as published in the Platts Natural Gas Alert by S&P Global Platts.

"Offering Document" has the meaning specified in Section 11.2.1.

"Oil Purchase Agreement" means an oil purchase agreement in substantially the form of Exhibit B-8.

"Ordinary Course of Business" means in the ordinary course of business consistent with past custom and practice (based on the twelve (12) month period prior to the Execution Date) of the applicable Sellers or their respective Affiliates.

"Outside Date" has the meaning specified in Section 12.1.5.

"Party" and "Parties" have the meanings specified in the introductory paragraph.

"Payout Balance" has the meaning specified in Section 3.21.

"Permits" has the meaning specified in Section 3.6.

### "Permitted Encumbrances" means any of the following:

- (a) royalties, nonparticipating royalty interests, net profits interests and any overriding royalties, reversionary interests and other similar burdens or encumbrances, in each instance, to the extent they do not, individually or in the aggregate, reduce the Sellers' Net Revenue Interest or Sellers' Net Acres or increase the Sellers' Working Interest (without at least a proportionate corresponding increase in the Sellers' Net Revenue Interest) in any Property from that described in Exhibit A-1;
- (b) Liens for Taxes for which payment is not due or which are being contested in good faith by appropriate proceedings; <u>provided</u>, <u>that</u>, to the extent such Tax contests exist as of the Execution Date, they are set forth on <u>Schedule 3.5</u>;
- (c) Liens of mechanics, materialmen, warehousemen, landlords, vendors and carriers and any similar Liens arising by operation of Law which, in each instance, arise in the Ordinary Course of Business and relate to sums not yet due;
- (d) operating agreements, unit agreements, unitization and pooling designations and declarations, gathering and transportation agreements, processing agreements, Hydrocarbon purchase contracts and all of the Contracts, easements, surface leases and other surface rights and the terms of all of the Real Property Interests, in each instance, to the extent they do not, individually or in the aggregate, materially interfere with the ownership and operation of the Properties subject thereto or affected thereby (as currently owned and operated), and do not reduce the Sellers' Net Revenue Interest or Sellers' Net Acres, or increase the Sellers' Working Interest (without at least a proportionate corresponding increase in the Sellers' Net Revenue Interest), in the Properties subject thereto;
- (e) all rights reserved to or vested in any Governmental Authority (i) to control or regulate any of the Properties in any manner and all obligations and duties under all applicable Laws, rules and orders of any such Governmental Authority or under any franchise, grant, license or permit issued by any such Governmental Authority, or (ii) to terminate any right, power, franchise, license or permit afforded by such Governmental Authority;
- (f) all rights to consent by, required notices to, or filings with Governmental Authorities in connection with the sale, disposition, transfer or conveyance

of federal, state, tribal or other governmental oil and gas leases or interests therein or related thereto, or the transfer of operations of any of the Wells, where the same are customarily obtained subsequent to the assignment, disposition or transfer of such oil and gas leases or interests therein, or such operations ("Customary Post-Closing Consents");

- (g) except to the extent (i) already triggered or (ii) triggered upon the consummation of the transactions contemplated herein, conventional rights of reassignment obligating the lessee to reassign or offer to reassign its interests in any lease prior to a release or abandonment of such lease;
- (h) non-governmental Third Party consents applying to the transactions contemplated hereby; <u>provided</u> that, with respect to Required Consents and PPRs, the Sellers shall have complied with the provisions of <u>Section 2.4</u>;
- (i) all Liens, defects or irregularities of title, if any, affecting the Properties (i) which would be accepted by a reasonably prudent operator engaged in the business of owning and operating oil and gas properties and (ii) to the extent they do not, individually or in the aggregate, materially detract from the value of or materially interfere with the ownership and operation of the Properties subject thereto or affected thereby (as currently owned and operated), and do not reduce the Sellers' Net Revenue Interest or Sellers' Net Acres, or increase the Sellers' Working Interest (without at least a proportionate corresponding increase in the Sellers' Net Revenue Interest), in the Properties subject thereto;
- (j) all defects or irregularities (i) resulting from the failure to record releases of Liens, production payments or mortgages that have expired by their own terms or (ii) to the extent affecting depths, intervals, formations, or strata outside of the Target Formation;
- (k) rights of tenants-in-common in and to the Properties, to the extent they do not, individually or in the aggregate, materially interfere with the ownership and operation of the Properties subject thereto or affected thereby (as currently owned and operated), and do not reduce the Sellers' Net Revenue Interest or Sellers' Net Acres, or increase the Sellers' Working Interest (without at least a proportionate corresponding increase in the Sellers' Net Revenue Interest), in the Properties subject thereto;
- (l) calls on Hydrocarbon production under existing Contracts;
- (m) any encumbrance on or affecting the Properties which is discharged by the Sellers at or prior to Closing at no expense to the Buyer;
- (n) any matters specifically shown on <u>Exhibit A-1</u>;
- (o) any matters shown on Schedule 3.4;

- (p) the Gas Imbalances; and
- (q) any encumbrance or loss of title resulting from a Seller's conduct of business in compliance with this Agreement.

**"Person"** means an individual, corporation, partnership, limited liability company, joint venture, association, joint stock company, trust, unincorporated organization or other entity or Governmental Authority.

**"Phase I Environmental Assessment"** means an environmental site assessment performed in accordance with the American Society for Testing and Materials ASTM E1527-13, or any similar environmental assessment.

**"Phase II Environmental Assessment**" means a further assessment regarding a recognized environmental condition identified in a Phase I Environmental Assessment.

"Phase II Request" has the meaning specified in Section 5.1.1.

"PPR" has the meaning specified in Section 2.4.

"**Pref Notice**" has the meaning specified in <u>Schedule 2.4.4</u>.

"**Properties**" means with respect to each Seller, all of such Seller's right, title and interest in and to the following, excluding the Excluded Assets:

- (a) all oil and gas leases, oil, gas and mineral leases, subleases, record title, operating rights, other leaseholds, working interests and net revenue interests covering any lands located in the Target Area, including those leases described in <u>Section I</u> of Parts 1 and 2 of <u>Exhibit A-1</u> attached hereto and made a part hereof, whether producing or non-producing, together with all overriding royalties, production payments, carried interests, net profits interests, reversionary interests, and other Hydrocarbon interests of any kind or character created thereby, derived therefrom or attributable thereto, as well as all amendments, renewals, extensions or ratifications thereof (collectively, the "**Real Property Interests**");
- (b) all oil and gas wells located on or within the geographical boundaries of the Lands whether producing, shutin, plugged or abandoned located in the Target Area, including those wells described in Section II of Parts 1 and 2 of Exhibit A-1, (the "Wells") and all tangible personal property, equipment, fixtures and improvements located in the Target Area or held for use primarily in connection with the operation of the Properties, including all injection wells, salt water disposal wells, well heads, casing, tubing, pumps, motors, gauges, valves, heaters, treaters, water lines, vessels, tanks, boilers, separators, treating equipment, compressors, other equipment, automation systems including meters and related telemetry on wells, power

lines, telephone and communication lines and towers (including communication tower leases), and other appurtenances owned in connection with the production, treating, storing, transportation or marketing of Hydrocarbons from the Wells (the "**Equipment**");

- (c) all presently existing unitization, pooling or communitization agreements, declarations or designations and statutorily, judicially or administratively created drilling, spacing or production units, whether recorded or unrecorded, insofar as the same are attributable or allocated to the Real Property Interests, Wells or Mineral Interests and all of such Seller's interest in and to the properties covered or units created thereby which are attributable to the Real Property Interests, Wells or Mineral Interests (collectively, the "Units");
- all contracts, agreements, or other legally binding arrangements presently existing to which the Sellers are a party or are bound to the extent covering, attributable to or relating to any of the Properties, including: operating agreements, crude oil, condensate, and natural gas purchase and sale agreements, gathering agreements, transportation agreements, marketing, disposal or injection agreements, farmout and farmin agreements, unitization, pooling and communitization agreements, exploration agreements, area of mutual interest agreements, exchange and processing contracts and agreements, partnership and joint venture agreements, confidentiality agreements, FCC licenses and any other similar contracts, agreements and instruments, including those contracts and agreements described on <a href="Exhibit C">Exhibit C</a> (collectively, the "Contracts"); provided that "Contracts" shall exclude (i) any master service agreements, (ii) the Drilling Rig Contracts and (iii) the instruments constituting the Real Property Interests, Mineral Interests, Easements and Surface Assets and the assignments or conveyances in the Sellers' chain of title to the Real Property Interests, Mineral Interests, Easements and Surface Assets;
- (e) all (i) Hydrocarbons in, on, under or produced from the Real Property Interests, Wells, Mineral Interests or any interests pooled or unitized therewith from and after the Effective Time and the proceeds therefrom and (ii) the Stored CEM NGLs and Stored CEM Oil;
- (f) all mineral interests and lessor royalties located in the Target Area, including those set forth on <u>Exhibit A-2</u> (the "**Mineral Interests**");
- (g) easements, surface leases, subsurface leases, permits, licenses, servitudes, rights-of-way and all other rights and appurtenances situated on or used in connection with the operation of the Properties ("Easements");
- (h) originals (or electronic copies where originals do not exist) of all files, records and data (including electronic data) including title-related orders, contracts, opinions and lease and land files, well files, abstracts of title,

leases, division of interest statements, maps, and similar title information, engineering or production files, regulatory filings, and environmental, legal and accounting records, in each case, only to the extent directly related to the ownership or operation of the Real Property Interests, Units, Wells, Mineral Interests, Contracts, Easements, or Assumed Obligations; <u>provided, however</u>, that the Sellers may retain the originals of such records as the Sellers have reasonably determined may be required for litigation, Tax, accounting and auditing purposes (in which case the Sellers shall provide the Buyer with copies of such records) ("**Records**");

- (i) all rights, benefits and obligations arising from or in connection with any Gas Imbalances as of the Effective Time;
- (j) G&G Data to the extent that it may be assigned to the Buyer without payment of a fee or other penalty (unless the Buyer has agreed in writing to pay such fee or accept such penalty);
- (k) all surface fee interests and other surface property interests located in the Target Area, including those described on <u>Exhibit A-3</u> and all buildings, offices, improvements, appurtenances, field offices and yards located thereon (the "**Surface Assets**");
- (l) all vehicles and ATVs set forth on Exhibit A-4;
- (m) all rights, claims and causes of action (including all rights of indemnity recovery, set-off or refunds) of the Sellers or their Affiliates to the extent that such rights, claims or causes of action relate to the Assumed Obligations for which Buyer has an indemnification obligation or other indemnification obligations hereunder of the Buyer;
- (n) the Permits;
- (o) the assets of MidCon set forth on Exhibit A-6 (the "Compressor Assets"); and
- (p) any real or personal property (including tools but excluding any vehicles or ATVs to the extent not set forth on Exhibit A-4), located in the Target Area, in each case, that is primarily related to the ownership, operation or use of the other Properties but not otherwise described above.

"**Property Costs**" has the meaning specified in <u>Section 2.6.2</u>.

"**Purchase Price**" has the meaning specified in Section 2.

"Real Property Interests" has the meaning set forth in the definition of Properties.

"Records" has the meaning set forth in the definition of Properties.

"Required Consent" means a consent to assignment by a Person (a) that expressly provides that such consent may be granted or withheld in the sole discretion of the Person holding such consent or (b) that if not obtained prior to the assignment of the affected Property, (i) makes the Assignment with respect to such Property void or voidable or be terminable, or (ii) results in the Person holding such consent having the right to terminate such Seller's interest in the Property (or the instrument creating Seller's rights in the affected Property) subject to such consent; provided, however, "Required Consent" does not include any consent which by its terms cannot be unreasonably withheld or any Customary Post-Closing Consent.

"Required Financial Information" means, on an accounting month basis, (a) audited statements of revenues and expenses with respect to the Properties for the fiscal years ended December 31, 2015, December 31, 2016 and December 31, 2017, (b) unaudited statements of revenues and expenses with respect to the Properties for the sixmonth period ended June 30<sup>th</sup>, 2018 and (c) monthly production statements for the fiscal years ended December 31, 2015, December 31, 2016 and December 31, 2017 and for the fiscal quarters ended March 31, 2018 and June 30, 2018.

"**Retained Liabilities**" means all Liabilities and obligations of a Seller (or otherwise applying to or burdening the Properties) arising out of, incident to or in connection with the following:

- (a) such Seller's failure to pay or incorrect payment of any Asset Taxes allocated to a Seller pursuant to <u>Section</u> 5.6;
- (b) Third Party claims, demands, damages, costs and expenses, obligations or other liabilities with respect to bodily injury or death arising from such Seller's operation of the Properties prior to the Closing Date, insofar as (i) such claims are attributable to such Seller's interest in the Properties and (ii) the Buyer provides notice of an indemnifiable claim related to this clause (b) on or before the first anniversary of the Closing Date;
- (c) except as pertaining to Buyer's Assumed Obligations relating to the Suspended Funds, (i) claims from lessors and other holders of burdens on production including overriding royalties, production payments, net profits interests and other similar burdens for the Sellers' failure to pay, underpayment or incorrect payment of royalties or other burdens on production with respect to proceeds from the sale of Hydrocarbons produced from any of the Properties and attributable to the Sellers' ownership and operation of the Properties during the period prior to the Closing Date and (ii) claims from working interest owners for the Sellers' failure to pay, underpayment or incorrect payment of amounts owed to such working interest owners with respect to proceeds from the sale of Hydrocarbons produced from any of the Properties during the period prior to the Closing Date, in each case, insofar as the Buyer provides notice of an

indemnifiable claim related to this clause (c) on or before the fourth anniversary of the Closing Date;

- (d) are caused by, arise out of, or result from off-site disposal of any substance defined or regulated as a "pollutant," "solid waste," "hazardous waste," or "hazardous substance" under any Environmental Law that was generated or used by the Sellers on the Properties, to the extent that such disposal occurred prior to the Closing Date, insofar as the Buyer provides notice of an indemnifiable claim related to this clause (d) on or before the second anniversary of the Closing Date;
- (e) those litigation matters set forth (or to the extent Sellers had Knowledge thereof as of the Execution Date and therefore should have been set forth) on <u>Schedule 3.4</u> attached hereto (collectively, the "**Retained Litigation Matters**");
- (f) the Excluded Assets;
- (g) any indebtedness obligations of Sellers or its Affiliates;
- (h) any criminal liability imposed by any Governmental Authorities resulting from or relating to the ownership, use or operation of the Properties prior to the Closing Date, insofar as the Buyer provides notice of an indemnifiable claim related to this clause (h) on or before the statute of limitations for any such matter; and
- (i) the December 21, 2016 Finding of Violation issued by the U.S. EPA to Chesapeake Energy Corporation (insofar as any such Liabilities relate to operations on the sites set forth on <u>Schedule 1.1(b)</u>) and only insofar as any such Liabilities relate to operations on such Properties prior to the Closing Date and the Buyer provides notice of an indemnifiable claim related to this clause (i) on or before the statute of limitations for any such matter.

"Retained Litigation Matters" has the meaning set forth in the definition of "Retained Liabilities."

"SEC Documents" has the meaning specified in Section 11.2.1.

"Securities Act" has the meaning specified in <u>Section 11.2.1</u>.

"Seller Auditor" has the meaning specified in Section 11.2.1.

"Seller Indemnified Parties" has the meaning specified in Section 10.3.

"Sellers Hedging Indemnities" has the meaning specified in Section 5.14.6.

"Sellers' Representative" has the meaning specified in Section 13.27.

- "Sellers True-Up Payments" has the meaning specified in Section 5.20.1B.
- "Special Warranty" has the meaning specified in Section 2.3.
- "Stored CEM NGLs" has the meaning specified in Section 2.6.5.
- "Stored CEM Oil" has the meaning specified in Section 2.6.6.
- "**Straddle Period**" means any Tax period beginning before and ending on or after the date on which the Effective Time occurs.
- "Surface Assets" has the meaning specified in the definition of Properties.
- "**Surface Deed**" means the Surface Deed which shall contain a limited warranty of title under applicable Laws of the State of Ohio with respect to the Surface Assets, in substantially the same form attached hereto as <u>Exhibit B-3</u>.
- "Suspended Funds" means proceeds of production which a Seller is holding which are owing to Third Party owners of royalty, overriding royalty, working or other interests in respect of past production of Hydrocarbons attributable to the Properties; provided, that "Suspended Funds" shall not include any such proceeds being held by a Seller at any time prior to Closing as a result of any type of claim made prior to Closing and related to underpayment, incorrect payment or other mispayment described in clause (c)(i) or (c)(i) in the definition of "Retained Liabilities."
- "Target Area" means the State of Ohio.
- "Target Closing Date" has the meaning specified in Section 8.
- "**Target Formation**" means the Utica Formation, with respect to the Real Property Interests, and the Currently Producing Zone, with respect to the Wells.
- "Tax Partnership" has the meaning specified in Section 3.5.1.
- "Tax Return" means any return, report or statement required to be filed with respect to any Tax (including any attachments thereto, and any amendment thereof), including any information return, claim for refund, amended return or declaration of estimated Tax.
- "Taxes" means taxes of any kind, levies or other like assessments, customs, duties, imposts, charges or fees, including income, gross receipts, commercial activity, ad valorem, value added, excise, real or personal property, asset, sales, use, royalty, license, payroll, transaction, capital, net worth and franchise taxes, escheat and unclaimed property obligations, withholding, employment, social security, workers compensation, utility, severance, production, unemployment compensation, occupation, premium, windfall profits, transfer and gains taxes or other governmental taxes imposed or payable to the United States or any other

Governmental Authority, and in each instance such term shall include any interest, penalties or additions to tax attributable to any such Tax, including penalties for the failure to file any Tax Return or report or any liability as transferee or successor by contract or otherwise with respect to any of the foregoing.

"TD Excess Amount" has the meaning specified in Section 2.2.2.

"Third Party" means any Person other than a Party or an Affiliate of a Party.

"Timely Election" has the meaning specified in Schedule 2.4.4.

"Title Benefit" has the meaning specified in Section 2.2.11.

"Title Defect" means, as to the Sellers' interest in (i) a Well or Real Property Interest, any Lien, encumbrance or defect that causes the Sellers' title to the Target Formation of such Real Property Interest or Well to be less than Defensible Title as of the Effective Time and the Closing Date (and all dates between) and (ii) any Real Property Interest, any such Real Property Interest that (a) as of the Execution Date is neither being held by production nor operations nor is capable of being held beyond the primary term thereof by delay rentals or shut-in payments and (b) has a primary term that expires between the Execution Date and the day that is three hundred and sixty-five (365) days from the Execution Date (except to the extent identified as such on Exhibit A-1) (a "Lease Expiration Defect"). Notwithstanding the foregoing, the following shall not be considered Title Defects:

- (a) defects based solely on a lack of information in a Seller's files or references to a document if such document is not in a Seller's files, unless the Buyer provides affirmative evidence that such lack of information actually results in another Person's superior claim of title to the relevant Property;
- (b) defects arising out of a lack of corporate or other entity authorization unless the Buyer provides affirmative written evidence that the action was not authorized and results in a Third Party's superior claim of title;
- defects based on a failure to record leases issued by any state or federal governmental body, or any assignments of such leases, in the real property, conveyance or other records of the county in which the Property is located, <u>provided</u> that such Properties or assignments (i) were properly filed in the applicable Governmental Authority's office or are otherwise reflected in the records of the applicable Governmental Authority lessor to be recorded in the real property or other county records of the county in which the applicable Property is located;
- (d) defects based on a gap in a Seller's chain of title in the county records as to fee leases, unless such gap is affirmatively shown to exist in such records

by an abstract of title, title opinion or landman's title chain which documents shall be included in the Defect Notice;

- (e) defects arising out of a lack of a survey, unless a survey is expressly required by applicable Laws;
- (f) defects in the chain of title consisting of the failure to recite marital status in a document or omissions of successions of heirship or estate proceedings, unless the Buyer provides affirmative evidence that such failure or omission is expected to result in a Third Party's superior claim of title;
- (g) defects as a consequence of cessation of production, insufficient production, failure to report production or report production timely, or failure to conduct operations on any of the Real Property Interests held by production, or lands pooled, communitized or unitized therewith, except to the extent that the Buyer provides affirmative evidence that such matter occurred and that it has given a right of the lessor or other Third Party to terminate the underlying lease (which documentation shall be included in the Defect Notice);
- (h) defects that have been cured by the passage of time, the doctrine of adverse possession, applicable laws of limitation or prescription or such other matter that would render such defect invalid according to applicable Law:
- (i) expiration of, or other defects affecting, Real Property Interests that are described on <u>Exhibit A-1</u> as having an expiration date on or prior to the Target Closing Date;
- (j) any unsubordinated mortgage Lien on the fee estate or mineral fee estate from which title to the relevant Real Property Interest is derived which is not currently subject to foreclosure or other enforcement proceedings by the holder of the mortgage Lien;
- (k) any encumbrance or loss of title resulting from a Seller's conduct of business in compliance with this Agreement; and
- (l) defects specifically shown in Exhibit A-1.

"Title Defect Amount" means the amount by which the Allocated Value of a Well or Real Property Interest is reduced as a result of the existence of a Title Defect, as determined in accordance with the applicable principles set forth in Section 2.2.

"Title Defect Deductible" means an amount equal to one and a half percent (1.5%) of the unadjusted Purchase Price.

"Total" means Total E&P USA, Inc.

"Total AMI" has the meaning specified in Schedule 2.4.4.

"Total Development Agreement" means that certain Development Agreement dated December 30, 2011 by and among Total, CELLC, CHK Utica, EnerVest Energy Institutional Fund IX, L.P., EnerVest Energy Institutional Fund IX-WI, L.P., EnerVest Energy Institutional Fund XI-A, L.P., EnerVest Institutional Fund XI-WI, L.P., CGAS Properties, L.P. and Belden and Blake Corporation.

"Total Pref Properties" has the meaning specified in Schedule 2.4.4.

**"Total Pref Right"** means Total's preferential right to purchase certain of the Properties pursuant to the Total Development Agreement.

"Total PSA" has the meaning specified in Schedule 2.4.4.

"Total Pref Right Break-fee" has the meaning specified in Schedule 2.4.4.

"Transfer Taxes" has the meaning specified in Section 13.21.

"True-Up Quantity" has the meaning specified in Section 5.20.4.

"True-Up Payments" has the meaning specified in Section 5.20.1B.

"True-Up Period" has the meaning specified in Section 5.20.1.

"TSA" means a transition services agreement in substantially the form of Exhibit B-5.

**"Unleased AR"** means all accounts receivable in respect of amounts owed to Sellers by owners of Unleased Interests for amounts paid or incurred by a Seller, prior to the Effective Time, in the operation or maintenance of any such Unleased Interests (including, costs for drilling, testing, completing, equipping and operating the Properties and any supervision charges).

"Unleased Interests" means all unleased oil, gas or mineral interests owned by non-participating owners in a Well.

"Unwinding Scenario" has the meaning specified in Section 5.14.5.

"Utica Formation" means the geological formations known as the Utica Shale and the Point Pleasant Formation, which generally includes the stratigraphic equivalent of the formations found in the Calvin Mangun well (API # 34019220730000), located in Augusta Township, Carroll County, Ohio, with the top (being the top of the Utica Formation) at the measured depth of 7,325 feet (as seen on the Well Log) and the base (being the top of the Trenton Formation) at the measured depth of 7,622 feet (as seen on the Well Log), recognizing that the actual depth will vary across the Utica area.

"Well Log" means the well log set forth on Exhibit F.

"Wells" has the meaning set forth in the definition of Properties.

"Working Interest" (or "WI") means, with respect to a Well, the decimal interest in and to such Well that is burdened with the obligation to bear and pay costs and expenses of maintenance, development and operations on or in connection with such property, but without regard to the effect of any valid lessor royalties, overriding royalties, production payments, net profits interests or other similar burdens upon, measured by or payable out of production therefrom.

"Williams Dry GGA" means that certain Amended and Restated Gas Gathering Contract – Fixed Fee Utica by and among CEM, Chesapeake Operating, L.L.C., CELLC, CHK Utica and Williams Field Services Company, L.L.C., dated September 1, 2015, as amended.

1.2 References. All references in this Agreement to Articles, Exhibits, Schedules, Sections, paragraphs, subsections and other subdivisions refer to the corresponding Articles, Exhibits, Schedules, Sections, paragraphs, subsections and other subdivisions of or to this Agreement unless expressly provided otherwise. Titles appearing at the beginning of any Articles, Sections, subsections or other subdivisions of this Agreement are for convenience only, do not constitute any part of this Agreement, and shall be disregarded in construing the language hereof. A defined term has its defined meaning throughout this Agreement regardless of whether it appears before or after the place where it is defined, and its other grammatical forms have corresponding meanings. Examples are not to be construed to limit, expressly or by implication, the matter they illustrate. Each accounting term not defined herein, and each accounting term partly defined herein to the extent not defined, will have the meaning given to it under GAAP as interpreted as of the Execution Date. All references to prices, values or monetary amounts refer to Dollars. Pronouns in masculine, feminine or neuter genders shall be construed to state and include any other gender, and words, terms and titles (including terms defined herein) in the singular form shall be construed to include the plural and vice versa, unless the context otherwise requires. References to a specific time shall refer to prevailing Central Time, unless otherwise indicated. Except as otherwise specifically provided in this Agreement, any agreement, instrument, or writing defined or referred to herein means such agreement, instrument, or writing, as from time to time amended, supplemented, or modified prior to the Execution Date. Reference herein to any federal, state, local, or foreign Law shall be deemed to also refer to all rules and regulations promulgated thereunder, unless the context requires otherwise, and shall also be deemed to refer to such Laws as in effect as of the Execution Date. Exhibits and Schedules referred to herein are attached to and by this reference incorporated herein for all purposes. With respect to any of the Properties operated, owned, or otherwise used by an Affiliate of any Seller that is not a party hereto, representations, warranties, covenants, and obligations of such Seller herein will be deemed to also include, cover, and serve as references to

such Affiliates. As used herein, the words (a) "day" means calendar day, (b) "or" shall include both the conjunctive and disjunctive, unless the context requires otherwise, (c) "extent" in the phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if", (d) "shall" and "will" have equal force and effect, (e) "including" (in its various forms) means including without limitation, (f) "this Agreement," "herein," "hereby," "hereunder" and "hereof," and words of similar import, refer to this Agreement as a whole and not to any particular subdivision unless expressly so limited, and (g) "this Section" and "this subsection," and words of similar import, refer only to the Section or subsection hereof in which such words occur.

# ARTICLE 2 PURCHASE AND SALE PRICE

2. Purchase and Sale; Purchase Price. At the Closing, and upon the terms and subject to the conditions of this Agreement, the Sellers agree to sell and convey to the Buyer all of the Properties, and the Buyer agrees to purchase, accept and pay for the Properties and to assume the Assumed Obligations. In consideration for the sale of the Properties, and upon the terms and subject to the conditions of this Agreement, the Buyer will pay to the Sellers the purchase price of **Two Billion Fifteen Million and Four Hundred Sixty-Three Thousand Dollars** (\$2,015,463,000), adjusted as set forth herein (the "**Purchase Price**"). Within two (2) Business Days after the Execution Date, the Buyer will deposit or cause to be deposited with the Escrow Agent an earnest money deposit equal to **Fifty Million Dollars** (\$50,000,000) (including any interest earned thereon, the "**Deposit**") to be held in an escrow account (the "**Escrow Account**") pursuant to the Escrow Agreement. At Closing, the Deposit will be applied against the Purchase Price in accordance with the provisions of this Agreement and the Parties shall instruct the Escrow Agent to pay the Deposit to Seller at Closing. However, if the Closing does not occur, the Parties shall instruct the Escrow Agent to pay the Deposit in accordance with <u>Section 12.3</u>. The Purchase Price will be adjusted (without duplication) as follows:

### 2.1 <u>Title and Environmental Defects.</u>

- 2.1.1 Except as set forth in <u>Section 2.1.3</u>, in the case of a Title Defect or Environmental Defect, as applicable, on which the Parties agree and that the Parties agree has not been cured or remediated prior to Closing and as to which the Parties agree on the associated Title Defect Amount or Environmental Defect Amount, then, subject to the limitations set forth in <u>Section 2.2</u>, there shall be a downward adjustment to the unadjusted Purchase Price equal to the Title Defect Amount or Environmental Defect Amount, as applicable, for such Title Defect or Environmental Defect, as applicable, and, except as otherwise provided herein, the affected Well or Real Property Interest, as applicable, shall be conveyed to the Buyers at Closing.
- 2.1.2 If a Title Defect or an Environmental Defect is reasonably susceptible of being cured or remediated, as applicable, the Buyer and the Sellers agree

that the Sellers will have the right to cure such defect on or before ninety (90) days after the Closing Date (the "Cure Period"); provided that such election to cure such defect shall not waive or be deemed to waive Sellers' right to dispute the existence of such alleged defect or the Purchase Price adjustment asserted with respect thereto.

- 2.1.3 If (a) the Sellers provide notice at or prior to the delivery of the Closing Statement pursuant to Section 2.8 of the Sellers' intent to attempt to cure a Title Defect or remediate an Environmental Defect within the Cure Period ("Curable Defect") or (b) the Sellers and the Buyer are unable, prior to the delivery of the Closing Statement pursuant to Section 2.8, to agree on an alleged Title Defect or Environmental Defect (including the adequate cure or remediation therefor) or Title Defect Amount or Environmental Defect Amount (a "Disputed Title/Enviro Matter"), and Sellers do not exercise their right under Section 2.2 to permanently exclude the affected Well or Real Property Interest from the Properties sold hereunder, then, subject to Sections 2.1.4 and 2.1.5, the other limitations set forth in Section 2.2: (i) the Buyer shall deposit, at the Closing, into the Escrow Account in accordance with the terms of this Agreement and the Escrow Agreement an amount, after application of the Individual Defect Threshold, Environmental Defect Deductible and Title Defect Deductible, as applicable, equal to the Buyer's reasonable estimate of the Title Defect Amount or Environmental Defect Amount, as applicable, relating to Curable Defect or the Disputed Title/Enviro Matter, as the case may be, (ii) the affected Well or Real Property Interest, as applicable, shall be excluded from the Properties conveyed to Buyer at Closing, and (iii) the unadjusted Purchase Price shall be adjusted downward by such Title Defect Amount or Environmental Defect Amount estimated by Buyer pursuant to clause (i) above, as applicable, so deposited into the Escrow Account.
- 2.1.4 Following the Closing, the amounts deposited in the Escrow Account pursuant to Section 2.1.3 shall be distributed to the appropriate Parties, as follows: (a) with respect to Curable Defects (other than Curable Defects that constitute Disputed Title/Enviro Matters), (i) in the event that a Curable Defect is fully cured by the Sellers prior to the conclusion of the Cure Period, the Sellers shall be entitled to receive the amount deposited (together with any interest thereon) by the Buyer in the Escrow Account therefor, (ii) in the event that a Curable Defect is not partially or fully cured by the Sellers prior to the conclusion of the Cure Period, the Buyer shall be entitled to receive the amount deposited (together with any interest thereon) by the Buyer in the Escrow Account therefor or (iii) in the event that a Curable Defect is partially cured by the Sellers prior to the conclusion of the Cure Period, (A) the Sellers shall be entitled to receive an amount equal to the reduction in the Title Defect Amount or Environmental Defect Amount, as applicable, for such Curable Defect attributable to such partial cure and (B) the Buyer shall be entitled to

receive the remainder of the amount deposited (together with any interest thereon) by the Buyer in the Escrow Account therefor; and (b) with respect to any Disputed Title/Enviro Matter, pursuant to the Consultant's decision rendered pursuant to Section 2.2.12. In order to effect any such distribution, the Parties shall deliver, within one (1) Business Day following the date on which the applicable amount has become due to the relevant Party, a joint written instruction to the Escrow Agent requesting that the Escrow Agent release those portions of the Escrow Account that are determined to be owed to such Party pursuant to this Section 2.1. Upon final resolution of the cure of a Curable Defect or all disputes with respect to a Disputed Title/Enviro Matter, subject to Section 2.2.4, Sellers shall execute and deliver (x) an assignment (in substantially the same form as the Assignment) of (a) the Well or (b) the Real Property Interest and associated Properties associated with such Curable Defect or Disputed Title/Enviro Matter and (y) the items referenced in Sections 8.2.4 (dated as of the date of such assignment), 8.2.5, 8.2.12 and 8.2.13, as applicable, in each case only with respect to the Properties being assigned.

- 2.1.5 If the Parties dispute whether a Title Defect or Environmental Defect has been cured, then the matter shall be resolved in a manner described in Section 2.2.12.
- 2.2 <u>Defect Notice</u>. Subject to the limitations set forth herein, the Buyer may conduct, at its sole risk, cost and expense the Assessment, as defined below in Section 5.1.1, in order to determine whether any Title Defects or Environmental Defects exist. In that regard, the Buyer must deliver to the Sellers, on or before 5:00 p.m. Central Standard Time on the date that is sixty (60) days after the Execution Date (the "Defect Notice Date"), a written notice specifying each defect associated with a Real Property Interest or Well that the Buyer identified in its Assessment (or other due diligence) and that the Buyer asserts constitutes a Title Defect or an Environmental Defect, a specific description of each such Title Defect or Environmental Defect and the basis for such assertion under the terms of this Agreement, the amount of the adjustment to the Purchase Price that the Buyer asserts based on such Title Defect or Environmental Defect and its method of calculating such adjustment, together with data and information evidencing the Buyer's review of public record or other information (or, if in the Sellers' files, references to such information) of such Title Defect or Environmental Defect reasonably necessary for the Sellers to verify the existence of the alleged Title Defect or Environmental Defect (each, a "Defect Notice"). The Buyer shall bear the burden of proof to establish each alleged Title Defect or Environmental Defect. Any matters that constitute Title Defects or Environmental Defects but of which the Sellers have not been specifically notified by the Buyer in writing by the Defect Notice Date in accordance with the foregoing, shall be deemed to have been waived by the Buyer for all purposes, except with respect to the Special Warranty and the indemnity provided under Section 10.2 as it relates to (a) breaches of the representation in Section 3.9, (b) any Retained Liabilities, or (c) breaches of the covenants in Section 5.1 or Section 5.2. All adjustments to the

Purchase Price based on Title Defects or Environmental Defects will be based on the Allocated Values attributable to the affected Real Property Interest or Well (except in the case of Title Defects described in Section 2.2.7 and Environmental Defects (in each case, to which the limitations in this sentence shall not apply)); in no event shall any such adjustment exceed the Allocated Value attributable to such affected Real Property Interest or Well. Upon timely delivery of a Defect Notice of a Title Defect or Environmental Defect under this Section 2.2, the Buyer and the Sellers will in good faith negotiate the validity of the claim and the amount of any adjustment to the Purchase Price using the following criteria:

- 2.2.1 No single Title Defect or Environmental Defect shall be taken into account unless the value of such Title Defect or Environmental Defect is determined to be more than **One Hundred and Fifty Thousand Dollars** (\$150,000) (the "Individual Defect Threshold"), in which event the entire value of each such defect shall be taken into account for purposes hereof.
- 2.2.2 No adjustment will be made to the Purchase Price, and no amounts shall be deposited in the Escrow Account pursuant to Section 2.1.3, for either uncured Title Defects or unremediated or unindemnified Environmental Defects except to the extent that the total of all individual adjustments for uncured Title Defects and unremediated or unindemnified Environmental Defects that exceed the respective Individual Defect Thresholds exceed (a) in the case of Title Defects, the Title Defect Deductible, after offsetting any Title Benefits (as used in this Section 2.2.2, such excess amount, the "TD Excess Amount"), following which, the Purchase Price will be decreased by the TD Excess Amount or (b) in the case of Environmental Defects, the Environmental Defect Deductible (as used in this Section 2.2.2, such excess amount, the "ED Excess Amount"), following which, the Purchase Price will be decreased by the ED Excess Amount. For the avoidance of doubt, the Title Defect Deductible and the Environmental Defect Deductible (i) are separate, standalone deductibles, (ii) shall not be aggregated together for any purposes hereunder and (iii) shall apply only, subject to the other limitations herein, (A) in the case of the Title Defect Deductible, to Title Defects and (B) in the case of Environmental Defect Deductible, to Environmental Defects.
- 2.2.3 Notwithstanding anything to the contrary, neither the Individual Defect Threshold nor the Title Defect Deductible shall apply to (or otherwise limit) Title Defects to the extent such Title Defects would constitute a breach of the Special Warranty under an assignment by the Sellers if a Third Party were to make a claim with respect to the circumstances constituting such Title Defect, and the Purchase Price shall be adjusted by the full Title Defect Amount for any such Title Defect. Further, the Parties agree that if the same Title Defect or Environmental Defect affects more than one Property, then the Title Defect Amounts or Environmental Defect Amounts, as applicable, shall be aggregated for the purposes of

- determining whether any threshold, deductible or similar limitation herein has been met.
- 2.2.4 With respect to any alleged Title Defects or Environmental Defects for which an adjustment equal to or in excess of the entire Allocated Value of the affected Property, is or will be made to the Purchase Price hereunder, at Sellers' option, as a condition precedent to such adjustment, the applicable Seller shall retain (a) the Well or (b) the Real Property Interest and associated Properties (or segregated area thereof or interest therein) for which the Purchase Price is to be reduced and the same shall be deemed Excluded Assets for all purposes hereunder.
- 2.2.5 If the alleged Title Defect is based on the Sellers owning a Net Revenue Interest in a Well or Real Property Interest which is less than that shown on <a href="Exhibit A-1"><u>Exhibit A-1</u></a>, then the Title Defect Amount with respect to such Well or Real Property Interest shall be calculated by multiplying the Allocated Value set forth on <a href="Exhibit A-1"><u>Exhibit A-1</u></a> for such Well or Real Property Interest less the Net Revenue Interest to which a Seller is actually entitled taking such Title Defect into account, and the denominator of which is the Net Revenue Interest shown on <a href="Exhibit A-1"><u>Exhibit A-1</u></a> for such Well or Real Property Interest; <a href="provided"><u>provided</u></a>, <a href="however"><u>however</u></a>, that, notwithstanding the foregoing, if a Title Defect results in Sellers' Net Revenue Interest in any Well or Real Property Interest being equal to or less than <a href="fifty percent">fifty percent</a> (50%) (proportionately reduced based on Sellers' collective Working Interest therein), the Title Defect Amount with respect to such Well or Real Property Interest shall be determined in accordance with <a href="Section 2.2.9">Section 2.2.9</a> below rather than this <a href="Section 2.2.5">Section 2.2.5</a>.
- 2.2.6 If the alleged Title Defect is based on a discrepancy between (A) the Net Acres for the affected Real Property Interest, and (B) the Net Acres stated in <a href="Exhibit A-1">Exhibit A-1</a> for such Real Property Interest the Title Defect Amount shall be the product of the Allocated Value of such Real Property Interest multiplied by a fraction, the numerator of which is the Net Acres decrease for such Real Property Interest and the denominator of which is the Net Acres of such Real Property Interest set forth on <a href="Exhibit A-1">Exhibit A-1</a>.
- 2.2.7 If the alleged Title Defect is based on a Lien upon a Real Property Interest or Well that is liquidated in amount, then the Title Defect Amount is the amount necessary to remove such Lien from the affected Real Property Interest or Well.
- 2.2.8 If the alleged Title Defect is an alleged Lease Expiration Defect, the Title Defect Amount shall be the product of the Allocated Value of the affected Real Property Interest multiplied by a fraction, (i) the numerator of which is (a) **365** *minus* (b) the number of days within the period beginning on the

- Execution Date and ending on the actual expiration date of the Real Property Interest as set forth on the applicable lease, and (ii) the denominator of which is **365**.
- 2.2.9 If the alleged Title Defect represents an obligation, encumbrance, burden or charge upon or other defect in title to the Well or Real Property Interest of a type not described above, the Title Defect Amount is the lesser of (i) the amount, net to the Sellers' interest, necessary to compensate the Buyer at Closing for the adverse economic effect on the affected Well or Real Property Interest (taking into account the legal effect of the Title Defect, the potential economic effect of the Title Defect over the life of the affected Well or Real Property Interest, the values placed upon the Title Defect by the Buyer and the Sellers and such other factors as are necessary to make a proper evaluation) and (ii) the Allocated Value of the affected Property.
- 2.2.10 If the defect claim is based on an obligation or liability to investigate, remediate or otherwise cure an Environmental Defect related to a Well or Real Property Interest, then the Environmental Defect amount is the amount, net to Sellers' interest, necessary to investigate (not including, for the avoidance of doubt, the cost of the Assessment), remediate or otherwise cure such Environmental Defect in the most cost effective manner permissible under Environmental Law and consistent with existing uses of the Well or Real Property Interest for production, processing, transportation or storage of Hydrocarbons, but in all cases excluding any exacerbation of, or contribution to, the Environmental Defect caused by the Buyer or its Affiliates ("Environmental Defect Amount").
- 2.2.11 If, after the Execution Date, a Seller discovers that its ownership of any (i) Well or Real Property Interest entitles such Seller to a larger Net Revenue Interest, (ii) Well entitles such Seller to a smaller Working Interest (without a smaller Net Revenue Interest) or (iii) Real Property Interest entitles such Seller to a greater number of Net Acres, in each case than that set forth on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest not listed on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest not listed on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest not listed on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest not listed on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest not listed on <a href="Exhibit A-1">Exhibit A-1</a> (including by way of its ownership of a Real Property Interest entitles such Sellers Notice Date, describing in such notice with reasonable detail each Title Benefit (a "Benefit Notice"). The Buyer shall not be under any obligation to furnish the Sellers with notice of any Title Benefit that is discovered by any of the Buyer's representatives, title attorneys, landmen or other title examiners while conducting the Buyer's due diligence with respect to the Properties prior to the Defect Notice Date. The amount of each Title Benefit shall be determined in the same manner as provided in this <a href="Section 2.2">Section 2.2</a> with respect to Title Defects. The Sellers shall be deemed to have conclusively waived any Title Benefit of which the Sellers fail to notify the Buyer in

writing in the manner described in this Section 2.2.11. After taking into account all of the Title Defects that are cured, the value of the Sellers' Title Benefits shall be used to further reduce the Sellers' aggregate Title Defect Amount by subtracting such value of the Title Benefits from the remaining aggregate Title Defect value and then, if there is any Title Benefit value remaining, as an upward adjustment to the Purchase Price to the extent that Title Benefits are in excess of any Title Defects asserted hereunder. Any Title Benefits or Title Benefit amounts on which the Parties cannot agree at Closing shall be resolved pursuant to the provisions of Section 2.2.12; provided, that, for the avoidance of doubt, the Buyer shall not be required to deposit, at the Closing, into the Escrow Account any amount for any such disputed Title Benefit. In the event there are any such disputed Title Benefits or Title Benefit amounts, (i) the Well or Real Property Interest, as applicable, affected by such Title Benefit shall be conveyed to the Buyers at Closing, (ii) such disputed Title Benefit shall be used at Closing to offset any Title Defect for purposes of determining any Purchase Price adjustment, (iii) such disputed Title Benefits shall not be used at Closing to increase the Purchase Price (other than to offset Title Defects) and (iv) once finally determined pursuant to Section 2.2.12, such disputed Title Benefits shall be treated in accordance with the terms of this Section 2.2.11 (without duplication of any offset of Title Defects at the Closing).

2.2.12 If, after the end of the Cure Period, there is a dispute between the Sellers and the Buyer involving (a) any Curable Defects or Disputed Title/Enviro Matters and (b) Title Benefits, the Sellers or the Buyer shall have until the date thirty (30) days after the end of the Cure Period to submit written notice to the other Party that such Party is initiating dispute resolution in accordance with this Section 2.2.12, such notice to describe in reasonable detail the nature and specifics of the dispute. The Buyer, with respect to Title Defects and Environmental Defects, and the Sellers, with respect to Title Benefits, shall be deemed to have conclusively waived any such disputed Title Defect or such disputed Environmental Defect for which the Sellers have not elected to provide an indemnity or cure thereof or any unresolved Title Benefit which such Party fails to submit for resolution as provided in this <u>Section 2.2.12</u> by the date thirty (30) days after the end of the Cure Period, except with respect to the Special Warranty and the indemnity provided under Section 10.2 as it relates to (i) breaches of the representation in Section 3.9, (ii) any Retained Liabilities, or (iii) breaches of the covenants in <u>Section 5.1</u> or <u>Section 5.2</u>. This <u>Section 2.2.12</u> shall be the exclusive method for the resolution of the disputes described in subparts (a) and (b) of this Section 2.2.12. The matter to be resolved shall be submitted to a title attorney practicing in the state where the applicable Property is located selected by the Sellers and the Buyer, in the case of a Title Defect or Title Benefit, or to an environmental expert selected by the Sellers and the Buyer, in the case of an Environmental Defect (each such title attorney or environmental expert hereinafter, who shall not have

represented either Party during the prior twelve (12) month period, a "Consultant"). In the event the Sellers and the Buyer are unable to agree on any Consultant within thirty (30) days after receipt of the initiating notice, the Sellers on the one hand and the Buyer on the other hand will each appoint one Consultant within twenty (20) days thereafter and the two Consultants so appointed will appoint a third Consultant within twenty (20) days after the second Consultant is appointed, and the three Consultants so appointed will resolve such matter. The cost of each Consultant shall be paid fifty percent (50%) by the Sellers and fifty **percent** (50%) by the Buyer. The Sellers and the Buyer shall each present to the Consultant(s), with a simultaneous copy to the other Party, a single written statement of its position on the defect, benefit or dispute in question, together with a copy of this Agreement and any supporting material that such Party desires to furnish, not later than ten (10) Business Days after appointment of the Consultant(s). The Buyer may only assert violations or defects set forth in the applicable Defect Notice and may not assert any new or additional defect, violation of law, benefit or dispute in such written statement. Similarly, the Sellers may only assert Title Benefits set forth in the applicable Benefit Notice and may not assert any new or additional Title Benefits in such written statement. In making their determination, the Consultant(s) shall be bound by the terms of this Agreement and, without any additional or supplemental submittals by either Party, may consider available legal and industry matters as in their opinion are necessary or appropriate to make a proper determination. Within sixty (60) days following the submission of such written statements to the Consultant(s), applying the principles set forth in this Section 2.2, the Consultant(s) shall consider only those items submitted by each Party and make a determination of the matter submitted based solely on the single written statement of each Party. The decision of the Consultant(s) shall be in writing and conclusive and binding on the Sellers and the Buyer and shall be enforceable against the Parties in any court of competent jurisdiction. In determining the proper value attributable to a Title Defect, Environmental Defect or Title Benefit, the Consultant(s) shall not assign a value to a Title Defect or Environmental Defect higher than the value asserted by the Buyer in the applicable Defect Notice nor assign a value to a Title Benefit higher than the value asserted by the Sellers in the applicable Benefit Notice, as applicable. Any amount owing by one Party as a result of such determination by the Consultant(s) will be paid as provided in Section 2.1.4 and Section 2.10 hereof, as applicable. The Consultant(s) shall act as experts for the limited purpose of determining the specific title or environmental dispute presented to them, shall not act as arbitrators, shall not consider, hear or decide any matters except the specific title or environmental disputes presented to them and shall not award damages, interest or penalties (including punitive or exemplary damages, lost profits, consequential, special or indirect damages) to either Party. In addition, the

Consultant(s) shall agree in writing to keep strictly confidential the specifics and existence of any matters submitted as well as all proprietary records of the Parties, if any, reviewed by the Consultants in the process of resolving such disputes.

- 2.2.13 On or before the last day of the Cure Period, subject to obtaining the Buyer's prior written consent, the Sellers may elect to indemnify the Buyer against all Liabilities resulting from an Environmental Defect in lieu of an adjustment to the Purchase Price or curing such Environmental Defect, with such indemnification obligation (i) being limited in time and to the Liabilities necessary to investigate (not including, for the avoidance of doubt, the cost of the Assessment), remediate or otherwise cure such Environmental Defect in the most cost effective manner permissible under Environmental Law and consistent with existing uses of the Real Property Interest or Well for production, processing, transportation or storage of Hydrocarbons, and (ii) excluding any post-Closing exacerbation of or contribution to the Environmental Defect.
- 2.2.14 Notwithstanding anything to the contrary in this Agreement, without limiting the Buyer's right to exclude Properties pursuant to Section 5.1.1 and except with respect to the Special Warranty and the indemnity provided under Section 10.2 as it relates to (a) breaches of the representation in Section 3.9, (b) any Retained Liabilities, or (c) breaches of the covenants in <u>Section 5.1</u> or <u>Section 5.2</u>, this <u>Section 2.2</u> together with <u>Section 2.1</u> is intended to be the sole and exclusive remedy that the Buyer and its Affiliates, as applicable, shall have against the Seller Indemnified Parties with respect to any matter or circumstance relating to (i) title to the Properties; and (ii) Environmental Laws, the release of materials into the environment or protection of the environment or health. Without limiting the Buyer's right to exclude Properties pursuant to Section 5.1.1 and except to the limited extent necessary to enforce the terms of this Section 2.2 together with Section 2.1, and except with respect to the Special Warranty and the indemnity provided under Section 10.2 as it relates to (A) breaches of the representation in Section 3.9, (B) any Retained Liabilities, or (C) breaches of the covenants in Section 5.1 or Section 5.2, the Buyer (on behalf of itself, its Affiliates and their respective insurers and successors in interest) hereby releases, discharges and waives any and all claims and remedies at Law or in equity, known or unknown, whether now existing or arising in the future, contingent or otherwise, against the Seller Indemnified Parties with respect to any matter or circumstance relating to (1) title to the Properties; and (2) Environmental Laws, the release of materials into the environment or protection of the environment or health EVEN IF SUCH CLAIMS OR DAMAGES ARE CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT, EXCLUDING GROSS NEGLIGENCE AND WILLFUL MISCONDUCT), STRICT LIABILITY OR OTHER **LEGAL FAULT**

- **OF THE SELLER INDEMNIFIED PARTIES**. The Parties acknowledge that Sellers have not made and will not make any representation or warranty regarding any matter or circumstance relating to (x) title to the Properties (except for the Special Warranty); and (y) Environmental Laws, the release of materials into the environment or protection of the environment or health (except as expressly provided in <u>Section 3.9</u>), and that nothing in <u>Article 3</u> or otherwise shall be construed as such a representation or warranty.
- 2.3 <u>Special Warranty</u>. The Assignment delivered at Closing will contain a special warranty of title with regard to the Properties whereby each Seller shall warrant Defensible Title to the Wells and Real Property Interests unto the Buyer and its successors and assigns against every Person whomsoever lawfully claiming or to claim the same or any part thereof by, through or under such Seller, but not otherwise, subject, however, to the Permitted Encumbrances (the "**Special Warranty**"). All claims in respect of the Special Warranty are subject to the survival period set forth in <u>Section 13.3</u>.
- 2.4 Preferential Purchase Rights; Required Consents. Within ten (10) Business Days after the Execution Date, the appropriate Seller shall use the Allocated Values to provide any required notifications of a preferential purchase right, right of first refusal or other agreement which gives a Third Party a right to purchase a Property (or any part thereof) (each such preferential purchase right, except for the Total Pref Right, a "PPR"), requesting waivers thereof, in connection with the transactions contemplated hereby and shall send notice letters to each holder of a PPR in accordance with the terms of such PPR. Within ten (10) Business Days after the Execution Date, the appropriate Seller shall send letters seeking all applicable consents to assignment that are applicable to the transactions contemplated hereby including Required Consents, but excluding Customary Post-Closing Consents. Reasonably promptly following their delivery or receipt of same, Sellers will provide Buyer a copy of such written notifications and letters and all other written notices and communications that are sent to or received from any Persons in connection therewith or related thereto. The Sellers shall thereafter until the Closing Date use their commercially reasonable efforts (at no cost to the Sellers, unless Buyer agrees to make any such payment or undertake any such obligation on behalf of Seller) to attempt to obtain such Required Consents and waivers of PPRs and the Buyer shall reasonably assist the Sellers in such effort. Notwithstanding anything contained herein to the contrary, the Sellers shall have no liability to the Buyer for failure to obtain such consents or waivers (to the extent the Sellers comply with their other obligations in this Section 2.4). If, prior to the Closing Date, any Party discovers any consents to assignment or PPRs (applying to the Properties) and for which notices have not been delivered pursuant this Section, then (i) the Party making such discovery shall provide the other Party with written notice of such consents to assignment or PPRs, as applicable, (ii) the Sellers, following delivery or receipt of such written notification, will promptly send notices to the holders of the applicable consents to assignment requesting such consents and notices to the holders of applicable

PPRs in compliance with the terms of such rights and requesting waivers of such rights arising thereunder and (iii) the terms and conditions of this <u>Section 2.4</u> shall apply to the Properties subject to such consents to assignment or preferential rights, as applicable.

- If, prior to the Closing Date, a holder of a PPR has notified the applicable Seller that it elects to exercise its PPR with respect to the Properties to which its PPR applies (determined by and in accordance with the agreement under which the PPR arises), then the Properties covered by that PPR, will not be sold to the Buyer at Closing (subject to the remaining provisions in this Section 2.4) and shall be deemed Excluded Assets and deleted from Exhibits A-1, A-2 or A-3, as applicable, and the other Exhibits and Schedules to this Agreement for all purposes (except in the event such Properties are assigned to Buyer post-Closing pursuant to the terms of Section 2.4.2, in which case such Properties shall be considered among the Properties and included on Exhibits A-1, A-2, or A-3, as applicable), and the unadjusted Purchase Price will be reduced by the Allocated Value (or portion thereof) of the interest in the Properties subject to such PPR. If, as of the Closing Date, a PPR has not been exercised or waived and the time for exercising such PPR has not expired, the Property covered by that PPR will be sold to the Buyer subject to any rights of the holder of the PPR and no adjustment to the Purchase Price will be made with respect thereto and, in the event the holder of any such PPR thereafter exercises such PPR, the Buyer will comply with all of the terms thereof and convey the applicable Property to the holder of the PPR. If, as of the Closing Date, a holder of a Required Consent has not yet delivered such Required Consent and the time for granting such consent has not yet expired, then the Property covered by that Required Consent will not be conveyed to the Buyer at Closing but shall still be considered part of the Properties in accordance with the provisions of Section 2.4.2, adjustments to the Purchase Price with respect to such Property will still be made pursuant to Sections 2.5, 2.6 and 2.7, with respect to such Property, and the Purchase Price will not be reduced as a result of such non conveyance at the Closing.
- 2.4.2 If Properties (or portions thereof) have been excluded from the Properties sold to the Buyer at the Closing due to a pre-Closing exercise of a PPR, and if for any reason the purchase and sale of the Properties covered by such PPR is not or cannot be consummated with the holder of the PPR that exercised its PPR, then the applicable Seller shall so notify the Buyer and within ten (10) Business Days after the Buyer's receipt of such notice, such Seller shall sell, assign and convey to the Buyer and the Buyer shall purchase and accept from such Seller such Properties pursuant to the terms of this Agreement (applied mutatis mutandis) and for the Allocated Value of such Properties described on Exhibit A-1, subject to adjustments in accordance with Sections 2.6 and 2.7. If Properties have been excluded from the Properties sold to the Buyer at the Closing due to a failure to

obtain a Required Consent in accordance with <u>Section 2.4.1</u>, and if a Required Consent has been received or deemed received pursuant to the terms of the underlying agreement on or before the end of the date that is one (1) year after the Closing Date, the applicable Seller shall so notify the Buyer and within ten (10) Business Days after the Buyer's receipt of such notice, such Seller shall assign and convey to the Buyer and the Buyer shall accept from such Seller such Properties, in each case, pursuant to the terms of this Agreement (applied mutatis mutandis) and an instrument substantially the same form as the Assignment. As between the Buyer and the Sellers, with respect to any Property for which a Required Consent has not been obtained by the Closing, (i) the applicable Seller shall hold such Property as nominee for the Buyer, effective as of the Effective Time, (ii) the Buyer shall pay any costs and expenses associated with that Property, and (iii) such Seller shall pay the Buyer any revenues associated with such Property for periods from and after the Effective Time. If any Required Consent has not been received or deemed received on or before the one (1) year anniversary of the Closing Date, the applicable Seller shall no longer hold such Property (a "Nonconsented Interest") as nominee for the Buyer, and each Party shall repay to the other Party any amounts previously paid hereunder in respect of the Nonconsented Interest (including the Allocated Value paid therefor and all other amounts of any adjustments pursuant to Sections 2.5, 2.6 and 2.7 with respect to such Nonconsented Interest), and such Nonconsented Interest will be deemed not to have been conveyed to the Buyer hereunder and shall thereafter be an Excluded Asset.

- 2.4.3 Properties excluded pursuant to this <u>Section 2.4</u> will not be deemed to be affected by Title Defects or Environmental Defects or be subject to <u>Section 2.1</u> and <u>Section 2.2</u> and the Allocated Value of such excluded properties shall not be applied toward the Title Defect Deductible or the Environmental Defect Deductible.
- 2.4.4 The provisions of <u>Schedule 2.4.4</u> shall apply to the Total Pref Right.
- 2.4.5 Notwithstanding anything herein to the contrary, this <u>Section 2.4</u> shall not apply to any of the Contracts set forth on <u>Exhibit D</u>, the transfer of which shall be subject to <u>Section 5.13</u>.
- 2.5 <u>Gas Imbalances</u>. The Purchase Price will be adjusted upward or downward, as applicable, by (i) the net mcf amount of such Seller's aggregate wellhead Gas Imbalances as of the Effective Time multiplied by \$1.50 per mcf (upward for underage and downward for overage); and (ii) to the extent not cash settled monthly, the mmbtu amount of any pipeline Gas Imbalances attributable to such Seller or the Properties as of the Effective Time multiplied by the actual settlement price per mmbtu (upward for over deliveries and downward for under deliveries). For the purposes of the Closing Statement, settlement will be estimated based on the latest available month's imbalance and then a final

settlement of the imbalance will be handled at the later of the Final Statement settlement or the last time the Sellers or their Affiliates markets the Buyer's gas.

- 2.6 <u>Certain Upward Adjustments</u>. The Purchase Price shall be increased by the following (without duplication):
  - 2.6.1 to the extent not previously sold (or otherwise transferred) by Sellers, and except for the Stored CEM NGLs and Stored CEM Oil, the value of all merchantable oil or other liquid Hydrocarbons in storage above the pipeline connection or existing in stock tanks, pipelines or plants (including line fill) owned by the Sellers at the Effective Time that is credited to the Properties with gauging and other customary industry procedures, such value to be the current market price at the Effective Time, less Asset Taxes attributable to such oil or other liquid Hydrocarbons and allocable to the Sellers pursuant to Section 5.6 but economically borne by the Buyer, royalties, marketing fees, and gravity adjustments deducted by the purchaser of such oil or other liquid Hydrocarbons (for purposes of this Section 2.6, "Production Burdens");
  - 2.6.2 the amount of all operating and production expenses (a) incurred in the Ordinary Course of Business and (b) in connection with the ownership, operation and maintenance of the Properties (including capital expenditures, rentals, overhead, royalties, prepayments, lease extensions (provided, that to the extent incurred between the Execution Date and the Closing Date, only to the extent permitted in Section 5.2 or Section 5.3), operating, drilling and completion costs, marketing and transportation costs (in each case whether owed or paid to Affiliates or Third Parties) and, without duplication, any similar charges and expenses billed by Third Parties under applicable operating agreements and in the case of wholly-owned Properties where a joint operating agreement may not exist, overhead rates consistent with those charged by such Seller on other wells in the area) (but excluding, in each instance, the Excluded Property Costs; subject to such exclusion, the "Property Costs"), in each case, paid by or on behalf of the Sellers and attributable to the ownership and operation of the Properties during the period on or after the Effective Time;
  - 2.6.3 the amount of any and all prepaid utilities, rentals, deposits and other prepays (other than Taxes) applicable to the period on or after the Effective Time that are attributable to the Properties and incurred by the Sellers in the Ordinary Course of Business;
  - 2.6.4 the amount of any Asset Taxes allocable to the Buyer pursuant to <u>Section 5.6</u> but paid or payable by a Seller;
  - 2.6.5 the amount as of the Closing Date determined by the number of barrels of natural gas liquids by purity product produced from the Properties and held in storage by CEM as of the Closing Date multiplied by the actual

purity product weighted average monthly settlement price per barrel as of the month of the Closing Date (the "Stored CEM NGLs") (less Production Burdens, if applicable); provided, that, the Closing Statement settlement will be estimated based on the latest available month's volumes and then a final settlement of the volumes will be handled at the later of the Final Statement settlement or the last time the Sellers or their Affiliates markets the Buyer's natural gas liquids pursuant to the TSA;

- 2.6.6 the amount as of the Closing Date determined by the number of barrels of oil produced from the Properties and held in storage by CEM as of the Closing Date multiplied by the monthly NYMEX Cal Month average (excluding weekend and holidays) less the monthly weighted average sales deduct as of the month of the Closing Date (the "Stored CEM Oil") (less Production Burdens, if applicable); provided, that, the Closing Statement settlement will be estimated based on the latest available month's volumes and then a final settlement of the volumes will be handled at the later of the Final Statement settlement or the last time the Sellers or their Affiliates markets the Buyer's oil pursuant to the TSA; and
- 2.6.7 any other amount provided for in this Agreement or agreed upon by the Buyer and the Sellers in writing.
- 2.7 <u>Certain Downward Adjustments</u>. The Purchase Price shall be decreased by the following (without duplication):
  - 2.7.1 the amount of any proceeds received by a Seller (and not remitted to the Buyer) from the sale of Hydrocarbons produced from the Properties from and after the Effective Time to a Person other than a Seller (net of (i) royalties and other similar burdens, (ii) marketing fees paid in the Ordinary Course of Business, and (iii) gravity adjustments deducted by the purchaser of such oil or other liquid Hydrocarbons);
  - 2.7.2 the amount of all Property Costs paid by or on behalf of the Buyer or its Affiliates that are attributable to the ownership, operation or maintenance of the Properties during the period before the Effective Time;
  - 2.7.3 the amount of any Asset Taxes allocable to a Seller pursuant to Section 5.6 but paid or payable by the Buyer;
  - 2.7.4 any amounts delivered to Buyer in accordance with <u>Section 5.7</u>;
  - 2.7.5 with respect to the Stored CEM NGLs and the Stored CEM Oil, the amount of any proceeds received by a Seller (and not remitted to the Buyer) from the sale of Hydrocarbons produced from the Properties from and after the Effective Time that constitutes any part of the Stored CEM NGLs and Stored CEM Oil; and

- 2.7.6 any other amount provided for in this Agreement or agreed upon by the Buyer and the Sellers in writing.
- Closing Date Estimates. On or before five (5) Business Days prior to the Target Closing Date (or, if the Closing does not occur on the Target Closing Date, three (3) Business Days prior to the Closing Date), the Sellers (with the cooperation of the Buyer) will prepare, in accordance with the provisions of this Agreement, and deliver to the Buyer a written statement prepared in good faith (the "Closing Statement") setting forth each adjustment to the Purchase Price (together with reasonable supporting information) required under this Agreement and showing the calculation of such adjustments. Within two (2) Business Days of receipt of the Closing Statement, the Buyer may deliver to Sellers a written report containing all proposed changes to the Closing Statement with an explanation therefor that the Buyer proposed to be made to the Closing Statement. The agreed upon Closing Statement will be used to adjust the Purchase Price at Closing; provided that if the Parties do not agree upon an adjustment set forth in the Closing Statement, the Sellers' proposed adjustment shall apply at Closing. Any final adjustments, if necessary, will be made pursuant to Section 2.9 of this Agreement.
- 2.9 Final Accounting. On or before one hundred and twenty (120) days after the Closing Date, the Sellers (with the cooperation of the Buyer) will prepare, in accordance with the provisions of this Agreement, and deliver to the Buyer a written post-closing statement setting forth a detailed calculation of all final adjustments to the Purchase Price (together with reasonable supporting information) which takes into account all such adjustments provided in this Agreement (except as otherwise set forth in Section 2.1 and Section 2.2) (the "Final Statement"). If the Buyer disputes any items in the Final Statement, then as soon as reasonably practicable, but in no event later than sixty (60) days after its receipt of the Final Statement, the Buyer will deliver to the Sellers a written exception report containing any changes the Buyer proposes to be made to the Final Statement. If the Buyer fails to deliver such exception report to the Sellers within that period, or to the extent an item is not disputed within a timely delivered exception report, the Final Statement (or any undisputed items therein, as applicable) as delivered by the Sellers will be deemed to be true and correct, and binding upon and not subject to dispute by any Party. If the Buyer delivers an exception report, as soon as reasonably practicable, but in no event later than thirty (30) days after the Sellers receive the Buyer's exception report, the Parties will meet and undertake to agree on the final post-Closing adjustments. If the Parties fail to agree on the final post-Closing adjustments within sixty (60) days after the Sellers' receipt of the Buyer's exception report, any Party will be entitled to submit the dispute for resolution by the Accounting Referee. The cost of the Accounting Referee shall be paid **fifty percent** (50%) by the Sellers and **fifty percent** (50%) by the Buyer. Not later than ten (10) Business Days after appointment of the Accounting Referee, the Sellers and the Buyer shall each present to the Accounting Referee, with a simultaneous copy to the other Party, a single written statement of its position on the disputes in question, together with a copy of this Agreement, the Closing Statement, the proposed Final Statement and

the Buyer's written exception report and any supporting material that such Party desires to furnish. In making its determination, the Accounting Referee shall be bound by the terms of this Agreement and, without any additional or supplemental submittals by either Party, may consider such other accounting and financial standards matters as in its opinion are necessary or appropriate to make a proper determination. The Parties shall direct the Accounting Referee to resolve the disputes within thirty (30) days after receipt of the written statements submitted for review and to render a decision in writing based upon such written statements. In determining the proper amount of any adjustment to the Purchase Price, the Accounting Referee shall not increase the Purchase Price more than the increase proposed by the Sellers nor decrease the Purchase Price more than the decrease proposed by the Buyer, as applicable. The Accounting Referee shall act as an expert for the limited purpose of determining the specific Final Statement dispute presented to it pursuant to and limited by the procedures set forth in this Section, shall refrain from ex parte communications with any Party or its representatives, shall not act as an arbitrator, shall not consider, hear or decide any matters except the specific Final Statement disputes presented and shall not award damages, interest or penalties (including punitive or exemplary damages, lost profits, consequential, special or indirect damages) to any Party. In addition, the Accounting Referee shall agree in writing to keep strictly confidential the specifics and existence of any matters submitted as well as all proprietary records of the Parties, if any, reviewed by the Accounting Referee in the process of resolving such disputes. Upon agreement of the Parties to the adjustments to the Final Statement, or upon resolution of such adjustments by the Accounting Referee, as the case may be, the Final Statement (as adjusted pursuant to such agreement or resolution by the Accounting Referee) will be deemed final and binding on all of the Parties, without right of appeal, and enforceable in any court of competent jurisdiction, and the aggregate amount due to either the Buyer or the Sellers pursuant to such Final Statement will be paid in accordance with Section 2.10.

- 2.10 <u>Payments</u>. The Purchase Price as adjusted (and less the Deposit) will be payable in immediately available funds at Closing (pursuant to wire transfer instructions designated in advance by the Sellers to the Buyer in writing), as further described in <u>Section 8.1.1</u>. Except as set forth in <u>Section 2.1.4</u>, payments to be made following the Closing under this <u>Article 2</u> shall be made by wire transfer of immediately available funds within five (5) Business Days after the final determination is made that such payments are due and payable (pursuant to wire transfer instructions designated in advance by the receiving Party to the paying Party in writing) for the account of the receiving Party.
- 2.11 <u>Tax Allocation</u>. The Sellers and the Buyer shall cooperate in the preparation of Internal Revenue Service Form 8594 to report the allocation of the Purchase Price (as determined for U.S. federal income tax purposes) among the Properties. The Sellers shall prepare and deliver to the Buyer a draft Form 8594 within twenty (20) days after the determination of the Final Statement and any further adjustments to the Purchase Price pursuant to this Agreement. If the Buyer

disputes any items in the Sellers' draft Form 8594, then as soon as reasonably practicable, but in no event later than twenty (20) days after receipt of such draft Form 8594, the Buyer will deliver to the Sellers in writing any changes the Buyer proposes to be made to the Form 8594. The Parties agree to use commercially reasonable efforts to reach an agreement on any and all disputed items in the draft Form 8594 within twenty (20) days after the later of the Buyer's receipt of the Sellers' draft Form 8594 and the Sellers' receipt of any written comments timely submitted by the Buyer. In the event the Parties agree to such an allocation, the Parties shall, and shall cause their Affiliates to (i) use such allocation as the basis for reporting asset values and other items for purposes of all federal, state, and local Tax Returns, including Internal Revenue Service Form 8594, (ii) not take a position on any Tax Return, or before any Governmental Authority in connection with the examination of a Tax Return or in any Tax proceeding that is in any manner inconsistent with such allocation, except as required by applicable Law or with the written consent of the other Party, and (iii) furnish the other Party a copy of Form 8594 (Asset Acquisition Statement under Section 1060) proposed to be filed with the Internal Revenue Service by such Party or any Affiliate thereof thirty (30) days prior to such filing.

- 2.12 Casualty Loss. Notwithstanding anything herein to the contrary from and after the Effective Time, if Closing occurs, the Buyer shall assume all risk of loss of the Properties, including all risk of loss with respect to production of Hydrocarbons through normal depletion (including watering out of any Well, collapsed casing or sand infiltration of any Well) and the depreciation of personal property due to ordinary wear and tear, in each case, with respect to the Properties; provided, however, that if after the Execution Date but prior to the Closing Date, any portion of the Properties is destroyed or taken as a result of a Casualty (a "Casualty Loss"), subject to its rights in respect of Section 6.1.5, the Buyer shall nevertheless be required to close and such Casualty Loss shall be treated as a downward Purchase Price adjustment equal to the amount of such Casualty Loss; provided, however, that if the amount of such Casualty Loss exceeds the Allocated Value of the affected Property, Sellers may, at their option, elect to exclude the affected Property from the Closing, in which case the Purchase Price will be adjusted downward by the Allocated Value of the affected Property and the affected Property shall be treated as an Excluded Asset for all purposes hereunder. The "Casualty Loss Amount" shall mean the amount that the Purchase Price is adjusted downward on account of a Casualty Loss, which, in accordance with the terms of this Section 2.12, shall either be (i) the amount of such Casualty Loss or (ii) if Sellers elect to exclude the Property affected by such Casualty Loss, the Allocated Value of such Property.
- 2.13 <u>Additional Consideration</u>. In the event Closing occurs, (a) if the 2022 NYMEX Price equals or exceeds \$3.00/mmbtu for any twenty (20) trading days out of a period of thirty (30) consecutive trading days during the period from (and including) the Execution Date until (and including) December 31, 2019, the Buyer shall pay to the Sellers Fifty Million Dollars (\$50,000,000) (the "2022 Additional Consideration") and (b) if the 2023 NYMEX Price equals or exceeds

**\$3.25**/mmbtu for any twenty (20) trading days out of a period of thirty (30) consecutive trading days during the period from (and including) the Execution Date until (and including) December 31, 2019, the Buyer shall pay to the Seller **Fifty Million Dollars** (**\$50,000,000**) (the "**2023 Additional Consideration**"). The 2022 Additional Consideration and the 2023 Additional Consideration, in each case if required to be paid pursuant to the above, shall be delivered by Buyer to Sellers on or before the day that is sixty (60) days from December 31, 2019 by wire transfer of immediately available funds to the account directed in writing by Sellers. Any amounts owed or paid pursuant to this Section 2.13 shall be a part of the purchase price for the Properties.

### ARTICLE 3 SELLERS' REPRESENTATIONS AND WARRANTIES

- 3. <u>Sellers' Representations and Warranties</u>. Each Seller, jointly and severally with each other Seller, represents and warrants to the Buyer as of the Execution Date, as follows:
  - 3.1 <u>Organization, Good Standing, Etc.</u> Such Seller is a duly formed, validly existing entity in good standing under the Laws of the state of its formation. Such Seller is duly qualified or licensed, as may be required by applicable Law, and in good standing in each jurisdiction it carries on business or owns assets.
  - Authorization. Such Seller has taken all necessary company action to authorize the execution, delivery and performance of this Agreement (and each other agreement, instrument or document executed or to be executed by such Seller in connection with the transaction hereunder) and has adequate company power, authority and legal right to enter into, execute, deliver and perform this Agreement and to consummate the transactions contemplated hereby. This Agreement is legal, valid and binding with respect to such Seller and is enforceable in accordance with its terms, except as the enforceability thereof may be limited by bankruptcy, insolvency or similar Laws affecting creditors' rights generally. Such Seller has all requisite power and authority to own and operate the Properties as currently operated by such Seller and to carry on its business as currently conducted.
  - No Breach. Except as disclosed in Schedule 3.3, and for the CFIUS Approval, compliance with the HSR Act, approval by FERC of the transfer of any FERC regulated Contracts and any Customary Post-Closing Consents, the execution, delivery, performance and consummation of this Agreement does not and will not: (a) violate, conflict with or constitute a default or an event that, with notice or lapse of time or both, would be a default, breach or violation under any term or provision of the governing documents of such Seller; (b) result in a material default, breach or violation or an event that, with notice or lapse of time or both, would be a default, breach or violation under any term or provision of any Contract, promissory note, indenture, mortgage, deed of trust or lease to which such Seller is a party or by which such Seller or its interest in any of the Properties is bound, except for any such violation, conflict or default (i) that has

been waived or consented to by the appropriate counterparty and (ii) for which Buyer shall not take any responsibility whatsoever; (c) violate, conflict with or constitute a breach of in any material respect any Law applicable to such Seller or by which such Seller or its interest in any of the Properties is bound; or (d) except for Permitted Encumbrances, result in the creation, imposition or continuation of any Lien on or affecting such Seller's interest in the Properties.

3.4 <u>Litigation</u>. Except as disclosed in <u>Schedule 3.4</u>, there is no action, suit, claim, cause of action or proceeding pending or, to such Seller's Knowledge, threatened in writing against the Sellers or their Affiliates involving such Seller's interest in the Properties or the transactions contemplated hereby.

### 3.5 <u>Taxes</u>.

- 3.5.1 Except as disclosed in Schedule 3.5, all Asset Taxes based on or measured by such Seller's or any Affiliate's ownership or operation of the Properties or the production or removal of Hydrocarbons or the receipt of proceeds therefrom, have been timely paid when due. All Tax Returns relating to Asset Taxes required to have been filed by such Seller or an Affiliate with respect to the Properties have been duly and timely filed (taking into account any extension of time to file granted or obtained) for all taxable periods with all taxing authorities. Such Tax Returns are true and correct in all material respects and there are no administrative proceedings or lawsuits pending or claims asserted with respect to such Taxes by any Governmental Authority other than for matters disclosed on <u>Schedule 3.5</u>. No claim has ever been made by an authority in a jurisdiction where such Seller does not file Tax Returns that such Seller is or may be subject to taxation in that jurisdiction with respect to ownership or operation of the Properties, and the Properties are not subject to Taxes in any jurisdiction in which such Seller has not filed Tax Returns. Except as disclosed on Schedule 3.5, none of the Properties is held in an arrangement that is treated as a partnership for federal, state or local income tax purposes (each, a "Tax Partnership"). None of the Properties is the subject of any agreement with any Governmental Authority with respect to Taxes that would be terminated or adversely affected as a result of the transactions contemplated by this Agreement. All of the Properties have been properly listed and described on the property tax rolls of the appropriate taxing jurisdiction in which such Properties are located for all periods prior to the Closing Date and none of the Properties constitutes omitted property for property tax purposes.
- 3.5.2 No Tax Partnership disclosed on <u>Schedule 3.5</u> has ever been classified as a corporation for U.S. federal, state, local or non-U.S. income Tax purposes. Any Tax Partnership which remains in existence that is disclosed on <u>Schedule 3.5</u> has a Code Section 754 election in effect. Schedule 3.5 sets forth the following estimates with respect to any Tax Partnership disclosed therein as of December 31, 2017: (a) a good faith estimate of each Seller's

capital account balance for U.S. federal income Tax purposes with respect to the Tax Partnership and (b) a good faith estimate of the total capital account balances for U.S. federal income Tax purposes of all members of the Tax Partnership.

- Permits. Such Seller has all licenses, approvals, authorizations, certifications, clearances, consents, variances, orders, franchises, registrations and permits issued, granted, given, or otherwise made available by or under the authority of any Governmental Authorities necessary or required to permit the operation of the Properties as presently operated by such Seller (excluding those required under Environmental Laws) (the "Permits") and each is in full force and effect and has been duly and validly issued, except where the absence of such, individually or in the aggregate, is not reasonably likely to cause such Seller (or the Buyer) to be liable for material fines, costs or expenses. To such Seller's Knowledge, there are no material violations of any of its Permits. The execution and delivery of this Agreement and the consummation of the transactions contemplated hereby will not result in any revocation, cancellation, suspension or modification of any such Permit except those Permits issued in the name of such Seller that are not transferable under applicable Law.
- 3.7 <u>Compliance with Laws.</u> During the period and to the extent that such Seller or its Affiliates have operated any Properties (and with respect to those periods of time during which Third Parties have operated any of the Properties, to such Seller's Knowledge), the Properties have been operated and are in material compliance with the provisions and requirements of all applicable Laws (excluding Environmental Laws which are addressed in <u>Section 3.9</u>), except for prior instances of non-compliance that have been fully and finally resolved (without any ongoing material obligations to which Buyer or the Properties will be subject from and after the Effective Time) to the satisfaction of all Governmental Authorities with jurisdiction over such matters.
- 3.8 <u>Material Contracts.</u> Such Seller has listed in <u>Schedule 3.8 Part 1</u> all of the Contracts of the type described below (collectively, the "**Material Contracts**") and Easements of the type described below, in each case, that will be binding on the Properties or the Buyer from and after the Effective Time:
  - 3.8.1 all farm-ins, farm-outs, exploration, development, joint venture, participation, purchase or acquisition agreements, joint operating agreements and similar Contracts (including any Contract that grants a Person any option to acquire any Properties or that includes an obligation for a deferred payment);
  - 3.8.2 all area of mutual interest agreements or other agreements that otherwise contain non-competition restrictions (or any agreements that purport to restrict, limit or prohibit such Seller from engaging in any line of business or the manner in which or the locations at which such Seller conducts business) or rights of first refusal, negotiation, or other similar restrictions

- (including "tag" and "drag" rights), in each case, including joint operating agreements;
- 3.8.3 all Hydrocarbon purchase contracts or sale contracts, Hedges, processing contracts, gathering contracts, transportation contracts, marketing contracts, and disposal or injection contracts affecting such Seller's interest in any of the Properties which are not, by the terms thereof, subject to termination without penalty upon sixty (60) days or less notice;
- 3.8.4 any Contract (including joint operating agreements) or Easement that can reasonably be expected to result in aggregate payments by or revenues to any Person of more than **Three Hundred Thousand Dollars** (\$300,000) during the current or any subsequent fiscal year;
- 3.8.5 all production payments, net profits interests and similar rights or interests burdening such Seller's interest in any of the Properties;
- 3.8.6 all Contracts between such Seller and any of its Affiliates which will be binding on the Buyer or the Properties after Closing;
- 3.8.7 all Contracts that contain any calls on, or options to purchase, Hydrocarbon production, other than pursuant to currently effective Hydrocarbon purchase and sale contracts to which the Properties will be subject after Closing;
- 3.8.8 any existing contract that is a seismic or other geophysical acquisition or sharing agreement or license for which the Buyer will be liable;
- 3.8.9 any Contract that constitutes a lease under which such Seller is the lessor or the lessee of real or personal property that cannot be terminated by such Seller (without penalty) upon sixty (60) days or less notice, which will be binding on Buyer or on any of the Properties after Closing, and which involves an annual base rental of more than **Three Hundred Thousand Dollars** (\$300,000);
- 3.8.10 all Contracts the sole purpose of which (as of the execution of such Contract) is for such Seller to indemnify another Person and which will be binding on Buyer or on any of the Properties after Closing; and
- 3.8.11 all Contracts that contain mandatory drilling requirements with respect to the Properties (other than provisions requiring optional drilling as a condition of maintaining or earning all or a portion of a presently non-producing Real Property Interest).

Sellers have made available to Buyer in the FTP Site complete copies of all of their Material Contracts in their possession prior to the Execution Date, to the extent it has the right under such Contracts to Buyer. Except as separately described on <u>Schedule 3.8 – Part 2</u>, to such Seller's

Knowledge, (a) all of its Material Contracts are in full force and effect and are legal, valid and binding obligations of such Seller and each party thereto, (b) such Seller is not and no other party is in material default or material breach under any contract listed in Schedule 3.8 – Part 1 except as disclosed in Schedule 3.8 – Part 2; and (c) no event has occurred that with notice or lapse of time or both would constitute any material default or material breach under any such Material Contract by such Seller or any other party thereto. Except as set forth on Schedule 3.8 – Part 2, no currently effective written notices have been received by such Seller of the exercise of any premature termination, price redetermination, market-out, shut-in or curtailment of or under any of its Material Contracts. To such Sellers' knowledge, no notice of a material default or material breach has been received or delivered by such Seller under any Material Contract, the resolution of which is still outstanding. There are no material oral agreements that will be binding on the Buyer or the Properties after Closing. The Total Development Agreement (not including, for the avoidance of doubt, the joint operating agreements entered into pursuant to the Total Development Agreement) will not be binding on the Properties following their assignment to Buyer and will not otherwise apply to (or be binding on) the Buyer.

- 3.9 Environmental and Safety Matters. Except as disclosed on Schedule 3.9, (a) except for any noncompliance that has been remediated in accordance with applicable Environmental Law, during the period that such Seller has operated any of the Properties and, to such Seller's Knowledge, during the period any other Person operated any of the Properties, such Properties have been operated in and have been in material compliance with applicable Environmental Laws; (b) such Seller has not nor (to such Seller's Knowledge) has any Affiliate of such Seller, received written notice from any Governmental Authority of any material release, spill, disposal, event, condition, circumstance, activity, practice or incident concerning any land, facility, asset or property included in the Properties that (i) materially interferes with or prevents compliance by such Seller with any Environmental Law or the terms of any license or Permit issued pursuant thereto, or (ii) gives rise to or results in any material liabilities under Environmental Law, which remains pending or unresolved; and (c) there are no actions, suits, or proceedings pending or, to such Seller's Knowledge, threatened in writing with respect to its Properties alleging material violations of Environmental Laws, or claiming remediation obligations or other liabilities under applicable Environmental Laws. Notwithstanding any other provisions of this Agreement, the representations and warranties in this Section 3.9 shall be the sole and exclusive representations and warranties with respect to environmental matters and/or Environmental Laws and no other representation or warranty appearing in the Agreement shall be constructed to cover any environmental matters or Environmental Laws.
- 3.10 <u>Broker's or Finder's Fees</u>. Such Seller has not incurred any liability, contingent or otherwise, for brokers' or finders' fees in respect of the transactions contemplated by this Agreement for which the Buyer will have any responsibility whatsoever.

- 3.11 <u>Bankruptcy.</u> There are no bankruptcy, reorganization or arrangement proceedings pending, being contemplated by or to such Seller's Knowledge threatened against the Sellers or any Affiliate of the Sellers. Such Seller is not now insolvent and will not be rendered insolvent by any of the transactions contemplated by this Agreement. As used herein, "insolvent" means that the sum of a Seller's debts and other probable liabilities exceeds the present fair saleable value of such Seller's assets (after giving effect to amounts that could reasonably be received by reason of indemnity, offset, insurance or other similar arrangement). Immediately after giving effect to the consummation of the transaction contemplated by this Agreement, such Seller will not have unreasonably small capital with which to conduct their present or proposed business, will not have incurred or intended to incur, and does not believe that it will incur, debt beyond its ability to pay such debt as such debt matures, and will have assets (calculated at fair market value) that exceed their liabilities. The interests contemplated to be transferred, conveyed and sold by Sellers pursuant to the terms set forth in this Agreement and in the Assignment are being sold for reasonably equivalent value.
- 3.12 <u>PPRs and Required Consents</u>. Except as disclosed in <u>Schedule 3.12</u> or <u>Exhibit D</u>, no interest of such Seller in a Property is subject to any (a) PPR or (b) Required Consent of any Third Party to the sale and conveyance of such Seller's interest in the Properties as provided for in this Agreement (except for Customary Post-Closing Consents).
- 3.13 Gas Balancing. Schedule 3.13 sets forth all Gas Imbalances as of the respective date(s) shown thereon.
- 3.14 <u>Hedges</u>. Except for the Natural Gas Hedges, there are no Hedges with respect to the sale of Hydrocarbons from any of the Properties that will be binding on the Properties after Closing.
- 3.15 <u>Current Commitments.</u> <u>Schedule 3.15</u> sets forth, as of the Execution Date, all outstanding authorizations for expenditures and other binding capital commitments (each, an "**AFE**") requiring, individually, an expenditure greater than **Three Hundred Thousand Dollars** (\$300,000.00) (net to such Seller's interest or the Sellers' interests in the aggregate) relating to the Sellers' ownership of the Properties. <u>Schedule 3.15</u> also sets forth any proposed AFEs requiring, individually, an expenditure greater than **Three Hundred Thousand Dollars** (\$300,000.00) (net to such Seller's interest or the Sellers' interests in the aggregate) relating to the Sellers' ownership of the Properties that have been proposed, but, as of the Execution Date, have not yet been approved or rejected (or deemed approved or rejected) under the applicable Contract. The foregoing statements shall be deemed qualified to such Seller's Knowledge with respect to any of the Properties for which such Seller does not serve as operator.
- 3.16 <u>Unleased AR</u>. To such Seller's Knowledge, <u>Schedule 3.16</u> sets forth the amount of Unleased AR as of the date set forth on such schedule.

- 3.17 <u>Leases</u>. Such Seller is not (and such Seller, to its Knowledge, has not received written notice of any other Person being) in material default or breach under any of the Real Property Interests, and no event has occurred that, with notice or lapse of time or both, would constitute a default or breach by such Seller. To Seller's Knowledge and except as listed on <u>Schedule 3.4</u>, (a) no royalty owner is currently performing an audit regarding the payment of any royalties under the Real Property Interests or any similar payment and (b) no Real Property Interest contains any express provisions obligating Seller to drill any wells on the Lands (other than provisions requiring optional drilling as a condition of maintaining or earning all or a portion of a presently non-producing Real Property Interest).
- 3.18 Wells; Equipment. Except as set forth on Schedule 3.18, (a) to such Seller's Knowledge, all Wells have been drilled and completed within the limits permitted by all applicable Real Property Interests, Units, Mineral Interests and Material Contracts or within the limits otherwise permitted by applicable Law, (b) no Well operated by such Seller or its Affiliates, and to such Seller's Knowledge no other Well, is subject to penalties on allowables after the Effective Time because of any overproduction or other violation of Laws; (c) there are no Wells in respect of which such Seller has received an order in writing from a Governmental Authority requiring that such Well be plugged and abandoned; (d) to such Seller's Knowledge all currently producing (or, in the case of shut-in Wells, capable of producing) Wells, Compressor Assets and Equipment are in an operable state of repair adequate to maintain normal operations in accordance with past practices in the Ordinary Course of Business, ordinary wear and tear excepted; and (e) such Seller holds good and marketable title to the Equipment and the Compressor Assets, free and clear of all Liens other than Permitted Encumbrances.
- 3.19 Operations; Take or Pay. Such Seller has not received any written notice of any vote (or other action) being taken (or any written notice threatening to commence such vote or other action) to remove such Seller or any of its Affiliates as the operator of any of the Properties. Such Seller is not obligated by virtue of any take or pay payment, advance payment or other similar payment (other than gas balancing arrangements) to deliver Hydrocarbons produced from or allocated to the Properties at any time after the Effective Time without receiving payment therefor.
- 3.20 <u>Suspended Funds</u>. <u>Schedule 3.20</u> sets forth, as of the dates set forth therein, all Suspended Funds.
- 3.21 <u>Payout Balances</u>. To such Seller's Knowledge, except for those Wells set forth on Part I of <u>Schedule 3.21</u>, Part II of <u>Schedule 3.21</u> contains a list, which is complete and accurate, of the status of the Payout Balance, as of the dates set forth therein, for each Well. As used herein, "**Payout Balance**" means the status, as of the date of the calculations, of the recovery by such Seller or a Third Party of a cost amount specified in the contract relating to a well out of the revenue from such well where the NRI of such Seller therein will be reduced or such Seller's WI therein will be increased when such amount has been recovered.

- 3.22 <u>Governmental Consent</u>. No consent, approval, or authorization of, or designation, or filing with, any Governmental Authority is required on the part of such Seller in connection with the valid execution and delivery of this Agreement or the consummation of transactions contemplated hereby, except for the CFIUS Approval, compliance with the HSR Act, approval by FERC of the transfer of any FERC regulated Contracts and any Customary Post-Closing Consents.
- 3.23 <u>Letters of Credit, Bonds and Guarantees</u>. To such Seller's Knowledge, except as set forth on <u>Schedule 3.23 and</u> except for such Seller's area-wide bonds, permits and licenses or other permits, licenses or authorizations used in the conduct of such Seller's business generally, there are no letters of credit, bonds, guarantees or other similar forms of credit support posted with Persons relating solely to such Seller's ownership or operation of the Properties.
- 3.24 <u>Absence of Certain Changes</u>. Since the Effective Time up to the Execution Date, to such Seller's Knowledge, there has not been any material Casualty Loss with respect to the Properties.

# ARTICLE 4 BUYER'S REPRESENTATIONS AND WARRANTIES

- 4. <u>Buyer's Representations and Warranties</u>. The Buyer represents and warrants to the Sellers as of the Execution Date as follows:
  - 4.1 <u>Organization and Standing</u>. The Buyer is duly formed and in good standing under the respective Laws of the state of its formation. The Buyer has (or, as of Closing, will have) the power and authority to acquire and own the Properties and to conduct business in the state of Ohio.
  - 4.2 <u>Powers</u>. The Buyer is duly authorized and empowered to execute, deliver and perform this Agreement and to consummate the transactions contemplated hereby. Neither the certificate of incorporation nor the bylaws of the Buyer, nor any other instrument to which the Buyer is a party or is bound, nor any court order or Law, rule or regulation, will be violated by the Buyer's execution and consummation of this Agreement.
  - 4.3 <u>No Restriction</u>. The Buyer is not subject to any order, judgment or decree, or the subject of any litigation, claim or proceeding, pending or threatened, or any other restriction of any kind or character known to the Buyer, which would affect the Buyer's ability to carry out the transactions contemplated by this Agreement.
  - 4.4 <u>Authorization</u>. The Buyer has taken all necessary action to authorize the execution, delivery and performance of this Agreement and has adequate power, authority and legal right to enter into, execute, deliver and perform this Agreement and to consummate the transactions contemplated hereby. This Agreement is legal, valid and binding with respect to the Buyer and is enforceable in accordance with its terms, except as the enforceability thereof may be limited

by bankruptcy, insolvency or similar Laws affecting creditors' rights generally.

- 4.5 <u>Non-Contravention</u>. The execution, delivery, performance and consummation by the Buyer of the transactions contemplated hereby will not (a) violate any provision of the certificate of formation or any other governing document of the Buyer, or (b) breach or violate, or result (with the giving of notice or the lapse of time or both) in the breach, violation, acceleration or termination of, any contract, indenture, Lien, note, lease, agreement, license or Law to which the Buyer is subject or by which any of its assets are bound or subject, except, with respect to any such breach, violation, acceleration or termination which would not prevent the consummation of the transactions contemplated hereby by the Buyer or result in the Sellers incurring any loss or liability therefrom.
- 4.6 <u>Governmental Consent</u>. No consent, approval, or authorization of, or designation, or filing with, any Governmental Authority is required on the part of the Buyer in connection with the valid execution and delivery of this Agreement or the consummation of transactions contemplated hereby, except for the CFIUS Approval, compliance with the HSR Act, approval by FERC of the transfer of any FERC regulated Contracts and any Customary Post-Closing Consents.
- 4.7 <u>Litigation, Etc.</u> There are no actions, proceedings, or investigations pending, or to the Buyer's knowledge, any basis or threat thereof, which could reasonably be likely to materially impair the Buyer or its ability to consummate the transactions contemplated hereby.
- 4.8 <u>Broker's or Finder's Fees</u>. The Buyer has not incurred any liability, contingent or otherwise, for brokers' or finders' fees in respect of the transactions contemplated by this Agreement for which the Sellers will have any responsibility whatsoever.
- 4.9 <u>Bankruptcy</u>. There are no bankruptcy, reorganization or arrangement proceedings pending, being contemplated by or to the Buyer's knowledge threatened against the Buyer or any Affiliate of the Buyer.
- 4.10 <u>Qualifications</u>. The Buyer is (or, as of the Target Closing Date and the Closing Date, will be) qualified with all applicable Governmental Authorities to own and operate the Properties.
- 4.11 <u>Funding; Investment</u>. The Buyer has, or will have at the Target Closing Date and the Closing Date, available (through cash on hand or existing credit arrangements or otherwise) all of the funds necessary for the acquisition of all of the Properties pursuant to this Agreement, as and when needed, and to perform its obligations under this Agreement. The Buyer (a) is sophisticated in the evaluation, purchase, ownership and operation of oil and gas properties and related facilities and is aware of the risks associated with the purchase, ownership and operation of such properties and facilities, (b) is capable of evaluating, and hereby acknowledges that it has so evaluated, the merits and risks of the Properties, ownership and operation thereof and its obligations hereunder, and (c) is able to bear the

economic risks associated with the Properties, ownership and operation thereof and its obligations hereunder. In making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, but, in all cases, subject and without prejudice to the Buyer's express rights and remedies set forth in this Agreement, the Buyer (i) has relied or shall rely solely on the terms and conditions of this Agreement and its own independent investigation and evaluation of the Properties and the advice of its own legal, Tax, economic, environmental, engineering, geological and geophysical advisors and acknowledges and agrees that, except for the representations and warranties set forth herein and confirmed in the certificate delivered at Closing by the Sellers pursuant to Section 8.2.4, (A) it has not been induced by and has not relied upon any representations, warranties or statements, whether express or implied, made at any time by the Sellers or any of their respective directors, officers, shareholders, employees, Affiliates, controlling Persons, agents, advisors or representatives or any other Person, whether or not any such representations, warranties or statements were made in writing or orally, (B) neither Seller nor any of their respective directors, officers, shareholders, employees, Affiliates, controlling Persons, agents, advisors or representatives or any other Person makes or has made any representation or warranty, either express or implied, as to the accuracy or completeness of any of the information provided or made available to the Buyer or its directors, officers, employees, Affiliates, controlling Persons, agents or representatives, including any information, document or material provided or made available, or statements made or provided to the Buyer (including its directors, officers, employees, Affiliates, controlling Persons, agents or representatives) in connection with the transactions contemplated by this Agreement, including any such information contained in or provided in "data rooms", management presentations or supplemental due diligence information provided by the Sellers or discussions or access to management of the Sellers; and (C) the information referred to in (B) above may include certain projections, estimates and other forecasts and plans and that there are uncertainties inherent in attempting to make such projections, estimates and other forecasts and plans and the Buyer is familiar with such uncertainties and takes full responsibility for making its own evaluation of the adequacy and accuracy of all such projections, estimates and other forecasts and plans and any use or reliance by the Buyer on such information referred to in (B) above is (or the projections, estimates and other forecasts and plans that may be contained therein) at the Buyer's sole risk; (ii) has satisfied or shall satisfy itself through the protections provided to the Buyer herein and its own due diligence as to the environmental and physical condition of and contractual arrangements and other matters affecting the Properties; and (iii) except as expressly set forth herein, agrees to the fullest extent permitted by Law that no Seller nor any of their respective directors, officers, employees, Affiliates, controlling Persons, agents or representatives shall have any liability or responsibility whatsoever to the Buyer or its directors, officers, employees, Affiliates, controlling Persons, agents or representatives on any basis (including in contract or tort, under Federal or state securities laws or otherwise) resulting from the distribution to the Buyer or the Buyer's use of any of the information referred to in clause (i)(B) above.

## ARTICLE 5 COVENANTS

- 5. <u>Covenants</u>. The Sellers and the Buyer hereby covenant and agree to perform the following:
  - 5.1 Access to Information.
    - Insofar as related to the Properties, from and after the Execution Date, the Sellers will give the Buyer and the Buyer's agents and representatives, reasonable access to the Properties to conduct, or cause to be conducted, a due diligence investigation of the Properties and to determine whether any Title Defects or Environmental Defects exist including: (a) a Phase I Environmental Assessment of all or any portion of the Properties operated by the Sellers; (b) a Phase II Environmental Assessment of the Properties, subject to the provisions of this Section 5.1.1; (c) visual inspections; and (d) a review of the Records, including the right to copy, at the Buyer's expense, the Records in the Sellers' possession (the "Assessment"); provided that the Sellers shall not be required to provide any of the foregoing information to the extent that the Sellers are prohibited by any Third Party agreement or Law from sharing such information with the Buyer; provided, however, that the Sellers shall use commercially reasonable efforts to obtain waivers to such prohibitions (provided further that "commercially reasonable efforts" shall not require the Sellers to incur any liability, costs or expenses that Buyer does not agree in writing to pay). Notwithstanding anything herein to the contrary, the Buyer shall not have access to, and shall not be permitted to conduct, any on-site due diligence (including any Phase I Environmental Assessment or Phase II Environmental Assessment) with respect to any Properties where the Sellers do not have the authority to grant access for such due diligence; provided that Sellers shall use commercially reasonable efforts to seek and obtain any and all required authorizations for Buyer to conduct on-site due diligence with respect to such Properties. The Assessment and all of the Buyer's activity conducted under this <u>Section 5.1.1</u> shall be subject to the indemnity provisions of <u>Section 5.1.2</u>. If a Phase I Environmental Assessment recommends that a Phase II Environmental Assessment is necessary in order for the Buyer to prove the existence of an Environmental Defect or establish an Environmental Defect Amount with respect to the applicable Property, the Buyer may (i) furnish the Sellers with a written description of the proposed scope of such Phase II Environmental Assessment, including a description of the specific activities to be conducted, and a description of the approximate location and expected timing of such activities (a "Phase II Request"), (ii) provide Sellers with a copy of the Phase I Environmental Assessment recommending a Phase II Environmental Assessment and (iii) subject to

the preceding sentence, request the prior written consent of Seller to undertake the Phase II Environmental Assessment described in the Phase II Request. In the event that (A) the Sellers do not approve any such Phase II Request, such approval subject to Sellers' sole and absolute discretion, on any Property operated by a Seller or its Affiliates within five (5) Business Days of the Buyer's delivery of such Phase II Request, the Buyer shall have the right, but not the obligation, to exclude any or all of the affected Properties from the transactions contemplated hereby, and any such Properties that are so excluded shall be deemed hereunder to be Excluded Assets and the Purchase Price shall be adjusted downward by the aggregate of the Allocated Values thereof or (B) if the Sellers provide their approval of the Phase II Request and (1) the Phase II Environmental Assessment is completed prior to Closing, then terms of Article 2 shall apply mutatis mutandis with respect to the affected Properties or (2) the Phase II Environmental Assessment is to be completed within sixty (60) days after Closing, then (w) the Properties (or portion thereof) subject to the Phase II Environmental Assessment shall be excluded from the Closing and the Purchase Price shall be reduced by the Allocated Value of the affected Properties; (x) the Buyer shall deposit with the Escrow Agent the Allocated Value of the affected Properties withheld from Closing and the applicable provisions in this Agreement with respect to unremediated Environmental Defects or disputed Environmental Defects will apply mutatis mutandis to the amounts deposited by the Buyer in the Escrow Account; (y) the Closing, the Defect Notice Date and Cure Period (in each case, with respect to the affected Properties only) and all other associated dates herein will be extended for sixty (60) days in order for Buyer to perform and complete its Phase II Environmental Assessment; and (z) upon completion of such Phase II Environmental Assessment, the treatment of such affected Properties shall be determined in accordance with the terms and conditions of Article 2 (applied mutatis mutandis). Except as otherwise set forth herein, the Assessment and Buyer's right of access granted under this <u>Section 5.1.1</u>, shall not, without the prior written consent of the Sellers, entitle the Buyer to operate equipment or conduct testing or sampling of any kind. The Sellers have the right to be present during any activities conducted on the Properties as part of the Assessment. The Buyer shall coordinate its Assessment with the Sellers to minimize any inconvenience to or interruption of the conduct of business by the Sellers. The Buyer shall abide by the Sellers', and any Third Party operator's, safety rules, regulations and operating policies while conducting its due diligence evaluation of the Properties, including the Assessment; provided, that such rules, regulations and operating policies are provided to the Buyer prior to such evaluation. The Buyer shall promptly provide the Sellers (but in any case no later than the Defect Notice Date) copies of all written reports prepared or compiled by the Buyer or any of its representatives or agents in connection with the Assessment. Neither Party shall be deemed by their delivery or receipt, as

applicable, of said documents or otherwise to have made any representation or warranty, expressed, implied or statutory, as to the condition of the Properties or the accuracy of said documents or the information contained therein. The Buyer shall hold all information or data provided or made available by the Sellers or obtained by the Buyer as part of the Assessment confidential pursuant to the terms and conditions of the Confidentiality Agreement and shall not use any of the same except in connection with the transactions set forth in this Agreement. In the event this Agreement is terminated prior to Closing, the Buyer shall return to the Sellers (or certify the destruction of) all copies of all such information and data, as well as any derivative reports, analysis or other items derived or based on any of such information or data. During all periods that the Buyer or any of its representatives are on the Properties, the Buyer shall maintain, at its sole expense and with reputable insurers, such insurance as is reasonably sufficient to support the Buyer's indemnity obligations under Section 5.1.2.

- 5.1.2 The Buyer hereby releases and agrees to indemnify, defend and hold harmless the Seller Indemnified Parties from and against any and all Liabilities (including court costs, expert fees and reasonable attorneys' fees), including Liabilities attributable to personal injuries, death, or property damage, arising out of or relating to the Assessment or access to the Properties by the Buyer, EVEN IF CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT), STRICT LIABILITY OR OTHER LEGAL FAULT OF ANY INDEMNIFIED PERSON EXCLUDING, HOWEVER, ANY CLAIMS, LIABILITIES, LOSSES, COSTS OR EXPENSES (I) CAUSED BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF ANY INDEMNIFIED PERSON OR (II) RESULTING FROM ENVIRONMENTAL CONDITIONS OR DEFECTS DISCOVERED OR UNCOVERED BY BUYER OR BUYER'S REPRESENTATIVES TO THE EXTENT (A) SUCH CONDITIONS OR DEFECTS EXISTED PRIOR TO THE BUYER'S ACCESS TO THE PROPERTIES AND (B) WERE NOT EXACERBATED BY BUYER OR BUYER'S REPRESENTATIVES. All information obtained by and access granted to the Buyer and its representatives under this Section 5.1.2 shall be subject to the terms of Section 13.20 and the terms of the Confidentiality Agreement.
- 5.2 <u>Conduct of Business</u>. From and after the Execution Date until Closing, the Sellers shall own and conduct their business related to the Properties in the Ordinary Course of Business. Without limiting the foregoing, except as (a) required by Law or Permit, (b) as necessary to maintain a Real Property Interest or (c) specifically contemplated in this Agreement (including pursuant to <u>Section 5.3</u>), from and after the Execution Date until Closing, each Seller:
  - 5.2.1 Without the prior consent of Buyer, will not convey, encumber, abandon or otherwise dispose of any part of such Seller's interest in the Properties, other than (a) the sale of Hydrocarbons or obsolete machinery and equipment, in each case, in the Ordinary Course of Business,

- (b) in accordance with (i) applicable PPRs pursuant to <u>Section 2.4</u> or (ii) applicable area of mutual interest agreements or (c) assignments to co-working interest or other beneficial interest owners in the Properties, to the extent, in the case of <u>clauses (b)(ii)</u> and (c), they do not individually or in the aggregate reduce the Sellers' Net Revenue Interest or Sellers' Net Acres or increase the Sellers' Working Interest (without at least a proportionate corresponding increase in the Sellers' Net Revenue Interest) in any Property from that described in Exhibit A-1.
- 5.2.2 Will not (a) amend or change any material terms of any material Real Property Interest, Contract, Permit, or Easement relating to the Properties in any material respect, (b) take any action to terminate, waive or surrender any material rights under any material Real Property Interest, Contract, Permit or Easement (including operatorship), or (c) enter into any material agreement, contract or commitment that, if such agreement, contract or commitment had been entered into prior to the Execution Date, would be required to be listed in <a href="Schedule 3.8">Schedule 3.8</a> attached to this Agreement, in each case, without first consulting with the Buyer and attempting in good faith to reach a mutually acceptable agreement with the Buyer regarding such agreement, contract or commitment; <a href="provided">provided</a>, <a href="however">however</a>, that the Buyer's consent will be required for any such action that will have an adverse effect on the Properties in any material manner or respect.
- 5.2.3 Except (x) as set forth on Part 5.2.3 of Schedule 5.2, (y) for operations covered by the AFEs set forth on Schedule 3.15 or (z) as required in the event of an emergency to protect life, property, or the environment, will not (i) propose any new drilling, reworking, or similar operations with respect to the Properties or (ii) elect to participate or not to participate in any new drilling, reworking, or similar operation reasonably anticipated to require future capital expenditures by the new owner of the Property to be incurred in excess of **Three Hundred Thousand Dollars** (\$300,000.00), chargeable to the Sellers' interests, in each case, without first consulting with the Buyer and attempting in good faith to reach a mutually acceptable agreement with the Buyer regarding such agreement, contract or commitment.
- 5.2.4 Will use commercially reasonable efforts to conduct drilling, completion and well readiness operations with respect to the Properties, in anticipation of Buyer's plan to conduct a one (1) rig drilling program as of September 15, 2018, and Buyer's plan to ramp up to a two (2) rig drilling program starting on December 1, 2018.
- 5.2.5 Will maintain insurance coverage on the Properties in the amount and of the types maintained by such Seller on the Execution Date, and not make

- any election to be excluded from any coverage by an operator for the joint account pursuant to a joint operating agreement or similar Contract.
- 5.2.6 Will not settle or compromise any actions, suits, claims, causes of action or proceedings relating to the Properties that individually or in the aggregate exceed **Three Hundred Thousand Dollars (\$300,000.00)** without Buyer's consent, except in connection with the Retained Liabilities or other matters for which Sellers owe an indemnity hereunder.
- 5.2.7 Will use commercially reasonable efforts to maintain in full force and effect all of its Real Property Interests, Permits and Easements located in the area described in Part 5.2.7 of Schedule 5.2.
- 5.2.8 Will give written notice to Buyer as soon as practicable, but in any event within ten (10) Business Days of such Seller acquiring Knowledge thereof, of (a) the receipt or delivery by such Seller of any written notice with respect to any material breach of any Real Property Interest, Material Contract, Easement, or Permit, (b) the receipt or delivery by such Seller of any written claim for damages or any suit, action, proceeding or litigation made by or against such Seller or its Affiliates with respect to the Properties, or (c) the occurrence of any Casualty Loss with respect to the Properties.
- 5.2.9 Will maintain or cause its Affiliates to maintain the books of account and records and will prepare and timely file any Tax returns and timely pay any Taxes relating to its Properties in the Ordinary Course of Business.
- 5.2.10 Will provide the Buyer with the following reports and updates with respect to the Properties on a mutually agreeable frequency in electronic format where possible; <u>provided</u> that Buyer may request interim updates (to such reports and updates) if there are material changes or as reasonably needed, so long as the provision of such reports does not cause an undue burden on Sellers' conduct in the Ordinary Course of Business:
  - A. Daily detailed drilling and completion reports, including end of well summaries;
  - B. Flowback reports;
  - C. Post job frac reports;
  - D. Bit/BHA records;
  - E. Mud reports;
  - F. Pason records or equivalent surface drilling data (WOB, ROP, torque, pump rates);
  - G. Mud motor specs;
  - H. End of well directional drilling reports & final directional surveys;
  - I. Geosteering reports;
  - J. All logs (open hole, cased hole, MWD) and mudlog data;
  - K. Wellbore schematics:

- L. Cost updates (AFE vs Actual);
- M. AFEs:
- N. Permits and surveys;
- O. Daily production and downtime reports;
- P. Monthly production and downtime reports;
- Q. Monthly injection reports including volumes and pressures;
- R. SIMOPs plans;
- S. Drilling and completion procedures;
- T. Approved state and federal permits and reports;
- U. DST reports, well test reports, oil/water/gas sample analyses, if applicable;
- V. Microseismic reports, if applicable;
- W. Artificial lift designs, if applicable;
- X. Pressure surveys;
- Y. Core and/or cuttings data, if applicable;
- Z. Monthly Oil, Gas, NGL settlement statements;
- AA. Monthly plant statements;
- BB. Monthly operated and non-Operated well-level LOS statements with gas gathering system identifier;
- CC. Monthly purchaser detail;
- DD. Monthly position report for all products including all natural gas balances;
- EE. Reconciliation summary each month of products produced and products sold;
- FF. Weekly drill schedule; and
- GG. Weekly well-readiness schedule.
- 5.2.11 Will not enter into any agreement committing to take any action prohibited by the foregoing, without the prior written consent of the Buyer.

Requests for approval of any action restricted by this <u>Section 5.2</u> shall be delivered to the following individual, who shall have full authority to grant or deny such requests for approval on behalf of Buyer:

Name: Michael Magilton

Email: michael.magilton@encinoenergy.com

Phone: 281-254-7043

Buyer's approval of any action restricted by this <u>Section 5.2</u> shall be considered granted within five (5) Business Days of a Seller's written notice to Buyer requesting such consent, unless Buyer notifies such Seller in writing to the contrary during that period. Notwithstanding the foregoing provisions of this <u>Section 5.2</u>, in the event of an emergency, each Seller may take such action as is reasonably necessary to either protect the Properties or comply with applicable Law or Contracts, and shall notify Buyer of such action promptly thereafter. Buyer's consent to any action requested pursuant to this <u>Section 5.2</u> by Sellers shall not be unreasonably withheld, conditioned or delayed.

- 5.3 <u>Lease Extensions</u>. The Sellers shall, from the Execution Date until the Closing Date, use commercially reasonable efforts to obtain lease extensions with respect to the Real Property Interests set forth on <u>Schedule 5.3</u>, and, notwithstanding anything herein to the contrary, shall be permitted to do so without the consent of Buyer to the extent that any such lease extension does not exceed the amount(s) set forth in <u>Schedule 5.3</u>.
- Non-Operated Properties. The Buyer acknowledges that the Sellers own undivided interests in certain of the interests comprising the Wells and Real Property Interests with respect to which it is not the operator, and the Buyer agrees that the acts or omissions of any working interest owner (including any operator) or any other Person who is not a Seller or an Affiliate of any Seller shall not constitute a breach of the provisions of Section 5.2, and no action required by a vote of working interest owners shall constitute such a breach so long as Sellers have voted their interest in a manner that complies with the provisions of Section 5.2.
- 5.5 <u>Revenues</u>. For all purposes including the Purchase Price adjustments under <u>Article 2</u> of this Agreement, the Sellers and the Buyer will properly allocate revenues before and after the Effective Time and will make payments to each other to the extent necessary for such proper allocation.
  - 5.5.1 Following the Closing, in the event either (a) the Buyer receives production or other revenues attributable to any of the Properties for any periods prior to the Effective Time or (b) a Seller receives production or other revenues attributable to any of the Properties for any periods after the Effective Time, in either case, which is not accounted for in the Closing Statement or the Final Statement, the receiving Party will hold such revenues for the exclusive benefit of the Party entitled thereto and, if not taken into account for purposes of the Closing Statement or the Final Statement, will pay any such amounts due to such Party within thirty (30) days after receipt.
  - 5.5.2 The Sellers shall be responsible for (and entitled to any refunds of) (i) Property Costs incurred up to the Effective Time and (ii) the Excluded Property Costs. The Buyer shall be responsible for (and entitled to any refunds of) Property Costs incurred from and after the Effective Time. Should the Buyer pay any Property Costs or Excluded Property Costs for which the Sellers are responsible under this Section 5.5 and that is not otherwise accounted for in connection with the preparation and calculation of the Final Statement pursuant to Section 2.9, the Sellers shall reimburse the Buyer promptly after receipt of the Buyer's invoice therefor accompanied by copies of the relevant vendor or other invoice or proof of payment. Should the Sellers pay any Property Costs for which the Buyer is responsible under this Section and that is not otherwise accounted for in connection with the preparation and calculation of the Final Statement pursuant to Section 2.9, the Buyer shall reimburse the Sellers promptly

after receipt of the Sellers' invoice therefor accompanied by copies of the relevant vendor or other invoice or proof of payment; <u>provided</u>, <u>however</u>, that, notwithstanding anything else to the contrary under this <u>Section 5.5</u>, Seller shall have no further obligations or other responsibility for any Property Costs incurred by Buyer with respect to the Properties following the day that is one (1) year following Closing.

5.5.3 Notwithstanding anything to the contrary herein, the Buyer shall not have any responsibility or other obligation for any Excluded Property Costs, for which Sellers shall be solely responsible.

### 5.6 Tax Matters.

- 5.6.1 Each Seller shall be allocated and bear, and indemnify and hold harmless Buyer for, all Asset Taxes relating to its Properties that are attributable to any Tax period (or portion thereof) ending prior to the Effective Time. Buyer shall be allocated and bear, and indemnify and hold harmless Sellers for, all Asset Taxes that are attributable to any Tax period (or portion thereof) beginning at or after the Effective Time. Each Party shall be responsible for its own income, franchise, commercial activity, and similar Taxes.
- 5.6.2 Notwithstanding any other provision of this Agreement, to the extent the actual amount of an Asset Tax is not determinable at the Closing or at the time of the determination of the Final Statement, as applicable, (i) the Parties shall utilize the most recent information available in estimating the amount of such Asset Tax for purposes of such adjustment, and (ii) upon the later determination of the actual amount of such Asset Tax, timely payments will be made from one Party to the other to the extent necessary to cause each Party to bear the amount of such Asset Tax that is allocable to such Party under this <u>Section 5.6</u>.
- 5.6.3 For purposes of this Agreement (i) Asset Taxes that are attributable to the severance or production of Hydrocarbons shall be allocated to the Tax period or portion thereof in which the severance or production giving rise to such Asset Taxes occurred, (ii) Asset Taxes that are based upon or related to income or receipts or imposed on a transactional basis (other than such Asset Taxes described in clause (i)), shall be allocated to the Tax period or portion thereof in which the transaction giving rise to such Asset Taxes occurred, and (iii) Asset Taxes that are ad valorem, property or other Asset Taxes imposed on a periodic basis pertaining to a Straddle Period shall be allocated between the portion of such Straddle Period ending immediately prior to the date on which the Effective Time occurs and the portion of such Straddle Period beginning on the date on which the Effective Time occurs by prorating each such Asset Tax based on the number of days in the applicable Straddle Period that occur before the date on which the Effective Time occurs, on the one hand, and the number of

- days in such Straddle Period that occur on or after the date on which the Effective Time occurs, on the other hand. For purposes of clause (iii) of the preceding sentence, the period for such Asset Taxes shall begin on the date on which ownership of the applicable Property gives rise to liability for the particular Asset Tax and shall end on the day before the next such date.
- 5.6.4 Except as required by applicable Law, in respect of Asset Taxes, (i) Sellers shall be responsible for the preparation and timely filing of (A) all Tax Returns due prior to the Closing Date, and (B) all Tax Returns with respect to Taxable periods ending prior to the Effective Time (regardless of when due), and (ii) Buyer shall be responsible for the preparation and timely filing of all other Tax Returns and the payment to the applicable Governmental Authority of all Asset Taxes that become due and payable with respect to such Tax Returns. Buyer shall indemnify and hold Sellers harmless for any failure to file such Tax Returns and to make such payments. Buyer shall prepare all such Tax Returns on a basis consistent with past practice except to the extent otherwise required by applicable Law. Buyer shall provide Sellers with a copy of any Tax Return relating to any Straddle Period for Sellers' review at least ten (10) days prior to the due date for the filing of such Tax Return is required to be filed less than ten (10) days after the close of such Taxable period), and Buyer shall incorporate all reasonable comments of Sellers provided to Buyer in advance of the due date for the filing of such Tax Return.
- 5.6.5 If a Party receives a refund (whether by way of refund, credit, offset or otherwise) of Taxes for which the other Party is responsible hereunder, the first Party shall promptly pay such amount to the other Party.
- 5.6.6 The Parties shall cooperate fully, as and to the extent reasonably requested by the other Party, in connection with the filing of Tax Returns and any audit, litigation, or other proceeding with respect to Taxes relating to the transactions contemplated hereby. Such cooperation shall include the retention and (upon another Party's request) the provision of records and information that are relevant to any such Tax Return or audit, litigation or other proceeding and making employees available on a mutually convenient basis to provide additional information and explanation of any material provided under this Agreement. The Parties agree to retain all books and records with respect to Tax matters pertinent to the Properties relating to any Tax period beginning before the Closing Date until the expiration of the statute of limitations of the respective Tax periods and to abide by all record retention agreements entered into with any Governmental Authority.
- 5.7 <u>Suspended Funds</u>. As part of the final accounting in connection with the Final Statement pursuant to <u>Section 2.9</u>, each Seller will deliver to the Buyer, in the form of an adjustment to the Purchase Price as set forth in <u>Section 2.7</u>,

any Suspended Funds in its possession and the Buyer shall administer all such Suspended Funds and assume all payment obligations relating to the Suspended Funds in accordance with all applicable Laws, and shall be liable for the payment thereof to the proper parties. The Suspended Funds, as of the dates set forth therein, are set forth on Schedule 3.20, along with the reasons that such funds are being held in suspense.

- Operatorship. The Sellers make no representation and do not warrant or guarantee that the Buyer will succeed in being appointed successor operator. Buyer acknowledges that Sellers make no representations or warranties to Buyer as to transferability or assignability of operatorship of any Properties. Rights and obligations associated with operatorship of such Properties are governed by operating and similar agreements covering the Properties and determined in accordance with the terms of such agreements. Upon reasonable request from Buyer, at Buyer's sole cost and expense, for a period of one-hundred twenty (120) days after the Closing Date, Sellers will use commercially reasonable efforts to assist Buyer in Buyer's efforts to succeed a Seller as operator of any Wells operated by such Seller. The Buyer shall promptly, following Closing, file and diligently pursue until receipt of any acknowledgement, consent or confirmation by Governmental Authorities all appropriate forms, permit transfers and declarations or bonds with federal and state agencies relative to the Buyer's assumption of operatorship. For all Properties operated by any of the Sellers, the Buyer shall execute and deliver to the Sellers, on forms proscribed by the applicable Governmental Authorities which are to be prepared by the Buyer and acceptable to the Sellers, and the Sellers shall promptly file, such applicable forms transferring operatorship of such Properties to the Buyer.
- 5.9 <u>Limitations on Representations and Warranties</u>.
  - 5.9.1 EXCEPT FOR THE EXPRESS AND SPECIFIC REPRESENTATIONS AND WARRANTIES OF SUCH SELLER IN <u>ARTICLE 3</u> OF THIS AGREEMENT, THE REPRESENTATIONS AND WARRANTIES CONFIRMED IN THE CERTIFICATE DELIVERED AT CLOSING BY THE SELLERS PURSUANT TO <u>SECTION 8.2.4</u> AND THE SPECIAL WARRANTY, AND SUBJECT AND WITHOUT PREJUDICE TO THE BUYER'S EXPRESS RIGHTS AND REMEDIES HEREIN, (a) THE BUYER ACKNOWLEDGES THAT THE SELLERS HAVE NOT MADE, AND EACH SELLER HEREBY EXPRESSLY DISCLAIMS AND NEGATES ANY OTHER REPRESENTATION OR WARRANTY (EXPRESS, IMPLIED, AT COMMON LAW, BY

STATUTE OR OTHERWISE), AND (b) THE BUYER HEREBY ACKNOWLEDGES THAT IT HAS NOT RELIED UPON AND EXPRESSLY WAIVES, ANY SUCH OTHER REPRESENTATION OR WARRANTY (EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE), OR ANY STATEMENT OR INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO THE BUYER OR ANY OF ITS AFFILIATES, OR ITS OR THEIR EMPLOYEES, AGENTS, OFFICERS, DIRECTORS, MEMBERS, MANAGERS, EQUITY OWNERS, CONSULTANTS, REPRESENTATIVES OR ADVISORS (INCLUDING ANY OPINION, INFORMATION, PROJECTION OR ADVICE THAT MAY HAVE BEEN PROVIDED TO THE BUYER BY ANY EMPLOYEE, AGENT, OFFICER, DIRECTOR, MEMBER, MANAGER, EQUITY OWNER, CONSULTANT, REPRESENTATIVE OR ADVISOR OF SELLER OR ANY OF ITS AFFILIATES).

5.9.2 FURTHER, WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, EXCEPT FOR THE EXPRESS REPRESENTATIONS AND WARRANTIES OF THE SELLERS IN ARTICLE 3, THE REPRESENTATIONS AND WARRANTIES CONFIRMED IN THE CERTIFICATE DELIVERED AT CLOSING BY THE SELLERS PURSUANT TO SECTION 8.2.4 AND THE SPECIAL WARRANTY, AND SUBJECT AND WITHOUT PREJUDICE TO THE BUYER'S EXPRESS RIGHTS AND REMEDIES SET FORTH HEREIN, EACH SELLER EXPRESSLY DISCLAIMS ANY REPRESENTATION OR WARRANTY (EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE) AS TO (A) TITLE OF THE PROPERTIES, (B) PRODUCTION RATES, RECOMPLETION OPPORTUNITIES, DECLINE RATES, GAS BALANCING INFORMATION, OR THE QUALITY, QUANTITY OR VOLUME OF THE RESERVES OF HYDROCARBONS, IF ANY, ATTRIBUTABLE TO THE PROPERTIES OR SUCH SELLER'S INTEREST THEREIN, (C) THE CONTENTS, CHARACTER, NATURE ACCURACY, COMPLETENESS OR MATERIALITY OF ANY RECORDS, INFORMATION, DATA OR OTHER MATERIALS (WRITTEN OR ORAL) NOW, HERETOFORE OR HEREAFTER FURNISHED TO THE BUYER BY OR ON BEHALF OF SUCH SELLER, INCLUDING (I) ANY DESCRIPTIVE MEMORANDUM, OR ANY REPORT OF ANY PETROLEUM ENGINEERING CONSULTANT, OR ANY GEOLOGICAL OR SEISMIC DATA OR INTERPRETATION, RELATING TO THE PROPERTIES, (II) ANY DESCRIPTIVE MEMORANDUM, REPORTS, BROCHURES, CHARTS OR STATEMENTS PREPARED BY THIRD PARTIES, AND (III) ANY OTHER MATERIALS OR INFORMATION THAT MAY HAVE BEEN MADE AVAILABLE OR COMMUNICATED TO THE BUYER OR ITS AFFILIATES, OR ITS OR THEIR EMPLOYEES, AGENTS, OFFICERS, DIRECTORS, MEMBERS, MANAGERS, EQUITY

OWNERS, CONSULTANTS, REPRESENTATIVES OR ADVISORS IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT OR ANY DISCUSSION OR PRESENTATION RELATING THERETO, (D) THE ENVIRONMENTAL OR OTHER CONDITION OF THE PROPERTIES, AND (E) ANY ESTIMATES OF THE VALUE OF THE PROPERTIES OR FUTURE REVENUES GENERATED BY THE PROPERTIES.

5.9.3 EXCEPT FOR THE EXPRESS REPRESENTATIONS AND WARRANTIES OF SUCH SELLER IN ARTICLE 3 OF THIS AGREEMENT, THE REPRESENTATIONS AND WARRANTIES CONFIRMED IN THE CERTIFICATE DELIVERED AT CLOSING BY THE SELLERS PURSUANT TO SECTION 8.2.4 AND THE SPECIAL WARRANTY AND SUBJECT AND WITHOUT PREJUDICE TO THE BUYER'S EXPRESS RIGHTS AND REMEDIES SET FORTH HEREIN, AND WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, EACH SELLER EXPRESSLY DISCLAIMS AND NEGATES, AND THE BUYER HEREBY WAIVES, AS TO PERSONAL PROPERTY, EQUIPMENT, INVENTORY, MACHINERY AND FIXTURES CONSTITUTING A PART OF THE PROPERTIES (A) ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, (B) ANY IMPLIED OR EXPRESS WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE, (C) ANY IMPLIED OR EXPRESS WARRANTY OF CONFORMITY TO MODELS OR SAMPLES OF MATERIALS, (D) ANY RIGHTS OF THE BUYER UNDER APPROPRIATE STATUTES TO CLAIM DIMINUTION OF CONSIDERATION OR RETURN OF THE PURCHASE PRICE, (E) ANY IMPLIED OR EXPRESS WARRANTY OF FREEDOM FROM DEFECTS, WHETHER KNOWN OR UNKNOWN, (F) ANY AND ALL IMPLIED WARRANTIES EXISTING UNDER APPLICABLE LAW, AND (G) ANY IMPLIED OR EXPRESS WARRANTY REGARDING ENVIRONMENTAL LAWS, THE RELEASE OF SUBSTANCES, WASTES OR MATERIALS INTO THE ENVIRONMENT, OR PROTECTION OF THE ENVIRONMENT OR HEALTH, IT BEING THE EXPRESS INTENTION OF THE BUYER AND EACH SELLER THAT, EXCEPT FOR THE EXPRESS REPRESENTATIONS AND WARRANTIES OF THE SELLERS IN ARTICLE 3 AND CONFIRMED IN THE CERTIFICATE DELIVERED AT CLOSING BY THE SELLERS PURSUANT TO SECTION 8.2.4 AND SUBJECT AND WITHOUT PREJUDICE TO THE BUYER'S EXPRESS RIGHTS AND REMEDIES SET FORTH HEREIN, THE PERSONAL PROPERTY, EQUIPMENT, INVENTORY, MACHINERY AND FIXTURES IN WHICH SUCH SELLER HAS ANY INTEREST ARE BEING ACCEPTED BY THE BUYER, "AS IS, WHERE IS, WITH ALL FAULTS" AND IN THEIR PRESENT CONDITION AND STATE OF

REPAIR, AND THE BUYER REPRESENTS TO EACH SELLER THAT THE BUYER WILL MAKE OR CAUSE TO BE MADE SUCH INSPECTIONS WITH RESPECT TO SUCH PERSONAL PROPERTY, EQUIPMENT, INVENTORY, MACHINERY, FIXTURES AND OTHER PROPERTIES AS THE BUYER DEEMS APPROPRIATE.

- 5.9.4 THE SELLERS AND THE BUYER AGREE THAT, TO THE EXTENT REQUIRED BY LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN WARRANTIES CONTAINED IN THIS SECTION 5.9 ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSES OF ANY LAW, RULE OR ORDER.
- 5.10 NORM, WASTES AND OTHER SUBSTANCES. THE BUYER ACKNOWLEDGES AND AGREES THAT THE PROPERTIES HAVE BEEN USED FOR EXPLORATION, DEVELOPMENT, AND PRODUCTION OF OIL AND GAS AND THAT EQUIPMENT AND SITES INCLUDED IN THE PROPERTIES MAY CONTAIN ASBESTOS, NATURALLY OCCURRING RADIOACTIVE MATERIAL ("NORM") OR OTHER HAZARDOUS SUBSTANCES. NORM MAY AFFIX OR ATTACH ITSELF TO THE INSIDE OF WELLS, MATERIALS, AND EQUIPMENT AS SCALE, OR IN OTHER FORMS. THE WELLS, MATERIALS, AND EQUIPMENT LOCATED ON THE PROPERTIES OR INCLUDED IN THE PROPERTIES MAY CONTAIN NORM AND OTHER WASTES OR HAZARDOUS SUBSTANCES. NORM CONTAINING MATERIAL OR OTHER WASTES OR HAZARDOUS SUBSTANCES MAY HAVE COME IN CONTACT WITH VARIOUS ENVIRONMENTAL MEDIA, INCLUDING AIR, WATER, SOILS OR SEDIMENT. SPECIAL PROCEDURES MAY BE REQUIRED FOR THE ASSESSMENT, REMEDIATION, REMOVAL, TRANSPORTATION, OR DISPOSAL OF ENVIRONMENTAL MEDIA, WASTES, ASBESTOS, NORM AND OTHER HAZARDOUS SUBSTANCES FROM THE PROPERTIES.

### 5.11 Governmental Approvals.

5.11.1 Each of Buyer and the Sellers shall submit as soon as practicable, but in no event later than ten (10) Business Days after the execution hereof, filings under the HSR Act. The Persons making such filings shall request early termination of any applicable HSR Act waiting period, shall promptly furnish each other with copies of any substantive or material procedural notices, correspondence or other written communication received from the relevant Governmental Authority, shall promptly make any necessary or advisable subsequent or supplemental filings or submissions required or requested by the relevant Governmental Authority and shall, to the extent reasonably necessary and appropriate, cooperate with one another in the preparation of such filings or submissions. No Party shall participate in any meeting or substantive discussion with any Governmental Authority in respect of any such filings or related investigations or other inquiries unless, to the extent

practicable, it consults with the other Parties in advance and, to the extent practicable and permitted by such Governmental Authority, gives the other Parties the opportunity to attend and participate in such meeting. Each of Buyer, on the one hand, and Sellers, on the other hand, shall be responsible for one-half of all filing fees in connection with filings under the HSR Act.

- 5.11.2 Each Party shall use its commercially reasonable efforts to resolve as promptly as practicable such objections, if any, as may be asserted by any Governmental Authority with respect to the transactions contemplated by this Agreement under the HSR Act. If any Governmental Authority shall seek an injunction or the enactment, entry, enforcement or promulgation of any governmental order restraining or prohibiting the transactions contemplated by this Agreement, each Party shall (i) defend claims, demands or other actions seeking such an injunction or the enactment, entry, enforcement or promulgation of any such governmental order and (ii) use its commercially reasonable efforts to avoid the entry of and seek to have lifted, vacated or reversed any such governmental order. Notwithstanding the foregoing, Buyer (including its Affiliates) shall not be obligated to propose, negotiate, commit to, or effect by consent decree, hold-separate order, or otherwise, any sale, divestiture, licensing, or disposition of any assets or businesses (including, following the Closing, the Properties), or any alteration of business conduct that would limit freedom of action with respect to, or its ability to retain, any businesses, product lines, or assets (including, following the Closing, the Properties).
- 5.11.3 Buyer and Sellers shall not, and shall cause their respective Affiliates not to, take any action that could reasonably be expected to adversely affect the approval of any Governmental Authority of any of the filings referred to in this <u>Section 5.11</u>.

### 5.12 <u>CFIUS</u>.

5.12.1 The Parties will jointly submit a draft notice to the Committee on Foreign Investment in the United States and its member agencies (collectively, "CFIUS") with respect to the transactions contemplated by this Agreement (as promptly as practicable, using commercially reasonable efforts to have such submission made within five (5) Business Days from the Execution Date) and proceed to finalize such notice as soon as practicable thereafter reflecting input on such draft filing received from CFIUS, provided that any such notices shall not be filed without the Sellers' prior approval thereof. The Parties shall respond fully, appropriately and timely to any request for information from CFIUS throughout the CFIUS process and in accordance with the CFIUS regulations. In addition, as consistent with any CFIUS requests for confidentiality, each Party shall cooperate and each Party shall have the opportunity to attend (or have its representatives participate) in any

meetings with CFIUS member agencies, attend any on-site visit by CFIUS member agencies to a Party's facility (if requested by a CFIUS member agency), and take any other commercially reasonable action in furtherance of the CFIUS Approval. The Parties shall, and shall cause each of their respective Affiliates to, take any and all commercially reasonable actions necessary, proper or advisable, to obtain CFIUS Approval as soon as is practicable and feasible, in accordance with the CFIUS timetable; provided, however, that such efforts with respect to the Buyer shall not require the Buyer to agree to any mitigation measures proposed by CFIUS or any of its member agencies that would have a material adverse economic effect on the business of the Buyer or its Affiliates or a material adverse impact on the governance of the Buyer's or its Affiliates' business or on the Properties (the "Mitigation Standard"); and provided further, that if CFIUS or any of its member agencies requests a mitigation measure that would obligate the Buyer or its Affiliates to sell, dispose of, divest, transfer, or hold separate or otherwise limit its ability to retain any of the Properties, the Buyer shall exclude any or all of such affected Properties from the transactions contemplated hereby, and any such Properties that are so excluded shall be deemed hereunder to be Excluded Assets and the Purchase Price shall be adjusted downward by the aggregate of the Allocated Values thereof.

5.12.2 Each Party shall keep the other Party reasonably apprised of the content and status of any communications with, and communications from, CFIUS and shall permit the other Party to review in advance (and to consider any comments made by the other Party in relation to) any proposed substantive communication by such Party to CFIUS (except to the extent such communication contains confidential or proprietary information not directly related to the transactions contemplated by this Agreement).

### 5.13 <u>Midstream Agreements</u>.

5.13.1 Prior to Closing, the Sellers and the Buyer will negotiate in good faith (with each other, and with the relevant Contract counterparties) mutually acceptable forms of assignment, assumption and, as applicable, consent and release agreements to effect, in accordance with the terms of this Agreement, the assignment to the Buyer at Closing of the Contracts set forth on <a href="Exhibit E">Exhibit E</a> (subject to the other provisions of this <a href="Section 5.13">Section 5.13</a>) (collectively, the "**Midstream Agreements**" and such assignment, assumption and consent agreements for the Midstream Agreements, collectively, the "**Midstream Assignments**"). At Closing, the Sellers will assign their rights and obligations (subject to the other provisions of this <a href="Section 5.13">Section 5.13</a>) in and to the Midstream Agreements, and the Buyer will assume (and, upon delivery by Sellers of the Midstream Assignments, the Buyer will be deemed to have assumed), and thereafter will be bound by and comply with and perform, all of the obligations of the Sellers contained in the Midstream Agreements.

- 5.13.2 The Buyer shall meet any creditworthiness and adequate assurance of performance requirements contained in any of the Midstream Agreements listed on Exhibit D and shall take all steps reasonably necessary to enable it to take an assignment of such Midstream Agreements at the Closing and to cause the counterparty of any such Midstream Agreements to grant its consent (including with respect to any Required Consents) to the assignment thereof to the Buyer. In the event that any of the Midstream Agreements on Exhibit D (other than the FERC Contracts which are the subject of Section 5.13.3) cannot be assigned to Buyer at Closing, the Sellers shall have no liability to Buyer with respect to such unassigned Midstream Agreements (except as set forth in this Section 5.13), except that the Parties will use commercially reasonable efforts to negotiate a mutually acceptable arrangement to address such unassigned Midstream Agreements, failing of which, at Sellers' option, the Parties will enter into the following arrangement with respect to such unassigned Midstream Agreements (the "Midstream Pass-Through Arrangement"): (a) for the first six (6) months following Closing Date, a Hydrocarbons Purchase Agreement and (b) in the event such Midstream Agreement has not been assigned to Buyer on or before the day that is six (6) months after the Closing Date, a long term purchase agreement with respect to the Hydrocarbon volumes covered by such Midstream Agreement, which agreement shall include such terms and conditions that are customary for a long term Hydrocarbons purchase agreement, including (regardless of whether customary or not): (w) a long term acreage dedication provision, (x) an obligation to pay marketing fees with respect to such Hydrocarbons not less than the Management Fees with respect to the Marketing Services under the TSA (as such terms are defined in the TSA), (y) creditworthiness and adequate assurance of performance requirements equal to those Sellers have in place with respect to such Midstream Agreements and (z) a parent guaranty from Buyer's parent guaranteeing their obligations thereunder; provided, that, with respect to any Midstream Agreement for which Sellers and Buyer have entered into the Midstream Pass-Through Arrangement, if the Buyer is able to obtain a replacement or assignment to it thereof, such Midstream Agreement shall no longer be subject to the Midstream Pass-Through Arrangement.
- 5.13.3 In the event any of the Midstream Agreements listed on Exhibit D are not assigned at Closing, (i) the Buyer shall nonetheless continue to meet any creditworthiness and adequate assurance of performance requirements contained in any of the Midstream Agreements listed on Exhibit D ((A) subject to the other provisions of this Section 5.13 and (B) provided that Buyer will not be required to provide or otherwise double-post any security that it has previously provided pursuant to the Midstream Pass-Through Arrangement), (ii) (A) Buyer shall continue to take all steps reasonably necessary and (B) Sellers shall continue to use commercially reasonable efforts, in each case, to enable Buyer to take an assignment of such Midstream Agreements as soon as possible after Closing and to cause

the counterparty of any such Midstream Agreement to grant its consent to the assignment thereof to the Buyer and (iii) Sellers will not amend, change or waive any material terms thereof in a manner that has an adverse effect on the Buyer, without first obtaining the Buyer's prior written consent (such consent not to be unreasonably withheld, conditioned, or delayed).

- 5.13.4With respect to any unassigned Midstream Agreement for gas transportation services on pipelines and facilities regulated by the FERC ("FERC Contracts") for which the contract counterparty has consented to the assignment but FERC has not, the Sellers shall hold such FERC Contracts for the benefit of the Buyer until they are assigned to the Buyer (or they otherwise terminate) on the terms and conditions of the Midstream Pass-Through Arrangement.
- 5.13.5 The Parties shall cooperate regarding the preparation and filing as soon as reasonably practicable following the Execution Date of a joint petition with the FERC for a temporary waiver of capacity release regulations and related policies for the assignment and release of those certain firm capacity transportation Contracts among the Parties. The Sellers agree, at the Buyer's cost and expense, to cooperate with and assist Buyer as reasonably necessary with that petition and any related follow-up in seeking all required relief from the FERC with respect to the transfer of such firm capacity transportation Contracts.
- 5.13.6 Notwithstanding anything to the contrary, to the extent Buyer's obligations under this <u>Section 5.13</u> (including those arising under any Midstream Pass-Through Arrangement at Closing) would require the provision of financial assurances or credit enhancement, including letters of credit, parent guaranties, deposits, or similar items, such obligations shall not require the Buyer to post, or cause the posting, of (i) any such instruments at Closing or (ii) any such instruments post-Closing pursuant to <u>Section 5.13.2(b)(y)</u>, collectively to the extent the aggregate exposure to Buyer thereunder would be in excess of **One Hundred and Sixty-Five Million Dollars (\$165,000,000)**; provided, however, that this <u>Section 5.13.6</u> shall not apply with respect to the provision of any additional financial assurances or credit enhancements required under any Midstream Agreement from and after the assignment of such Midstream Agreement to the Buyer.

### 5.14 <u>Hedges and Novation Agreements</u>.

5.14.1 On or before one (1) Business Day following the Execution Date, the Sellers shall enter into those swaps, collars or other costless hedges for those volumes of natural gas, those periods and those prices as more particularly set forth on <u>Schedule 5.14.1</u> ("**Natural Gas Hedges**").

- 5.14.2 The Sellers shall pay to any counterparty any reasonable out-of-pocket costs or fees to authorize the initiation of the Natural Gas Hedges; <u>provided</u>, if Closing occurs, the Buyer shall reimburse the Sellers for any such out-of-pocket costs or fees (such costs and fees excluding, for the avoidance of doubt, any price discounts reflected or otherwise incorporated in the prices of the Natural Gas Hedges).
- 5.14.3 Notwithstanding anything to the contrary in this Agreement, the Natural Gas Hedges and the underlying contracts and the actions to be taken by Sellers in accordance with this <u>Section 5.14</u> are an exception to, and will under no circumstance constitute a breach of any of Sellers' covenants in <u>Section 5.2</u> or its representations and warranties in <u>Section 3.14</u>.
- 5.14.4 Buyer shall pay, be responsible for, and defend, indemnify, and hold harmless the Seller Indemnified Parties from and against any and all losses, damages, claims, liabilities, debts, obligations, or expenses (including interest, reasonable legal fees, and expenses of litigation and attorneys' fees in enforcing this indemnity) incurred, suffered, paid by or resulting to any of the Seller Indemnified Parties that results from Seller's entry into any Natural Gas Hedges in accordance with this <a href="Section 5.14">Section 5.14</a>, if Closing occurs, (each, as applicable, the "Assumed Hedges") including: (a) payments made by Sellers or amounts payable by Sellers to any counterparty related to entering into the Assumed Hedges, or any amendments, waivers, transfers or terminations of the Assumed Hedges; (b) payments made by Sellers or amounts payable by Sellers related to transferring to, or novating in favor of, Buyer the Assumed Hedges; (c) payments made by Sellers or amounts payable by Sellers related to any monthly settlement of an Assumed Hedge; and (d) payments made by Sellers or amounts payable by Sellers under any Assumed Hedge pursuant to an Unwinding Scenario (as defined below) (the "Buyer Hedging Indemnities"). The Buyer shall make any required payments pursuant to the Buyer Hedging Indemnities to the Sellers within ten (10) Business Days after receipt of an invoice with respect thereto.
- 5.14.5 If Buyer fails to assume the Natural Gas Hedges or the transactions contemplated thereby at Closing for any reason whatsoever (other than by reason of any actions of any counterparty to the Natural Gas Hedges) (in either case, an "**Unwinding Scenario**"), then Sellers may, at the sole cost and expense of Buyer, unwind all of the Natural Gas Hedges and the transactions contemplated thereby in accordance with the terms of the confirmations contained in each of the Natural Gas Hedges.
- 5.14.6 If Closing does not occur, the Sellers shall pay, be responsible for, and defend, indemnify, and hold harmless the Buyer Indemnified Parties from and against any and all losses, damages, claims, liabilities, debts, obligations, or expenses (including interest, reasonable legal fees, and expenses of litigation and attorneys' fees in enforcing this indemnity)

incurred, suffered, paid by or resulting to any of the Buyer Indemnified Parties that results from Seller's entry into any Natural Gas Hedge in accordance with this <u>Section 5.14</u> (the "**Sellers Hedging Indemnities**"). The Sellers shall make any required payments pursuant to the Sellers Hedging Indemnities to the Buyer within ten (10) Business Days after receipt of an invoice with respect thereto.

- 5.14.7 For the avoidance of doubt, (i) if Closing occurs, Buyer shall be entitled to and responsible for, as applicable, all revenues, gains, proceeds, losses, costs and expenses related to the Natural Gas Hedges and the transactions contemplated thereby, including all revenues, gains, proceeds, losses, costs and expenses in an Unwinding Scenario, it being understood that all such revenues, gains and proceeds shall be the property of the Buyer and (ii) if Closing does not occur, Sellers shall be entitled to and responsible for, as applicable, all revenues, gains, proceeds, losses, costs and expenses related to the Natural Gas Hedges and the transactions contemplated thereby, it being understood that all such revenues, gains and proceeds shall be the property of the Sellers.
- 5.14.8 The Sellers and the Buyer shall take, on or prior to the Closing, all steps necessary to effectuate the novation of the Natural Gas Hedges to Buyer at Closing pursuant to the Novation Agreements (including Buyer entering into an ISDA Master Agreement with each Natural Gas Hedge counterparty).
- 5.15 <u>Total Development Agreement</u>. The Parties shall exercise commercially reasonable efforts to satisfy any requirements set forth in the Total Development Agreement with respect to the assignment of the Total Pref Properties by Sellers to Buyer, including, in the case of Buyer, that Buyer be a "Technically Qualified Assignee" and a "Credit-Worthy Assignee" (as both terms are defined in the Total Development Agreement), and each Party shall take all steps reasonably necessary to satisfy Total as of the Closing that such requirements have been met and to cause Total to confirm in writing that such requirements have been met.
- 5.16 <u>Completion Contracts</u>. From and after the Execution Date and for a period of 90 days from and after Closing, the Sellers shall use commercially reasonable efforts to assist the Buyer in Buyer's procurement of post-Closing completion crew services with the Completion Contract Counterparties similar to the completion crew services used by Sellers (including utilization of the same crews and individuals performing such services) in the Ordinary Course of Business in connection with Sellers' ownership and operation of the Properties.
- 5.17 <u>Financing Cooperation</u>.
  - 5.17.1 Prior to the Closing Date, Sellers shall use commercially reasonable efforts to:

- A. deliver to Buyer the Required Financial Information on or before the day that is thirty-five (35) days from the Execution Date (for the avoidance of doubt, the delivery of such Required Financial Information on or before the day that is thirty-five (35) days from the Execution Date shall not be subject to any commercial reasonableness qualifier applicable to this Section 5.17.1);
- B. cooperate, or cause its accountants to cooperate, with the Buyer in its preparation of the pro forma financial statements required pursuant to the Debt Commitment Letter and any customary comfort letters related thereto (<u>provided</u>, that, for the avoidance of doubt, the Sellers shall not be required to deliver or cause the delivery of any such comfort letter);
- C. request any consents of accountants for use of their reports in any materials relating to the Debt Financing; and
- D. provide monthly production and accounting lease operating statements in respect of the Properties promptly after they are prepared, and in any event, within thirty (30) days after the end of each applicable month ended at least thirty (30) days prior to the Closing Date (for the avoidance of doubt, the delivery of such monthly production and accounting lease operating statements shall not be subject to any commercially reasonableness qualifier applicable to this <u>Section 5.17.1</u>);

it being understood and agreed that all materials and information obtained by each Buyer Party pursuant hereto may be shared with the Debt Financing Sources; <u>provided</u> that this <u>Section 5.17.1</u> shall not require travel or the obligation to incur any out-of-pocket Third Party costs by any of the subject Persons in order to comply with the terms hereof and that the Buyer will make reasonable efforts to minimize any disruption associated with the cooperation contemplated by such Persons hereby. In each case, Sellers' cooperation pursuant to this <u>Section 5.17.1</u> shall be at the Buyer's written request with reasonable prior notice to the Sellers, and no such cooperation by the Sellers shall be required to the extent it could cause any representation or warranty in this Agreement to be breached, cause any condition to the Closing set forth in <u>Article VI</u> to fail to be satisfied or otherwise cause any breach of this Agreement.

Notwithstanding anything to the contrary herein, it is understood and agreed that the condition precedent set forth Article 6, as applied to each Seller's obligations under this Section 5.17.1, (other than with respect to each Seller's obligations to deliver the Required Financial Information pursuant to Section 5.17.1(A) above or monthly production and lease operating statements pursuant to Section 5.17(D) above), shall be deemed to be satisfied unless the Financing contemplated by the Debt Commitment Letters has not been obtained as a direct result of any Seller's willful breach of its obligations under this Section.

5.17.2 The Buyer shall promptly reimburse the Sellers and their Affiliates for all reasonable and documented internal and external costs and expenses, including any auditor and attorneys' fees and expenses, incurred by the Sellers or their Affiliates to perform their respective obligations pursuant to the foregoing.

### 5.18 <u>Drilling Rig Contracts</u>.

- 5.18.1 From and after the Execution Date and for a period of ninety (90) days from and after Closing, Sellers shall (a) use commercially reasonable efforts to cooperate with and assist Buyer in negotiating with the Drilling Rig Contract Counterparty to facilitate Buyer's entering into a drilling rig contract with the Drilling Rig Contract Counterparty on substantially the same terms as Drilling Rig Contract 1 (effective at Closing or as soon as practicable thereafter and, for the avoidance of doubt, covering "Rig 571") or (b) if the Drilling Rig Contract Counterparty consents to release Sellers from any liabilities under Drilling Rig Contract 1 arising following assignment, assign Drilling Rig Contract 1 (or any replacement thereof) to Buyer pursuant to one or more assignments in substantially the form of the Assignment (effective at Closing or as soon as practicable thereafter).
- 5.18.2 In the event that Closing occurs, Sellers shall (i) use commercially reasonable efforts to cooperate with and assist Buyer in negotiating with the Drilling Rig Contract Counterparty to facilitate Buyer's entering into a drilling rig contract with the Drilling Rig Contract Counterparty on substantially the same terms as Drilling Rig Contract 2 on or before December 31, 2018 (for the avoidance of doubt, covering "Rig 580") or (ii) if the Drilling Rig Contract Counterparty consents to release Sellers from any liabilities under Drilling Rig Contract 2 arising following assignment, assign Drilling Rig Contract 2 (or any replacement thereof) to Buyer pursuant to one or more assignments in substantially the form of the Assignment on or before December 31, 2018.
- 5.18.3 In the event of assignment to Buyer, each Drilling Rig Contract shall become a part of the Properties.
- 5.18.4 Notwithstanding anything else herein, Buyer shall be solely responsible for any mobilization costs arising out of the assignment or transfer to Buyer of (or replacement of the applicable rig under) Drilling Rig Contract 1 or Drilling Rig Contract 2.

#### 5.19 FCC Licenses.

5.19.1 Within a reasonable period of time after the Execution Date, Buyer and Sellers shall file or cause to be filed with the FCC all appropriate applications required to be filed by Closing with respect to the assignment to Buyer of the FCC licenses to be conveyed to Buyer pursuant to the transactions contemplated herein (the "FCC Assignment Applications"). The FCC Assignment Applications and any supplemental information furnished in connection therewith shall (i) be in substantial compliance with Title 47 of the Code of Federal Regulations, as amended from time to time, and any policies or published decisions issued pursuant to such

- regulations or the Communications Act, or (ii) be responsive to a request of the FCC.
- 5.19.2 Buyer and Sellers shall furnish to each other such necessary information and reasonable assistance as the other may reasonably request in connection with the preparation, filing and prosecution of the FCC Assignment Applications. Buyer shall bear all expenses in connection with the preparation, filing and prosecution of the FCC Assignment Applications. Buyer and Sellers shall each use their commercially reasonable efforts to prosecute the FCC Assignment Applications and shall furnish to the FCC any documents, materials, or other information reasonably requested by the FCC.

## 5.20 <u>Production True-Up Payments</u>.

- 5.20.1 For a period beginning on the Closing Date until December 31, 2018 (the "**True-Up Period**"), upon the last day of each calendar month during the True-Up Period (each such month-end, a "**Determination Date**"):
- A. If the Average Henry Hub Determination Price for such month is greater than \$3.00, then the Buyer shall owe to the Sellers, with respect to such month (subject to <u>Section 5.20.2</u>) an amount in Dollars equal to the product of: (i) the True-Up Quantity, multiplied by (ii) the difference expressed in Dollars between (x) the Average Henry Hub Determination Price on such Determination Date, minus (y) \$3.00 (the "**Buyer True-Up Payments**").
- B. If the Average Henry Hub Determination Price for such month is less than \$3.00, then the Sellers shall owe to the Buyer with respect to such month (subject to Section 5.20.2) an amount in Dollars equal to the product of: (i) the True-Up Quantity, multiplied by (ii) the difference expressed in Dollars between (x) \$3.00, minus (y) the Average Henry Hub Determination Price on such Determination Date (the "Sellers True-Up Payments" and, together with the Buyer True-Up Payments, the "True-Up Payments").
  - 5.20.2 Any True-Up Payments owed by the Parties with respect to this <u>Section 5.20</u> will be credited or debited against amounts owed amongst the Parties under the TSA for the month in which the Determination Date occurs, in accordance invoicing procedures in Section 8 of the TSA. In the event that the TSA is terminated (and the "Accounting Settlement Period" thereunder has completed) without all True-Up Payments having been accounted for, any remaining unaccounted for Buyer True-Up Payments or Seller True-Up Payments, as applicable, will be offset against the other, and the owing Party will pay and deliver such net amounts to the Party owed within ten (10) Business Days of the expiration of the True-Up Period.
  - 5.20.3 "**Average Henry Hub Determination Price**" means, as determined with respect to a calendar month in which a Determination Date occurs, a number, expressed in Dollars to two decimal places, that is equal to such

monthly "Dollars per Million Btu" figure reported by the U.S. Energy Information Administration on the natural gas "Spot Prices" chart (which chart may be accessed at the following link: https://www.eia.gov/dnav/ng/hist/rngwhhdM.htm or any successor source of the same information) for such calendar month in which a Determination Date occurs; provided, that, for the avoidance of doubt, and as an example calculation of the "Average Henry Hub Determination Price" for the month of June, 2018, such "Average Henry Hub Determination Price" for such month equals \$2.97.

- 5.20.4 "**True-Up Quantity**" means a quantity of gas (in mmbtus per month) to be agreed by the Parties prior to Closing for each month of the True-Up Period based on the historic production from the Properties producing as of the Closing Date in the Ordinary Course of Business; <u>provided</u>, <u>however</u>, that (a) if the Parties fail to reach such agreement prior to Closing, the "True-Up Quantity" shall be deemed to be the quantity of gas set forth on Schedule 5.20 and (b) such quantity shall be prorated for the month in which the Closing Date occurs and shall be multiplied by a fraction, the numerator of which is the number of days in the period between the Closing Date and the Determination Date of the month in which the Closing Date occurs and the denominator of which is thirty (30).
- 5.21 <u>Environmental Records</u>. From the Execution Date until the second (2<sup>nd</sup>) anniversary of the Closing Date, Sellers shall (a) retain all Environmental Records and provide Buyer reasonable access thereto, and (b) provide periodic updates (and, in any event, prompt updates with respect to any notices or demands involving matter(s) of a material nature) to Buyer with respect thereto.

## ARTICLE 6 BUYER'S CONDITIONS PRECEDENT

- 6. <u>Buyer's Conditions Precedent</u>. The obligation of the Buyer to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver (subject to applicable Laws) at or prior to the Closing Date of each of the following conditions:
  - 6.1.1 No preliminary or permanent injunction or other order will have been issued (and remain in force) by any Governmental Authority having appropriate jurisdiction preventing consummation of the transactions contemplated by this Agreement;
  - 5.1.2 No action will have been commenced by any Third Party against any Seller, the Buyer or any of their respective Affiliates, associates, officers or directors seeking to prevent or challenge the transactions contemplated by this Agreement or seeking material damages arising from the transactions contemplated by this Agreement;

- 6.1.3 The representations and warranties of the Sellers set forth in <u>Article 3</u> shall be true and correct as of the Closing Date as though made on and as of the Closing Date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except for such breaches, if any, as would not (individually or in the aggregate) have a Material Adverse Effect (<u>provided</u> that to the extent such representation or warranty is qualified by its terms by materiality or Material Adverse Effect, such qualification in its terms shall be inapplicable for purposes of this <u>Section 6.1.3</u> and the Material Adverse Effect qualification contained in this <u>Section 6.1.3</u> shall apply in lieu thereof);
- 6.1.4 The Sellers will have performed or satisfied in all material respects on or prior to the Closing Date, all obligations, covenants, agreements and conditions contained in this Agreement to be performed or complied with by the Sellers on or prior to the Closing Date;
- 6.1.5 The sum of all downward adjustments to the Purchase Price (without duplication) that would result from (a) all valid uncured Title Defects (net of upward adjustments for Title Benefits), (b) all unremediated Environmental Defects, (c) all Properties excluded pursuant to Section 5.1.1 (in each case of (a), (b) and (c), asserted in good faith by the Buyer pursuant to a timely delivered Defect Notice in accordance with Section 2.2), (d) all Properties excluded pursuant to Section 5.12.1, and (e) all Casualty Loss Amounts, when taken together will not exceed **twenty percent** (20%) of the unadjusted Purchase Price;
- 6.1.6 (a) HSR Approval shall have been obtained, and (b) CFIUS Approval shall have been obtained, and no requirements or conditions to mitigate any national security concerns shall have been imposed by CFIUS on Buyer that exceed the Mitigation Standard unless agreed to by Buyer in its sole and absolute discretion or as provided in the second proviso of Section 5.12.1;
- 6.1.7 (i) The Total Pref Right shall have been waived or the time for exercising the Total Pref Right shall have expired or (ii) the Total Pref Right shall have been exercised and Buyer has made a positive Non-Total Pref Properties Election and the closing under the Total PSA has occurred (or is set to occur on the same day as Closing hereunder); and
- 6.1.8 (i) The counterparties to the Midstream Agreements set forth on Exhibit D shall have agreed to the assignment of such Contracts to the Buyer or (ii) if some or all of the counterparties to the Midstream Agreements set forth on Exhibit D have not agreed to the assignment of such Contracts to the Buyer, the Sellers shall have agreed to hold any such Midstream Agreements for the benefit of the Buyer until they are assigned to the Buyer (or they otherwise terminate) on the terms and conditions of the

Midstream Pass-Through Arrangement pursuant to <u>Section 5.13.2</u>; <u>provided</u>, <u>however</u>, that, for purposes of this <u>Section 6.1.8</u> and <u>Section 7.1.8</u> only, either Party may, from time to time prior to Closing, propose to the other Party that a Contract be deemed removed from <u>Exhibit D</u>, and if such other Party consents (such consent not to be unreasonably withheld) to such removal, such Contract shall be deemed removed for such purposes.

# ARTICLE 7 SELLERS' CONDITIONS PRECEDENT

- 7. <u>Sellers' Conditions Precedent</u>. The obligation of the Sellers to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver (subject to applicable Laws) at or prior to the Closing Date of each of the following conditions:
  - 7.1.1 No preliminary or permanent injunction or other order will have been issued (and remain in force) by any Governmental Authority having appropriate jurisdiction preventing consummation of the transactions contemplated by this Agreement;
  - 7.1.2 No action will have been commenced by any Third Party against any Seller, the Buyer or any of their respective Affiliates, associates, officers or directors seeking to prevent or challenge the transactions contemplated by this Agreement or seeking material damages arising from the transactions contemplated by this Agreement;
  - 7.1.3 The representations and warranties of the Buyer set forth in <a href="Article 4">Article 4</a> shall be true and correct as of the Closing Date (other than representations and warranties that refer to a specified date, which need only be true and correct on and as of such specified date), except for such breaches, if any, as would not (individually or in the aggregate) have a material adverse effect on the Buyer's ability to perform its obligations under this Agreement or to consummate the transactions contemplated by this Agreement; <a href="provided">provided</a> that to the extent such representation or warranty is qualified by its terms by materiality or Material Adverse Effect, such qualification in its terms shall be inapplicable for purposes of this <a href="Section 7.1.3">Section 7.1.3</a> and the materiality qualification contained in this <a href="Section 7.1.3">Section 7.1.3</a> shall apply in lieu thereof:
  - 7.1.4 The Buyer will have performed or satisfied in all material respects on or prior to the Closing Date, all obligations, covenants, agreements and conditions contained in this Agreement to be performed or complied with by the Buyer on or prior to the Closing Date;
  - 7.1.5 The sum of all downward adjustments to the Purchase Price (without duplication) that would result from (a) all valid uncured Title Defects (net

of upward adjustments for Title Benefits), (b) all unremediated Environmental Defects, (c) all Properties excluded pursuant to <u>Section 5.1.1</u> (in each case of (a), (b) and (c), asserted in good faith by the Buyer pursuant to a timely delivered Defect Notice in accordance with <u>Section 2.2</u>), (d) all Properties excluded pursuant to <u>Section 5.12.1</u>, and (e) all Casualty Loss Amounts, when taken together will not exceed **twenty percent** (20%) of the unadjusted Purchase Price;

- 7.1.6 (a) HSR Approval shall have been obtained, and (b) CFIUS Approval shall have been obtained, and no requirements or conditions to mitigate any national security concerns (other than as provided in the second proviso of Section 5.12.1) shall have been imposed by CFIUS on the Sellers, unless agreed to by the Sellers in their sole and absolute discretion;
- 7.1.7 (i) The Total Pref Right shall have been waived or the time for exercising the Total Pref Right shall have expired or (ii) the Total Pref Right shall have been exercised and Buyer has made a positive Non-Total Pref Properties Election and the closing under the Total PSA has occurred (or is set to occur on the same day as Closing hereunder); and
- 7.1.8 Subject to the proviso in <u>Section 6.1.8</u>, (i) the counterparties to the Midstream Agreements set forth on <u>Exhibit D</u> shall have agreed to the assignment of such Contracts to the Buyer or (ii) if some or all of the counterparties to the Midstream Agreements set forth on <u>Exhibit D</u> have not agreed to the assignment of such Contracts to the Buyer, the Sellers shall have agreed to hold any such Midstream Agreements for the benefit of the Buyer until they are assigned to the Buyer (or they otherwise terminate) on the terms and conditions of the Midstream Pass-Through Arrangement pursuant to <u>Section 5.13.2</u>.

## ARTICLE 8 CLOSING

8. The Closing. Subject to the terms and conditions hereof, unless extended as provided herein, the Closing will take place at 10:00 a.m. local time in the offices of Shearman & Sterling, LLP, 1100 Louisiana Street, Suite 3300, Houston, Texas 77002 on October 9, 2018 (the "Target Closing Date") (provided, that, if the Closing has not occurred by October 27, 2018, and the Total Pref Right becomes exercisable again with respect to the transactions contemplated under this Agreement (such date being the 91<sup>st</sup> day following the Execution Date), the "Target Closing Date" shall be automatically changed to November, 16, 2018), or on such date as the Buyer and the Sellers may agree upon in writing; provided that if the conditions to Closing in Article 6 and Article 7 (other than those conditions which by their nature are to be satisfied at the Closing) have not yet been satisfied or waived by the Target Closing Date, then subject to the Parties' rights under Article 12, Closing shall occur within five (5) Business Days after such conditions have been satisfied or waived. The date on which the Closing actually occurs shall be the "Closing Date." The Parties may, by mutual written consent, change the Closing Date to any other date that they may agree upon.

- 8.1 <u>Buyer's Deliveries</u>. On the Closing Date, and subject to the simultaneous performance by the Sellers of their obligations under <u>Section 8.2</u>, the Buyer will deliver or cause to be delivered the following items:
  - 8.1.1 <u>Purchase Price</u>; <u>Escrow Payment</u>. The Purchase Price to the Sellers as directed in writing by the Sellers (as adjusted pursuant to <u>Section 2.8</u> and less the Deposit) and any payments required pursuant to <u>Section 2.1.3</u> to the Escrow Agent;
  - 8.1.2 <u>Assignments</u>. An original counterpart of the (a) Assignment for each county in which Real Property Interests and Wells are located, and covering the Properties (other than the Surface Assets, Mineral Interests and those Properties to be excluded in accordance with the terms hereof) in recordable form, (b) the Surface Deed for each county in which the Surface Assets are located, and covering the Surface Assets and (c) the Mineral Deed for each county in which the Mineral Interests are located, and covering the Mineral Interests;
  - 8.1.3 Operatorship. To the extent required under any Law or Governmental Authority for any federal or state lease, the Buyer shall execute and deliver forms prescribed by the applicable Governmental Authorities to transfer status of operatorship of those Properties which such Seller or its Affiliate operates from a Seller or its Affiliate to the Buyer or the Buyer's designee;
  - 8.1.4 <u>Closing Statement</u>. A duly executed Closing Statement;
  - 8.1.5 <u>Officer's Certificate</u>. A duly executed certificate from an officer of the Buyer certifying on behalf of the Buyer that the conditions set forth in <u>Section 7.1.3</u> and <u>Section 7.1.4</u> have been fulfilled by the Buyer;
  - 8.1.6 <u>Transition Services Agreement</u>. A duly executed counterpart of the TSA;
  - 8.1.7 <u>Novation Agreements</u>. Duly executed counterparts of the Novation Agreements;
  - 8.1.8 Core Access Agreement. A duly executed counterpart of the Core Access Agreement;
  - 8.1.9 <u>Midstream Assignments</u>. Duly executed counterparts of the Midstream Assignments;
  - 8.1.10 <u>Joint Written Instruction</u>. A duly executed joint written instruction to the Escrow Agent to release the Deposit; and
  - 8.1.11 <u>Additional Documents</u>. Such additional documents customary in similar transactions as might be reasonably requested by the Sellers and are reasonably required to consummate the transactions contemplated by this Agreement.

- 8.2 <u>Sellers' Deliveries</u>. On the Closing Date, and subject to the simultaneous performance by the Buyer of its obligations under <u>Section 8.1</u>, each Seller will deliver or cause to be delivered to the Buyer the following items (all documents will be duly executed and acknowledged where required):
  - 8.2.1 <u>Assignments</u>. An original counterpart of the (a) Assignment for each county in which Real Property Interests and Wells are located, and covering the Properties (other than the Surface Assets, Mineral Interests and those Properties to be excluded in accordance with the terms hereof) in recordable form, (b) the Surface Deed for each county in which the Surface Assets are located, and covering the Surface Assets and (c) the Mineral Deed for each county in which the Mineral Interests are located, and covering the Mineral Interests;
  - 8.2.2 <u>Certificate of Non-Foreign Status</u>. A certificate of non-foreign status, in substantially the form of the attached <u>Exhibit B-4</u>, executed by an authorized officer of each Seller;
  - 8.2.3 <u>Closing Statement</u>. A duly executed Closing Statement;
  - 8.2.4 <u>Officer's Certificate</u>. A duly executed certificate from an officer of each Seller certifying on behalf of such Seller that the conditions set forth in Section 6.1.3 and Section 6.1.4 have been fulfilled by such Seller;
  - 8.2.5 <u>State and Federal Lease Assignments</u>. All assignments, on appropriate forms, of state and of federal leases comprising portions of the Properties, duly executed by the Sellers;
  - 8.2.6 <u>Transition Services Agreement</u>. A duly executed counterpart of the TSA;
  - 8.2.7 <u>Novation Agreements</u>. Duly executed counterparts of the Novation Agreements;
  - 8.2.8 <u>Core Access Agreement</u>. A duly executed counterpart of the Core Access Agreement;
  - 8.2.9 <u>Midstream Assignments</u>. Duly executed counterparts of the Midstream Assignments;
  - 8.2.10 <u>CHK Parent Guaranty</u>. A duly executed CHK Parent Guaranty;
  - 8.2.11 <u>Joint Written Instruction</u>. A duly executed joint written instruction to the Escrow Agent to release the Deposit;

- 8.2.12 <u>Releases</u>. Duly executed and acknowledged (where applicable) releases and terminations of any mortgages, deeds of trust, security interests, and other arrangements substantially equivalent thereto put in place by a Seller or its Affiliates and, in each case, burdening such Seller's interest in any of the Properties in support of existing amounts due and payable by a Seller or its Affiliates, in sufficient counterparts to facilitate recording in each county in which the Properties are located, as applicable, or (for terminations of financing statements) filing with the Secretary of State in the state where the applicable Seller is organized; and
- 8.2.13 <u>Additional Documents</u>. Such additional documents customary in similar transactions as might be reasonably requested by the Buyer and are reasonably required to consummate the transactions contemplated by this Agreement.
- 8.3 <u>Post-Closing Adjustments</u>. The Buyer and the Sellers agree that the Purchase Price will be further adjusted after the Closing Date in accordance with the provisions of <u>Article 2</u> of this Agreement.
- 8.4 <u>Post-Closing Deliveries</u>. At the Buyer's sole cost and expense, the Sellers shall deliver to the Buyer the Records as promptly as practicable, but no later than thirty (30) days after the termination of the TSA.

## ARTICLE 9 PRESS RELEASES

9. Press Releases. No Party shall make any press release or other public announcements concerning this transaction, without the consent of the other Party, which consent shall not be unreasonably withheld. Any Party desiring to make a public announcement shall first give the other Party twenty-four (24) hours written notification of its desire to make such a public announcement. The written notification shall include (a) a request for consent to make the announcement, and (b) a written draft of the text of such public announcement. Nothing contained herein shall prohibit any Party from issuing or making a public announcement or statement if such Party, upon advice of counsel, deems it necessary to do so in order to comply with any applicable Law, or the rules of any stock exchange upon which the Party's capital stock is traded, provided, however, that the foregoing procedure of written notification shall first be followed.

## ARTICLE 10 INDEMNIFICATION

- 10. <u>Indemnification</u>. Upon and after the Closing of the transactions contemplated by this Agreement, the Parties will indemnify each other as follows:
  - 10.1 <u>Assumed Obligations</u>. Upon the consummation of the Closing, the Buyer will assume all of the Assumed Obligations.

- Sellers' Indemnification. From and after the Closing, the Sellers shall pay, defend, indemnify, reimburse and hold harmless the Buyer, its Affiliates and each of their respective directors, partners, members, managers, officers, agents and employees (the "Buyer Indemnified Parties") for, from and against any loss, damage, diminution in value, claim, liability, debt, obligation or expense (including interest, reasonable legal fees, and expenses of litigation and reasonable attorneys' fees in enforcing this indemnity) incurred, suffered, paid by or resulting to any of the Buyer Indemnified Parties to the extent arising out of or in connection with: (a) any (i) breach of or default in any representation or warranty of the Sellers set forth in <a href="Article 3">Article 3</a> or confirmed in any certificate delivered by the Sellers at Closing or (ii) failure by a Seller to perform any covenant or obligation set forth in this Agreement which is not cured as provided in <a href="Section 12.1">Section 12.1</a> of this Agreement; (b) any of the Retained Liabilities; and (c) any breach of the Special Warranty or the limited warranties of title in the Mineral Deed and Surface Deed.
- Buyer's Indemnification. From and after the Closing, the Buyer shall pay, defend, indemnify, reimburse and hold harmless the Sellers, their Affiliates and each of their respective directors, partners, members, managers, officers, agents and employees (the "Seller Indemnified Parties") for, from and against any loss, damage, diminution in value, claim, liability, debt, obligation or expense (including interest, reasonable legal fees, and expenses of litigation and reasonable attorneys' fees in enforcing this indemnity) incurred, suffered, paid by or resulting to any of the Seller Indemnified Parties to the extent arising out of or in connection with (excluding, for purposes of certainty, any matters for which Sellers are obligated (or, upon the delivery of a claim notice, would otherwise be obligated) hereunder to indemnify the Buyer Indemnified Parties at the time the relevant claim notice is presented):

  (a) any (i) breach of or default in any representation or warranty of the Buyer set forth in this Agreement or confirmed in any certificate delivered by the Buyer at Closing or (ii) failure by the Buyer to perform any covenant or obligation set forth in this Agreement which is not cured as provided in Section 12.1 of this Agreement; and (b) any of the Assumed Obligations.
- Indemnification Procedure. If any indemnified party (i.e., the relevant Buyer Indemnified Parties or the Seller Indemnified Parties, as applicable) discovers or otherwise becomes aware of the basis for an indemnification claim arising under this Agreement, such party will promptly give written notice to the indemnifying party (i.e., the Party obligated hereunder to provide an indemnity to the Buyer Indemnified Parties or the Seller Indemnified Parties, as applicable), specifying such claim, and may thereafter exercise any remedies available to such indemnified party under this Agreement; provided, however, the failure of any indemnified party to give notice as provided herein will not relieve the indemnifying party of any obligations hereunder, to the extent the indemnifying party is not materially prejudiced thereby. Further, promptly after receipt by an indemnified party hereunder of written notice of the commencement of any action or proceeding with respect to which a claim for indemnification may be made

against any indemnifying party, the indemnified party will give written notice to the indemnifying party of the commencement of such action; <u>provided</u>, <u>however</u>, the failure of any indemnified party to give notice as provided herein will not relieve the indemnifying party of any obligations hereunder, to the extent the indemnifying party is not materially prejudiced thereby.

10.5 Defense. If any action discussed in Section 10.4 is brought against an indemnified party, the indemnifying party will be entitled to participate in and to assume the defense thereof to the extent that it may wish, and after notice from the indemnifying party to such indemnified party of the indemnifying party's election to assume the defense thereof, the indemnifying party shall not be liable to such indemnified party for any legal or other expenses subsequently incurred by the indemnified party in connection with the defense thereof unless the indemnifying party has failed to assume and diligently prosecute the defense of such claim. Notwithstanding any of the foregoing to the contrary, the indemnified party will be entitled to select its own counsel and assume the defense of any action brought against it if the indemnifying party fails to select counsel, the expenses of such defense to be paid by the indemnifying party. As a condition to the indemnifying party's obligations hereunder, the indemnified party will in good faith cooperate with and assist the indemnifying party in the prosecution or defense of such indemnified claim at no unreasonable expense to the indemnified party. No indemnifying party shall consent to entry of any judgment or enter into any settlement with respect to a claim either (a) without the consent of the indemnified party, which consent shall not be unreasonably withheld, conditioned or delayed, or (b) unless such judgment or settlement includes as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability with respect to such claim. Further, the Sellers shall not consent to entry of any judgment or enter into any settlement with respect to a Retained Litigation Matter that adversely affects the Properties in any material manner or respect for any time period from and after the Effective Time, without first consulting with the Buyer and attempting in good faith to reach a mutually acceptable agreement with the Buyer regarding such matter; provided, that, for the avoidance of doubt, any such adverse effect on the Properties for any time period from and after the Effective Time arising from the settlement of any such Retained Litigation Matter shall constitute a Retained Liability. No indemnified party shall consent to entry of any judgment or enter into any settlement of any such action, the defense of which has been assumed by an indemnifying party, without the consent of such indemnifying party, which consent shall not be unreasonably withheld, conditioned or delayed.

#### 10.6 Certain Limitations on Indemnity Obligations.

10.6.1

A. No individual claim of the Buyer or the Buyer Indemnified Parties pursuant to <u>Section 10.2(a)(i)</u> shall be made hereunder unless such individual claim exceeds an amount equal to **One Hundred Thousand Dollars** (\$100,000) (each an "**Individual Claim**").

In addition, no claim of the Buyer or the Buyer Indemnified Parties pursuant to Section 10.2(a)(i) shall be made hereunder until the total of all Individual Claims exceeds **two percent** (2.0%) of the unadjusted Purchase Price (the "Basket"). If the total amount of all of the Buyer's or the Buyer Indemnified Parties' Individual Claims pursuant to Section 10.2(a)(i) exceed the Basket, then the Sellers' obligations under Section 10.2(a)(i) shall be limited to the amount by which the aggregate amount of such Individual Claims exceeds the Basket.

- B. All of the representations and warranties set forth in this Agreement or confirmed in any certificate delivered at Closing that are qualified as to "material," "materiality," "material respects," "material adverse effect," "Material Adverse Effect" or words of similar import or effect shall be deemed to have been made without any such qualification for the purposes of determining whether a claim exceeds the Individual Claim threshold or Basket. Further, the determination hereunder of whether any breach or default of a representation or warranty confirmed in a certificate delivered by Buyer pursuant to <u>Section 8.2.4</u> shall be made without regard to the Material Adverse Effect qualification that appears at the end of <u>Section 6.1.3</u> and the proviso at the end of such Section.
- C. Notwithstanding anything to the contrary (including the foregoing), the limitations in this <u>Section 10.6.1</u> shall not apply to Fundamental Representations.
  - 10.6.2 Except with respect to Tax liabilities, notwithstanding anything in this Agreement to the contrary, in no event will the Sellers' aggregate liability (i) under <u>Section 10.2(a)(i)</u> (other than Fundamental Representations, to which this <u>clause (i)</u> shall not apply) exceed **twelve and one half percent (12.5%)** of the unadjusted Purchase Price or (ii) under or by reason of this Agreement or any of the transactions contemplated hereby exceed the unadjusted Purchase Price.
  - 10.6.3 The amount of any indemnification provided under <u>Section 10.2</u> or <u>10.3</u> shall be net of any amounts actually recovered by the indemnified party under insurance policies. Likewise, each Seller Indemnified Party or Buyer Indemnified Party, as applicable, shall use commercially reasonable efforts to mitigate his, her or its respective Liabilities upon and after becoming aware of any event or condition that would reasonably be expected to give rise to any Liabilities that are indemnifiable hereunder.
  - 10.6.4 Notwithstanding anything to the contrary contained in this Agreement, except for the rights of the Parties under Article 12, Section 2.2.13, Section 2.4.4, Section 5.1.2, Section 5.6, Section 5.14.4, Section 5.14.6, Section 11.2.5, Section 13.26, and Section 13.27, this Article 10 contains the Parties' exclusive remedy against each other with respect to breaches of this Agreement, including breaches of the Special Warranty, breaches of the representations and warranties contained in Article 3 and Article 4 or confirmed in any certificate delivered at Closing, the covenants and agreements that survive the Closing pursuant to the terms of this Agreement and the affirmations of such representations, warranties, covenants and agreements contained in the certificate delivered by each

Party at Closing pursuant to <u>Sections 8.1.5</u> or <u>8.2.4</u>, as applicable. Except for the remedies contained in this Article 10 and for the rights of the Parties under Article 12, Section 2.2.13, Section 2.4.4, Section 5.1.2, Section 5.6, Section 5.14.4, Section 5.14.6, Section 11.2.5, Section 13.26 and Section 13.27, each Party (on behalf of itself, each of the other Buyer Indemnified Parties or the other Seller Indemnified Parties, as applicable, and their respective insurers and successors in interest) releases, waives, remises and forever discharges the Buyer Indemnified Parties or the Seller Indemnified Parties, as applicable, from any and all suits, legal or administrative proceedings, claims, remedies, demands, damages, losses, costs, Liabilities, interest, or causes of action whatsoever, in Law or in equity, known or unknown, which such Party might now or subsequently may have, based on, relating to or arising out of this Agreement, the ownership, use or operation of the Properties, or the condition, quality, status or nature of the Properties, including rights to contribution under the CERCLA, as amended, and under other Environmental Laws, breaches of statutory or implied warranties, nuisance or other tort actions, rights to punitive damages and common law rights of contribution, and rights under agreements between any Party, as applicable, and any Persons who are Affiliates of such Party, as applicable, EVEN IF CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT, BUT EXCLUDING, IN EACH CASE, INSTANCES OF FRAUD, GROSS NEGLIGENCE OR WILLFUL MISCONDUCT) OF ANY RELEASED PERSON.

- 10.6.5 Notwithstanding anything stated herein to the contrary, no Party will have any liability to another Party or such other Party's indemnified parties under this <u>Section 10</u> with respect to any item for which (and to the extent) a specific adjustment has already been made to the Purchase Price or payment made under the terms of this Agreement. For the avoidance of doubt, claims for Title Defects or Environmental Defects made pursuant to <u>Section 2.2</u>, whether or not resulting in a Purchase Price adjustment, may not be claimed under this <u>Section 10</u>, may not be included for purposes of determining whether the limitations set forth in this <u>Section 10.6</u> have been met and may not be included in the Basket for purposes of the limitations set forth in this <u>Section 10.6</u>.
- 10.6.6 Notwithstanding anything to the contrary contained in this Agreement, if any Party elects to proceed with the Closing with knowledge by such Party of any failure of the condition set forth in Section 6.1.3 or Section 7.1.3, as applicable, to be satisfied in its favor as a result of a breach of any representation and warranty by the other Party, then such condition (to the extent of, and only with respect to, the matter directly causing or resulting in such failed condition) that is unsatisfied and such representation and warranty (to the extent of, and only with respect to, the matter directly causing or resulting in such breach) that is breached will be deemed waived by such Party, and such Party shall be deemed to fully release and

forever discharge the other Party on account of any and all claims, demands or charges, known or unknown, with respect to such condition, representation or warranty (to the extent of, and only with respect to, the matter directly causing or resulting in such failed condition or breach), in each case, if, and only if, the matter(s) that directly caused or resulted in such failed condition or breach of representation and warranty first occurred during the period between the Execution Date and the Closing Date.

- 10.6.7 Any claim for indemnity to which a Seller Indemnified Party or Buyer Indemnified Party is entitled hereunder must (and can only) be asserted by and through a Seller or the Buyer, as applicable, pursuant to the terms of this <a href="Article 10">Article 10</a>; and, for the avoidance of doubt, no Seller Indemnified Party or Buyer Indemnified Party other than the Parties shall have rights against either a Seller or the Buyer under the terms of this <a href="Article 10">Article 10</a> except as may be exercised on its behalf by the Buyer or a Seller, as applicable, pursuant to this <a href="Article 10">Article 10</a>. The Parties may elect to exercise or not exercise indemnification rights under this <a href="Article 10">Article 10</a> on behalf of the other Seller Indemnified Parties or Buyer Indemnified Parties, as applicable, in its sole discretion and shall have no liability to any such other Seller Indemnified Parties or Buyer Indemnified Parties, as applicable, for any action or inaction under this Article 10.
- 10.7 EXTENT OF INDEMNIFICATION. WITHOUT LIMITING OR ENLARGING THE SCOPE OF THE INDEMNIFICATION, DEFENSE AND ASSUMPTION PROVISIONS SET FORTH IN THIS AGREEMENT, TO THE FULLEST EXTENT PERMITTED BY LAW, AN INDEMNIFIED PERSON SHALL BE ENTITLED TO INDEMNIFICATION HEREUNDER IN ACCORDANCE WITH THE TERMS OF SECTIONS 10.2 OR 10.3, REGARDLESS OF WHETHER THE ACT, OCCURRENCE OR CIRCUMSTANCE GIVING RISE TO ANY SUCH INDEMNIFICATION OBLIGATION IS THE RESULT OF THE SOLE, ACTIVE, PASSIVE, CONCURRENT OR COMPARATIVE NEGLIGENCE, STRICT LIABILITY, BREACH OF DUTY (STATUTORY OR OTHERWISE), OR OTHER FAULT OR VIOLATION OF ANY LAW OF OR BY ANY SUCH INDEMNIFIED PERSON, PROVIDED THAT NO SUCH INDEMNIFICATION SHALL BE APPLICABLE TO THE EXTENT OF ANY GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF THE INDEMNIFIED PERSON.

## ARTICLE 11 BOOKS AND RECORDS

- 11. <u>Preservation of Books and Records; Financial Statements.</u>
  - 11.1 <u>Preservation of Books and Records</u>. Except as provided in <u>Section 5.6.6</u>, for a period of five (5) years after the Closing Date, the Buyer will, using procedures

consistent with its current record retention procedures, preserve and retain all books and records in its possession that relate to the Properties including any documents relating to any governmental or nongovernmental actions, suits, proceedings or investigations arising out of the operation of the Properties prior to the Closing Date. During such five (5) year period, the Buyer agrees to provide the Sellers and their agents and representatives with reasonable access to such books and records upon reasonable notice and at reasonable times, to the extent that such books and records are in the possession of the Buyer and not any of the Sellers (and to the extent doing so is not in violation of any Contracts or Laws or results in the waiver of any protections).

### 11.2 Financial Statements.

11.2.1 Sellers acknowledge that Buyer and its Affiliates may be required to include financial statements and other financial information related to the Properties, including carve-out financial statements related to the Properties, as of and for the years ended December 31, 2017 and 2016 and financial statements for any interim periods ending on or prior to the Closing and the supplemental disclosures of oil and gas required by Accounting Standards Codification Topic 932-235-50-2 for any such periods (collectively, the "Financial **Statements**"), and that such Financial Statements may be required to be audited or reviewed in accordance with GAAP and may need to comply with the requirements of the Securities Exchange Commission for inclusion or incorporation by reference into one or more registration statements, reports or other documents (collectively, "SEC Documents") required to be filed by Buyer or its Affiliates under the Securities Act of 1933, as amended (the "Securities Act"), the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the rules set forth in Regulation S-X, or other rules promulgated thereunder or in an offering memorandum relating to a private placement of securities exempt from registration under the Securities Act ("**Offering Document**"). From and after the Closing Date until the second anniversary thereof, Sellers shall, shall cause their Affiliates to, and shall use commercially reasonable efforts to cause their accountants and counsel to, cooperate with Buyer, its Affiliates and their respective agents, advisors and representatives in their effort to prepare the Financial Statements to the extent that Sellers or such other Persons has information available or can obtain such information using commercially reasonable efforts. Further, from and after the Closing Date until the second anniversary thereof and following reasonable advance notice from Buyer to Sellers, Sellers shall, shall cause their Affiliates to, and shall use commercially reasonable efforts to cause their accountants and counsel to, provide reasonable access during normal business hours to Buyer and its Affiliates and their agents and advisors to any and all books, records, information and documents if reasonably required by Buyer or its Affiliates in connection with the creation and audit or review of the Financial Statements. Sellers shall use commercially reasonable efforts to cause its independent public accountants (the "Seller Auditor") to

cooperate with Buyer in the creation and audit or review of the Financial Statements. Buyer shall be responsible, and obligated to promptly reimburse Sellers, for any and all reasonable external and internal costs and expenses incurred by Sellers or their Affiliates to the extent associated with the preparation of the Financial Statements pursuant to this <u>Section 11.2</u> and otherwise complying with the provisions of this <u>Section 11.2</u>.

- 11.2.2 To the extent reasonably requested by Buyer, from and after Closing until February 15, 2021, Sellers shall use their commercially reasonable efforts to obtain representation letters and similar documents (in each case, in form and substance customary for representation letters provided to external audit firms by management of a company whose financial statements are the subject of an audit or review used in filings of acquired company financial statements under the Exchange Act) from applicable personnel of Sellers and their Affiliates as may be required in connection with the preparation and audit or review of the Financial Statements or delivery of a "comfort letter" for a securities offering by Buyer or its Affiliates and solely to the extent directly related to the Financial Statements; provided, that Buyer shall provide customary indemnity for any officer or employee of Sellers or their Affiliates executing any such representation letter. To the extent requested by Buyer, from and after Closing until the second anniversary thereof, Sellers shall use their commercially reasonable efforts to cause each independent audit firm that audits or reviews the Financial Statements to provide consents necessary for the inclusion or incorporation by reference of the Financial Statements in any SEC Document or any Offering Document in which the Financial Statements are required to be included or incorporated.
- 11.2.3 Requests by Buyer for cooperation, access and documentation pursuant to this <u>Section 11.2</u> shall be given with reasonable specificity and with reasonable advance notice to Sellers so as to not unreasonably interfere with any of Sellers' or any of Sellers' representatives' conduct of business. In no event shall Sellers' obligations under this Section 11.2 (a) require any Seller to create or obtain information or records that are not reasonably available to such Seller or its Affiliates or (b) interfere with Sellers' ability to prepare their own financial statements or their regular conduct of business.
- 11.2.4 For a period of two (2) years following the Closing Date, Sellers shall, and shall cause its respective Affiliates to, use commercially reasonable efforts to retain all books, records, information and documents in their or their Affiliates' possession that are necessary to prepare and audit the Financial Statements, except to the extent originals or copies thereof are transferred to Buyer in connection with Closing.
- 11.2.5 Except for the express representations and warranties of such Seller in <u>Article 3</u>, the representations and warranties confirmed in the certificate

delivered at Closing by the Sellers pursuant to Section 8.2.4, and the Special Warranty and subject and without prejudice to the Buyer's indemnification rights under Article 10: (a) all of the information provided by Sellers or their Affiliates pursuant to this Section 11.2 is given without any representation or warranty, express or implied, regarding the completeness or accuracy of such information, and no Seller Indemnified Parties shall have any liability or responsibility with respect thereto; (b) Buyer, for itself and for its Affiliates, hereby releases, remises and forever discharges each Seller Indemnified Party from any and all suits, legal or administrative proceedings, claims, demands, damages, losses, costs, liabilities, interest, or causes of action whatsoever, in Law or in equity, known or unknown, which any of Buyer or its Affiliates might now or subsequently may have, based on, relating to or arising out of the completeness or accuracy of any information provided by any Seller pursuant to this Section 11.2; and (c) from and after Closing, Buyer agrees to indemnify, defend and hold harmless the Seller Indemnified Parties from and against any and all claims, liabilities, losses, costs and expenses (including court costs, expert fees and reasonable attorneys' fees), based on, relating to or arising out of the completeness, accuracy or use of any information provided by any Seller pursuant to this Section 11.2, EVEN IF CAUSED IN WHOLE OR IN PART BY THE NEGLIGENCE (WHETHER SOLE, JOINT OR CONCURRENT), STRICT LIABILITY OR OTHER LEGAL FAULT OF ANY INDEMNIFIED PERSON EXCLUDING, HOWEVER, ANY CLAIMS, LIABILITIES, LOSSES, COSTS OR EXPENSES CAUSED BY THE FRAUD, GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF ANY INDEMNIFIED PERSON.

## ARTICLE 12 TERMINATION

- 12. <u>Termination</u>. This Agreement may be terminated and the transactions contemplated hereby may be abandoned as follows:
  - 12.1 <u>Right to Terminate</u>. Subject to <u>Section 12.2</u>, this Agreement may be terminated at any time prior to the consummation of the Closing upon the occurrence of any one or more of the following:
    - 12.1.1 by mutual consent of the Sellers and the Buyer;
    - 12.1.2 by the Buyer, if the Sellers have materially breached this Agreement and such breach causes any of the conditions to Closing set forth in <u>Article 6</u> not to be satisfied as of the Target Closing Date; <u>provided</u>, <u>however</u>, that in the case of a breach that is capable of being cured, the Sellers shall have a period of ten (10) days following receipt of such notice to attempt to cure the breach and the termination under this <u>Section 12.1.2</u> shall not become effective unless the Sellers fail to cure such breach prior to the

- earlier of (a) the end of such ten (10) day period and (b) one (1) Business Day prior to the Outside Date;
- 12.1.3 by the Sellers, if the Buyer has materially breached this Agreement and such breach causes any of the conditions to Closing set forth in <u>Article 7</u> not to be satisfied as of the Target Closing Date; <u>provided</u>, <u>however</u>, that in the case of a breach that is capable of being cured, the Buyer shall have a period of ten (10) days following receipt of such notice to attempt to cure the breach and the termination under this <u>Section 12.1.3</u> shall not become effective unless the Buyer fails to cure such breach prior to the earlier of (a) the end of such ten (10) day period and (b) one (1) Business Day prior to the Outside Date;
- 12.1.4 by the Sellers if the Buyer fails to pay the Deposit on or before 5:00 p.m. (Central Time) on the second (2nd) Business Day after the Execution Date;
- 12.1.5 by the Sellers or the Buyer if the Closing shall not have occurred on or before December 31, 2018 (the "**Outside Date**"); or
- 12.1.6 by the Buyer applicable pursuant to Schedule 2.4.4;

<u>provided</u> that a Party may not terminate this Agreement pursuant to <u>Sections 12.1.2</u> or <u>12.1.3</u> above if such Party is, at such time, in material breach of this Agreement, and <u>provided</u>, <u>further</u>, that a Party may not terminate this Agreement pursuant to <u>Section 12.1.5</u> above if the Closing has not occurred as a result of the terminating Party's breach of this Agreement.

12.2 Effect of Termination. In the event of termination, written notice thereof will be given to the other Party specifying the provision pursuant to which such termination is made. Except as specifically provided in Section 12.3, on the termination of this Agreement the Deposit will be refunded to the Buyer. If this Agreement is terminated in accordance with Section 12.1, the provisions contained in this Section 12 and in Sections 1.2, 2.4.4 (as to only the Buyer Pref Termination Right and the payment of the Total Pref Right Break-fee, as set forth in Schedule 2.4.4), 5.1.2, 5.14), 9, 11.2.5, 13.1, 13.2, 13.5 - 13.18, 13.22, 13.25, 13.27 - 13.29 and such defined terms in Article 1 as may be required to give meaning to such sections, and the Confidentiality Agreement, shall survive such termination of this Agreement. Except as set forth in Section 12.3.1, no termination of this Agreement under Section 12.1 shall relieve any Party of liability for any material breach of this Agreement arising prior to such termination. For the avoidance of doubt, in the event of termination of this Agreement, no Debt Financing Source will have any liability to any Seller, any of its or their respective Affiliates or any of its or their direct or indirect stockholders hereunder or otherwise relating to or arising out of the transactions contemplated hereunder (including for any willful and material breach); provided that the foregoing shall not preclude any liability of any Debt Financing Source to the

Buyer under the terms of any commitment letter, purchase agreement or definitive documentation with respect to the Financing, as applicable.

## 12.3 <u>Distribution of Deposit Upon Termination</u>.

- 12.3.1 If this Agreement is terminated by the Sellers pursuant to Section 12.1.3 or Section 12.1.5 (and at such time the Buyer does not have the right to terminate this Agreement pursuant to Section 12.1) and the Sellers have performed or are ready, willing and able to perform all of their agreements and covenants contained herein which are to be performed or observed at or prior to Closing, the Sellers may receive the Deposit as liquidated damages and as the Sellers' sole and exclusive remedy for any breach or failure to perform by the Buyer under this Agreement, and all other remedies (except those under Section 5.1.2 and the Confidentiality Agreement, which shall remain in full force and effect) are hereby expressly waived by the Sellers. Subject to an election by the Sellers under Section 12.3.2, the Sellers and the Buyer agree upon the Deposit as liquidated damages due to the difficulty and inconvenience of measuring actual damages and the uncertainty thereof, and the Sellers and the Buyer agree that such amount would be a reasonable estimate of the Sellers' loss in the event of any such breach or failure to perform by the Buyer. Upon such termination, the Sellers shall be free immediately to enjoy all rights of ownership of the Properties and to sell, transfer, encumber or otherwise dispose of the Properties to any Person without any restriction under this Agreement, and no later than one (1) Business Day of such termination, the Parties shall deliver joint written instructions to the Escrow Agent requesting for it to immediately release the Deposit to the Sellers.
- 12.3.2 If the Sellers have the right to terminate this Agreement pursuant to Section 12.1.3 or Section 12.1.5 (and at such time the Buyer does not have the right to terminate this Agreement pursuant to Section 12.1) and the Sellers have performed or are ready, willing and able to perform all of their agreements and covenants contained herein which are to be performed or observed at or prior to Closing, then, in lieu of termination of this Agreement, the Sellers shall be entitled to specific performance of this Agreement, it being specifically agreed that monetary damages will not be sufficient to compensate the Sellers if the Sellers determine the same in their sole discretion. If the Sellers elect to seek specific performance of this Agreement pursuant to this Section 12.3.2, the Deposit shall be held by the Escrow Agent, not as liquidated damages, until a non-appealable final judgment or award on the Sellers' claim for specific performance is rendered, at which time the Deposit shall be distributed as provided in the judgment or award resolving the specific performance claim or shall be applied as provided in Article 2 of this Agreement.
- 12.3.3 If this Agreement is terminated by the Buyer pursuant to <u>Section 12.1.2</u> or <u>12.1.5</u> (and at such time the Sellers do not have the right to terminate this

Agreement pursuant to <u>Section 12.1</u>) and the Buyer has performed or is ready, willing and able to perform all of its agreements and covenants contained herein which are to be performed or observed at or prior to Closing, then, no later than one (1) Business Day of such termination, the Parties shall deliver joint written instructions to the Escrow Agent requesting for it to immediately release the Deposit to the Buyer (free of any claims by the Sellers or any other Person with respect thereto) and the Buyer shall be entitled to seek money damages from the Sellers available at Law for the Sellers' applicable breach of this Agreement, as the Buyer's sole and exclusive remedy for any breach or failure to perform by the Sellers under this Agreement, and all other remedies (except under the Confidentiality Agreement and any other provisions of this Agreement that survive termination hereof, including <u>Section 2.4.4</u> (as to only the Buyer Pref Termination Right and the payment of the Total Pref Right Break-fee, as set forth in <u>Schedule 2.4.4</u>), which shall remain in full force and effect) are hereby expressly waived by the Buyer, and the Sellers shall be free immediately to enjoy all rights of ownership of the Properties and to sell, transfer, encumber or otherwise dispose of the Properties to any Person without any restriction under this Agreement.

- 12.3.4 If the Buyer has the right to terminate this Agreement pursuant to Section 12.1.2 or Section 12.1.5 (and at such time the Sellers do not have the right to terminate this Agreement pursuant to Section 12.1) and the Buyer has performed or is ready, willing and able to perform all of its agreements and covenants contained herein which are to be performed or observed at or prior to Closing, then, in lieu of termination of this Agreement, the Buyer shall be entitled to specific performance of this Agreement, it being specifically agreed that monetary damages will not be sufficient to compensate the Buyer if the Buyer determines the same in its sole discretion. If the Buyer elects to seek specific performance of this Agreement pursuant to this Section 12.3.4, the Deposit shall be held by the Escrow Agent, until a non-appealable final judgment or award on the Buyer's claim for specific performance is rendered, at which time the Deposit shall be distributed as provided in the judgment or award resolving the specific performance claim or shall be applied as provided in Article 2 of this Agreement.
- 12.3.5 If this Agreement terminates for reasons other than those set forth in Section 12.3.1 or Section 12.3.3 (unless termination has occurred pursuant to Section 12.1.4), no later than one (1) Business Day of such termination, the Parties shall deliver joint written instructions to the Escrow Agent instructing it to deliver the Deposit to the Buyer, free of any claims by the Sellers or any other Person with respect thereto, and each Party shall have no further Liability hereunder of any nature whatsoever to the other Party (except under the Confidentiality Agreement and any other provisions of this Agreement that survive termination hereof, including Section 2.4.4 (as to only the Buyer Pref Termination Right and the payment of the Total

Pref Right Break-fee, as set forth in <u>Schedule 2.4.4</u>), which shall remain in full force and effect), and the Sellers shall be free immediately to enjoy all rights of ownership of the Properties and to sell, transfer, encumber or otherwise dispose of the Properties to any Person without any restriction under this Agreement.

Buyer Indemnification Support. If, prior to the return of the Deposit to Buyer pursuant to this Article 12, Sellers have asserted any indemnification claim against Buyer pursuant to this Agreement which is at that time unsatisfied (for purposes of this Section 12.4, the amount of any such unsatisfied claim being the "indemnity claim amount"), then, notwithstanding anything else herein to the contrary, the Parties shall only be required to deliver joint written instructions to the Escrow Agent to deliver the Deposit less the indemnity claim amount to Buyer and the Escrow Agent shall retain the indemnity claim amount from the Deposit pursuant to the terms of this Section 12.4. Upon final resolution or determination of all such claims of Sellers, the Parties shall deliver joint instructions to the Escrow Agent to release to Sellers the aggregate amount payable (if any) to Sellers to satisfy such finally resolved or determined claims and the remaining portion of the Deposit, if any, shall be released to Buyer.

## ARTICLE 13 MISCELLANEOUS

- 13. <u>Miscellaneous</u>. It is further agreed as follows:
  - Time. This Agreement contains a number of dates and times by which performance or the exercise of rights is due, and the Parties intend that each and every such date and time be the firm and final date and time, as agreed. For this reason, each Party hereby waives and relinquishes any right it might otherwise have to challenge its failure to meet any performance or rights election date applicable to it on the basis that its late action constitutes substantial performance, to require the other Party to show prejudice (except as may expressly be set forth herein), or on any equitable grounds. Without limiting the foregoing, time is of the essence of this Agreement. If the date specified in this Agreement for giving any notice or taking any action, including the tendering of any payment hereunder, is not a Business Day (or if the period during which any notice is required to be given or any action taken expires on a date which is not a Business Day), then the date for giving such notice, taking such action, or making such payment (and the expiration date of such period during which notice is required to be given or action taken) shall be the next day which is a Business Day.
  - Notices. All notices and communications required or permitted under this Agreement shall be in writing addressed as indicated below, and any communication or delivery hereunder shall be deemed to have been duly delivered upon the earliest of: (a) actual receipt by the Party to be notified; (b) if sent by U.S. certified mail, postage prepaid, return receipt requested, then the date shown as received on the return notice; (c) if by email, then upon an affirmative

reply by email by the intended recipient that such email was received (<u>provided</u> that, for the avoidance of doubt, an automated response from the email account or server of the intended recipient shall not constitute an affirmative reply); or (d) if by Federal Express overnight delivery (or other reputable overnight delivery service), the date shown on the notice of delivery. Addresses for all such notices and communication shall be as follows:

To the Sellers: Chesapeake Exploration, L.L.C.

6100 North Western Avenue Oklahoma City, Oklahoma 73118 Attention: Mr. Bryan Lemmerman Telephone: (405) 935-1818

Email: bryan.lemmerman@chk.com

With a copy to (which will not constitute notice):

Chesapeake Exploration, L.L.C. 6100 North Western Avenue Oklahoma City, Oklahoma 73118 Attention: Mr. James R. Webb Telephone: 405-935-9594 Email: jim.webb@chk.com

To the Buyer: EAP Ohio, LLC

675 Bering Drive, Suite 360 Houston, Texas 77057

Attention: Mr. Michael Magilton Telephone: 281-254-7043

Email: mmagilton@encinoenergy.com

With a copy to (which will not constitute notice):

EAP Ohio, LLC

675 Bering Drive, Suite 360 Houston, Texas 77057

Attention: Mr. Hardy Murchison

Telephone: 713-805-6440

Email: hardy@encinoenergy.com

Either Party may, upon written notice to the other Party, change the address(es) and person(s) to whom such communications are to be directed.

13.3 <u>Survival</u>. The respective representations and warranties of the Sellers contained in <u>Article 3</u> of this Agreement or confirmed in any certificate delivered in connection with this Agreement (together with the indemnification rights with respect thereto, subject to the proviso below) (other than the Fundamental Representations and the representations and warranties set forth in <u>Section 3.5</u>, in each case, which shall survive the Closing until sixty (60) days following the expiration of the applicable statute of limitations) will survive the Closing for a

period of twelve (12) months after the Closing Date and shall thereafter be of no further force or effect (the "Expiration Date"). The Special Warranty and the limited warranties of title in the Mineral Deed and Surface Deed (together with the indemnification rights with respect thereto, subject to the proviso below) will survive for a period of twenty-four (24) months from and after the Closing Date. Covenants to be performed at or prior to Closing shall terminate twelve (12) months after the Closing Date and covenants to be performed after Closing shall survive until performed. The remainder of this Agreement shall survive the Closing without time limit, except as may otherwise be expressly provided herein. The intended effect of the expiration or termination of representations, warranties, covenants, agreements and the Special Warranty (and the indemnification rights with respect thereto, subject to the proviso below) is to bar, from and after the date of expiration or termination, any claim or cause of action based on (a) the alleged inaccuracy of such representation or breach of such warranty (including the Special Warranty), (b) the alleged breach or failure to fulfill such covenant or agreement, or (c) claims for indemnity with respect to clauses (a) and (b) above; provided, however, that, notwithstanding anything to the contrary, any representation, warranty, covenant or agreement as to which a claim shall have been asserted prior to the Expiration Date or its termination date, as applicable, shall survive solely with respect to such claim until such claim and the indemnity with respect thereto are resolved.

- 13.4 <u>Cooperation</u>. After Closing, the Parties agree to execute and deliver, or cause their respective Affiliates to execute and deliver, such documents and do, or cause to be done, such other acts as reasonably requested by any Party to this Agreement for (a) carrying out the purposes of this Agreement, it being the intent of the Sellers to transfer, assign, and convey unto the Buyer all of the collective right, title and interest of each Seller and its respective Affiliates in the Properties and (b) transferring to the Buyer any interest or asset located within the Target Area of Affiliates of the Sellers that would otherwise constitute a Property hereunder were such Affiliates joined hereto as Sellers (which thereafter shall constitute "Properties" for all purposes hereof).
- 13.5 No Third Party Beneficiaries. Except for the indemnification rights of the Seller Indemnified Parties and the Buyer Indemnified Parties under Article 10, nothing in this Agreement, express or implied, is intended to confer upon anyone, other than the Parties and their respective successors and assigns, any rights or remedies under or by reason of this Agreement or to constitute any Person a Third Party beneficiary of this Agreement; provided, however, that notwithstanding the foregoing, the Debt Financing Sources, their Affiliates and their respective Representatives shall be express third party beneficiaries of, and shall be entitled to enforce (and entitled to rely upon), Section 12.2, this Section 13.5, Section 13.7, Section 13.8, Section 13.11, Section 13.27, Section 13.28, and Section 13.29.
- 13.6 <u>Cumulative Remedies</u>. Subject to the other provisions hereof, no failure on the part of any Party to this Agreement to exercise and no delay in exercising any

right hereunder will operate as a waiver thereof, nor will any single or partial exercise by any Party of any right hereunder preclude any other or further right of exercise thereof or the exercise of any other right.

- 13.7 CHOICE OF LAW; VENUE. THIS AGREEMENT AND THE LEGAL RELATIONS AMONG THE PARTIES WILL BE INTERPRETED, CONSTRUED, GOVERNED AND ENFORCED IN ACCORDANCE WITH THE LAWS OF THE STATE OF TEXAS, WITHOUT REGARD TO ANY RULES OR PRINCIPLES OF CONFLICTS OF LAW THAT MIGHT OTHERWISE REFER TO OR REQUIRE THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION; PROVIDED, THAT WITH RESPECT TO ANY ACTION INCLUDING THE DEBT FINANCING SOURCES, THE LAWS OF THE STATE OF NEW YORK SHALL GOVERN. THE VENUE FOR ANY ACTION BROUGHT UNDER THIS AGREEMENT SHALL BE HARRIS COUNTY, TEXAS. THE PARTIES HEREBY IRREVOCABLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAWS, ANY OBJECTION WHICH THEY MAY NOW OR HEREAFTER HAVE TO THE LAYING OF VENUE OF ANY SUCH DISPUTE, CONTROVERSY OR CLAIM BROUGHT IN ANY SUCH COURT OR ANY DEFENSE OF INCONVENIENT FORUM FOR THE MAINTENANCE OF SUCH DISPUTE, CONTROVERSY OR CLAIM.
- 13.8 JURISDICTION. EACH PARTY HEREBY IRREVOCABLY SUBMITS AND CONSENTS TO THE EXCLUSIVE JURISDICTION IN ANY ACTION BROUGHT IN THE UNITED STATES FEDERAL COURTS LOCATED WITHIN HARRIS COUNTY, TEXAS (OR, IF JURISDICTION IS NOT AVAILABLE IN THE UNITED STATES FEDERAL COURTS, TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE STATE COURTS LOCATED IN HARRIS COUNTY, TEXAS) WITH RESPECT TO ANY DISPUTE, CLAIM OR CONTROVERSY ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THIS AGREEMENT, AND EACH OF THE PARTIES AGREES THAT ANY ACTION INSTITUTED BY IT AGAINST THE OTHER WITH RESPECT TO ANY SUCH DISPUTE, CONTROVERSY OR CLAIM (EXCEPT TO THE EXTENT A DISPUTE, CONTROVERSY, OR CLAIM ARISING OUT OF OR IN RELATION TO OR IN CONNECTION WITH THE RESOLUTION OF ANY CURABLE DEFECTS, DISPUTED TITLE OR ENVIRONMENTAL DEFECTS, OR TITLE BENEFITS PURSUANT TO SECTION 2.2.12, OR THE DETERMINATION OF PURCHASE PRICE ADJUSTMENTS PURSUANT TO SECTION 2.9 IS REFERRED TO AN EXPERT PURSUANT TO THOSE SECTIONS) WILL BE INSTITUTED EXCLUSIVELY IN THE UNITED STATES FEDERAL COURTS LOCATED WITHIN HARRIS COUNTY, TEXAS (OR, IF JURISDICTION IS NOT AVAILABLE IN THE UNITED STATES FEDERAL COURTS, TO PERSONAL JURISDICTION IN ANY ACTION BROUGHT IN THE STATE COURTS LOCATED IN HARRIS COUNTY, TEXAS). EACH PARTY AGREES THAT A JUDGMENT IN ANY SUCH DISPUTE MAY BE ENFORCED IN OTHER JURISDICTIONS BY SUIT ON THE JUDGMENT OR

IN ANY OTHER MANNER PROVIDED BY APPLICABLE LAW. NOTWITHSTANDING ANYTHING TO THE CONTRARY, THE PARTIES FURTHER AGREE THAT THE PARTIES SHALL NOT BRING ANY SUIT ARISING OUT OF OR RELATING TO THIS AGREEMENT OR ANY OF THE TRANSACTIONS AGAINST THE DEBT FINANCING SOURCES, INCLUDING WITH RESPECT TO ANY DISPUTE ARISING OUT OF OR RELATING IN ANY WAY TO ANY OF THE FINANCING OR IN RESPECT OF ANY DEFINITIVE DOCUMENTATION IN RESPECT THEREOF OR THE PERFORMANCE THEREOF, IN ANY FORUM OTHER THAN THE UNITED STATES DISTRICT COURT FOR THE SOUTHERN DISTRICT OF NEW YORK OR ANY COURT OF THE STATE OF NEW YORK SITTING IN THE BOROUGH OF MANHATTAN IN THE CITY OF NEW YORK, AND AGREE THAT THE WAIVER OF JURY TRIAL SET FORTH IN SECTION 13.16 SHALL BE APPLICABLE TO ANY SUCH PROCEEDING.

- 13.9 Entire Agreement. This Agreement and the Exhibits and Schedules attached hereto, the Assignment, the Surface Deed, the Mineral Deed and the other documents contemplated by this Agreement, together with the Confidentiality Agreement, constitute the entire agreement between the Parties with respect to the subject matter hereof and there are no agreements, understandings, warranties or representations except as set forth herein or therein. In the event of a conflict between the Confidentiality Agreement and this Agreement, the terms and provisions of this Agreement shall prevail.
- 13.10 <u>Assignment</u>. Except as contemplated in <u>Section 13.19</u>, it is agreed that no Party may assign this Agreement in whole or in part without the express written consent of Sellers (in the case of Buyer) or Buyer (in the case of any Seller), as the case may be, and no such assignment under <u>Section 13.19</u> shall be deemed to have released the assigning Party from any of its obligations under this Agreement. The restrictions on or requirements for assignment in this <u>Section 13.10</u> shall not limit or apply to the Buyer's (or its successors or assigns) ability to assign all or part of its interest in the Properties after Closing (including, for collateral security purposes, to the Debt Financing Sources).
- Amendment. Neither this Agreement, nor any of the provisions hereof may be amended, supplemented, changed, waived, discharged, or terminated, except by an instrument in writing signed by all Parties. Any waiver of performance of any obligation or term contained in this Agreement by any Party, or waiver by one Party of another's default hereunder will not operate as a waiver of performance of any other obligation or term of this Agreement or a future or continuing waiver of the same obligation or a waiver of any future default. Notwithstanding anything to the contrary in this Section 13.11 or in Article 12, (i) this Agreement may not be amended, supplemented or modified with respect to Section 12.2, Section 13.5, Section 13.7, Section 13.8, this Section 13.11, Section 13.27, Section 13.28, Section 13.29 or the definition of "Debt Financing Sources", and (ii) no term or condition of this Agreement with respect to Section 13.5, Section 13.7, Section 13.8, this Section 13.11, Section 13.27, Section 13.28, Section 13.29

or the definition of "Debt Financing Sources" may be waived by any Party, in the case of each of <u>clauses (i)</u> and <u>(ii)</u>, to the extent such amendment, supplement, modification or waiver would modify the substance of such sections, in a manner that impacts the express rights of the Debt Financing Sources without the written consent of the Debt Financing Sources.

- 13.12 <u>Severability</u>. If any clause or provision of this Agreement is illegal, invalid or unenforceable under any present or future Law, the remainder of this Agreement will not be affected thereby. It is the intention of the Parties that if any such provision is held to be illegal, invalid or unenforceable, there will be added in lieu thereof a provision as similar in terms to such provisions as is possible to make such provision legal, valid and enforceable.
- 13.13 <u>Attorney Fees</u>. If any Party institutes an action or proceeding against any other Party relating to the provisions of this Agreement, the Party to such action or proceeding that does not prevail will reimburse the prevailing Party therein for the reasonable expenses of attorneys' fees and disbursements incurred by the prevailing Party.
- 13.14 <u>Counterparts; Electronic Transmission</u>. This Agreement may be executed in multiple counterparts, each of which will be an original instrument, but all of which when combined shall constitute but one agreement. The execution and delivery of this Agreement by any Party may be evidenced by electronic transmission (including scanned documents delivered by email), which shall be binding upon any Party so executing.
- 13.15 <u>LIMITATION ON DAMAGES</u>. NOTWITHSTANDING ANYTHING TO THE CONTRARY, EXCEPT IN CONNECTION WITH ANY DAMAGES INCURRED BY THIRD PARTIES FOR WHICH INDEMNIFICATION IS SOUGHT UNDER THE TERMS OF THIS AGREEMENT, NONE OF BUYER, SELLERS OR ANY OF THEIR RESPECTIVE AFFILIATES SHALL BE ENTITLED TO CONSEQUENTIAL, SPECIAL, PUNITIVE OR EXEMPLARY DAMAGES IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY AND, EXCEPT AS OTHERWISE PROVIDED IN THIS SENTENCE, EACH OF BUYER AND SELLERS, FOR ITSELF AND ON BEHALF OF ITS AFFILIATES, HEREBY EXPRESSLY WAIVES ANY RIGHT TO CONSEQUENTIAL, SPECIAL, PUNITIVE OR EXEMPLARY DAMAGES IN CONNECTION WITH THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY.
- 13.16 WAIVER OF JURY TRIAL. EACH OF THE BUYER AND THE SELLERS HEREBY KNOWINGLY, VOLUNTARILY, INTENTIONALLY AND IRREVOCABLY WAIVES, TO THE MAXIMUM EXTENT NOT PROHIBITED BY LAW, ANY RIGHT IT MAY HAVE TO A TRIAL BY A JURY IN RESPECT OF ANY LITIGATION BASED HEREON, OR DIRECTLY OR INDIRECTLY AT ANY TIME ARISING OUT OF, UNDER OR IN CONNECTION WITH THIS AGREEMENT OR ANY TRANSACTION CONTEMPLATED HEREBY OR ASSOCIATED HEREWITH.

- 13.17 <u>Mutuality</u>. The Parties acknowledge and declare that this Agreement is the result of extensive arms-length negotiations between them. Accordingly, if there is any ambiguity in this Agreement, there shall be no presumption that this instrument was prepared solely by any Party and any rule of construction that a contract be construed against the drafter shall not apply to the interpretation or construction of this Agreement.
- 13.18 Schedules. The inclusion of any information (including Dollar amounts) in any section of the disclosure Schedules hereto shall not be deemed to be an admission or acknowledgment by a Seller that such information is required to be listed on such Schedule or is material to or outside the Ordinary Course of Business of such Seller. The information contained in this Agreement, the Exhibits and the Schedules hereto is disclosed solely for purposes of this Agreement, and no information contained herein or therein shall be deemed to be an admission by any Party to any Third Party of any matter whatsoever (including any violation of a legal requirement or breach of contract or that a matter does, or may have a Material Adverse Effect). Matters may be disclosed on a Schedule to this Agreement for purposes of information only. Matters disclosed in each Schedule shall qualify the representation and warranty in which such Schedule is referenced and any other representation and warranty to which the matters disclosed reasonably relate.
- 13.19 Possible Exchange. Each Seller reserves the right to structure the transaction contemplated under the terms of this Agreement as a non-simultaneous like-kind exchange pursuant to Section 1031 of the Code, and its implementing regulations. In connection with effectuating a non-simultaneous like-kind exchange, such Seller reserves the right, at or prior to Closing, to assign its rights under this Agreement to a Qualified Intermediary (as that term is defined in Section 1.1031(k)-1(g)(4) of the Treasury Regulations) or to a Qualified Exchange Accommodation Titleholder (as that term is defined in Revenue Procedure 2000-37). In addition, should a Seller choose to structure the transaction provided under the terms of this Agreement as a non-simultaneous like-kind exchange, the Buyer agrees to execute all documents reasonably necessary to effectuate the non-simultaneous like-kind-exchange; provided that Buyer shall incur no additional costs, expenses, fees or liabilities as a result of or in connection with the exchange requested by a Seller. Buyer and Sellers acknowledge and agree that any assignment of rights under this Agreement to a Qualified Intermediary shall not release either Party from any of its liabilities and obligations to the other Party under this Agreement, and that neither Party represents to the other Party that any particular Tax treatment will be given to any Party as a result thereof. In no event will either Party be liable to the other for any failure of any transaction to qualify for like-kind treatment under Section 1031 of the Code.
- 13.20 <u>Confidentiality</u>. The Buyer acknowledges that, pursuant to its right of access to the Records and the Properties, as set forth in <u>Section 5.1</u>, the Buyer will become privy to confidential and other information of the Sellers and that such confidential information shall be held confidential by the

Buyer and the Buyer's representatives in accordance with the terms of the Confidentiality Agreement and this Section 13.20. To the extent not in conflict with any information sharing obligations under the Contracts or Sellers' contracts in place as of the Execution Date, from the Execution Date until the termination of the TSA and for a period of one (1) year thereafter, Sellers agree not to disclose to Third Parties geological, seismic, land (to the extent not of public record or part of a "virtual courthouse" (other than information that discloses lease expiration dates, unless already of record) of Sellers), and other proprietary data related to the Properties, unless (i) the Buyer shall have consented thereto in writing, (ii) disclosure is required pursuant to a court order or by subpoena or similar legal process, (iii) disclosure is made on advice of its counsel, pursuant to a request by a Governmental Authority, pursuant to Applicable Laws, or to comply with the rules and regulations of any stock exchange, (iv) such information has been previously made public (except as a result of a breach of this Agreement), (v) such disclosure is related to the Retained Liabilities, Excluded Assets, or indemnification obligations of Sellers under this Agreement or (vi) such disclosure is related to Sellers' exercise of a right or remedy under this Agreement; provided, however, Sellers may disclose such information to its Affiliates, representatives, agents, nominees, auditors, attorneys, consultants, banks, potential financing sources and institutional investors as needed, but in such event, the Sellers shall use reasonable efforts to cause such persons to keep such information confidential.

- 13.21 Sale or Use Tax, Recording Fees, and Similar Taxes and Fees. The Buyer shall bear any sales, use, excise, real property transfer, documentary, stamp or transfer Taxes, recording fees and similar Taxes and fees incurred and imposed upon, or with respect to, the property transfers or other transactions contemplated hereby ("Transfer Taxes"). The Sellers will determine, and the Buyer agrees to cooperate with the Sellers in determining, Transfer Taxes, if any, that applicable Law requires the Sellers to collect from the Buyer in connection with the sale of the Properties hereunder, and the Buyer agrees to pay any such Transfer Taxes to the Sellers at Closing; provided, however, that the Sellers' failure to collect any such Transfer Taxes at Closing shall not absolve the Buyer from the Buyer's responsibility for such Transfer Taxes. If such transfers or transactions are exempt from any such Taxes or fees upon the filing of an appropriate certificate or other evidence of exemption, the Buyer will timely furnish to the Sellers such certificate or evidence. Either Party will cooperate as reasonably requested by the other with respect to collecting and filing certificates or other documentation in support of filing or qualifying for an exception with respect to Transfer Taxes.
- 13.22 Expenses. Except as provided in Section 13.21, all expenses incurred by the Sellers in connection with or related to the authorization, preparation or execution of this Agreement, the conveyances delivered hereunder and the Exhibits and Schedules hereto and thereto, and all other matters related to the Closing, including all fees and expenses of counsel, accountants and financial advisers

- employed by the Sellers, shall be borne solely and entirely by the Sellers, and all such expenses incurred by the Buyer shall be borne solely and entirely by the Buyer.
- 13.23 <u>Change of Name</u>. Unless otherwise authorized by the Sellers in writing, as promptly as practicable, but in any case within ninety (90) days after the Closing Date, the Buyer shall eliminate the name **CHESAPEAKE** and any variants thereof from the Properties acquired pursuant to this Agreement and, except with respect to such grace period for eliminating existing usage, shall have no right to use any logos, trademarks or trade names belonging to the Sellers or any of their Affiliates.
- 13.24 Replacement of Bonds, Letters of Credit and Guarantees. The Parties understand that none of the bonds, letters of credit and guarantees, if any, posted by the Sellers with Governmental Authorities and relating to the Properties are to be transferred to the Buyer. On or before Closing, the Buyer shall obtain, or cause to be obtained in the name of the Buyer (or its operating Affiliate), substantially similar replacements for such bonds, letters of credit and guarantees set forth on Schedule 13.24 as necessary to permit the cancellation of such bonds, letters of credit and guarantees set forth on Schedule 13.24. The Buyer may also provide evidence that such replacements are not necessary as a result of existing bonds, letters of credit or guarantees that the Buyer has previously posted as long as such existing bonds, letters of credit or guarantees are adequate to secure the release of those set forth on Schedule 13.24.
- 13.25 <u>CONSPICUOUS</u>. THE PARTIES AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE OR ENFORCEABLE, THE PROVISIONS IN THIS AGREEMENT IN BOLD-TYPE OR ALL-CAPS FONT ARE "CONSPICUOUS" FOR THE PURPOSE OF ANY APPLICABLE LAW.
- 13.26 <u>Specific Performance</u>. Each Party hereby acknowledges and agrees that the rights of each Party hereunder are special, unique, and of extraordinary character and that, if any Party violates or fails or refuses to perform any covenant or agreement made by it herein, the non-breaching Party may be without an adequate remedy at Law. Accordingly, and without limiting the Parties' rights in <u>Article 12</u>, from and after Closing, if any Party violates or fails or refuses to perform any covenant or agreement made by such Party pursuant to <u>Sections 5.13</u>, <u>8.4</u>, <u>13.4</u>, <u>13.20</u>, <u>13.23</u>, subject to the terms hereof and in addition to any remedy at Law for damages or other relief permitted under this Agreement, the non-breaching Party may institute and prosecute any action in any court of competent jurisdiction to enforce specific performance of such covenant or agreement or seek any other equitable relief, without the necessity of proving actual damages or posting of a bond.
- 13.27 <u>Seller Liability</u>. Each Seller hereby appoints CELLC to serve as its representative hereunder (the "**Sellers' Representative**") and agrees that the Sellers' Representative shall be authorized on such Seller's behalf and responsible

hereunder, to, among other things, (a) deliver and receive all notices, statements, reports, and other information given by or addressed to Sellers hereunder, (b) receive all amounts payable to Sellers hereunder, (c) make, on a joint basis, all elections that Sellers are entitled to make hereunder, and (d) otherwise perform all of the obligations of Sellers under this Agreement. The Sellers' Representative shall be responsible for delivering to each Seller all such notices, statements, reports, and other information, and disbursing to the Sellers their respective proportionate shares of all amounts received by Sellers' Representative on their behalf. Buyer may act, and shall be fully protected in acting, in reliance upon any and all acts and things done and performed by or agreements made by Sellers' Representative with respect to the foregoing described matters on behalf of the Sellers as fully and effectively as though each had done, performed, made or executed the same. SELLERS' REPRESENTATIVE AGREES TO INDEMNIFY AND HOLD HARMLESS, AND EACH SELLER RELEASES, BUYER FROM AND AGAINST ANY AND ALL DAMAGES IN FAVOR OF ANY SELLER ARISING OUT OF, OR RESULTING IN ANY WAY FROM, THE PERFORMANCE AND DISCHARGE, OR THE FAILURE TO PERFORM OR DISCHARGE, BY SELLERS' REPRESENTATIVE OF ITS RESPONSIBILITIES UNDER THIS SECTION. FURTHER, EACH SELLER WILL BE JOINTLY AND SEVERALLY LIABLE FOR ANY SELLER'S (OR ALL OF SELLERS') OBLIGATIONS HEREUNDER. For the avoidance of doubt, in the event an individual defect or claim affects or applies to two or more of the Sellers, for purposes of determining whether the value of a defect or claim hereunder exceeds any threshold, deductible, or similar limitation herein, the value of such individual defect or claim attributable to each such Seller shall be aggregated, without duplication.

- 13.28 No Recourse. Notwithstanding anything to the contrary set forth in this Agreement, except as set forth in the CHK Parent Guaranty, no past, present or future director, officer, employee, member, partner, investor, lender, shareholder or other owner (whether direct or indirect), Affiliate, agent, attorney, financing source (including the Debt Financing Sources) or other representative of Buyer or Sellers, respectively, shall be responsible for any Liabilities of Buyer or Sellers, respectively, under this Agreement or for any Liabilities based on, in respect of, or by reason of this Agreement, any Assignment or any of the transactions contemplated hereby or thereby.
- 13.29 <u>Waiver of Claims Against Debt Financing Sources</u>. Notwithstanding anything in this Agreement to the contrary, each Seller agrees, on behalf of itself and its Affiliates, that (a) none of the Debt Financing Sources (solely in their respective capacities as Debt Financing Sources) or any former, current or future director, officer, agent, attorney, employee, Affiliate, manager, assignee, incorporator, controlling Person, fiduciary, representative, general or limited partner, stockholder, manager or member of any Debt Financing Source shall have any Liability to such Seller or its Affiliates relating to or arising out of this Agreement or the transactions contemplated by this Agreement, including the financing of the transactions contemplated by this Agreement, whether at law or equity, in

contract, in tort or otherwise, and (b) neither the Sellers nor any of their Affiliates (i) will have any rights or claims against any Debt Financing Source (solely in their respective capacities as Debt Financing Sources) or any former, current or future director, officer, agent, attorney, employee, Affiliate, manager, assignee, incorporator, controlling Person, fiduciary, representative, general or limited partner, stockholder, manager or member of any Debt Financing Source under this Agreement or any other agreement contemplated by, or entered into in connection with, the transactions contemplated by this Agreement, including any commitments by the Debt Financing Sources in respect of financing the transactions contemplated by this Agreement, (ii) seek to enforce this Agreement against any Debt Financing Source (solely in their respective capacities as Debt Financing Sources) or any former, current or future director, officer, agent, attorney, employee, Affiliate, manager, assignee, incorporator, controlling Person, fiduciary, representative, general or limited partner, stockholder, manager or member of any Debt Financing Source or (iii) bring any claim or cause of action against any Debt Financing Source (solely in their respective capacities as Debt Financing Sources) or any former, current or future director, officer, agent, attorney, employee, Affiliate, manager, assignee, incorporator, controlling Person, fiduciary, representative, general or limited partner, stockholder, manager or member of any Debt Financing Source under this Agreement or any other agreement contemplated by, or entered into in connection with, the transactions contemplated by this Agreement, including any commitments by the Debt Financing Sources in respect of financing the transactions contemplated by this Agreement.

## [REMAINDER OF PAGE INTENTIONALLY LEFT BLANK SIGNATURE PAGES FOLLOW]

-105-

#### **SELLERS:**

## CHESAPEAKE EXPLORATION, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## CHESAPEAKE APPALACHIA, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## CHK UTICA L.L.C., a Delaware limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## MC MINERAL COMPANY, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT BY AND AMONG CHESAPEAKE EXPLORATION, L.L.C., CHESAPEAKE APPALACHIA, L.L.C., CHK UTICA, L.L.C., MC MINERAL COMPANY, L.L.C., CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C., CHESAPEAKE ENERGY MARKETING, L.L.C., CHESAPEAKE ROYALTY, L.L.C., MIDCON COMPRESSION, L.L.C. AND EAP OHIO, LLC

# CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## CHESAPEAKE ENERGY MARKETING, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## CHESAPEAKE ROYALTY, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

## MIDCON COMPRESSION, L.L.C., an Oklahoma limited liability company

By: /s/ Bryan J. Lemmerman

Name: Bryan J. Lemmerman

Vice President — Business Development

SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT BY AND AMONG CHESAPEAKE EXPLORATION, L.L.C., CHESAPEAKE APPALACHIA, L.L.C., CHK UTICA, L.L.C., MC MINERAL COMPANY, L.L.C., CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C., CHESAPEAKE ENERGY MARKETING, L.L.C., CHESAPEAKE ROYALTY, L.L.C., MIDCON COMPRESSION, L.L.C. AND EAP OHIO, LLC

IN WITNESS WHEREOF, the Buyer has executed this Agreement effective as of the Execution Date.

## **BUYER:**

## **EAP Ohio, LLC**

By: Encino Acquisition Partners, LLC, as sole member

By: Encino Energy, LLC, Manager

By: /s/ Michael Magilton

Name: Michael Magilton Chief Financial Officer

SIGNATURE PAGE TO PURCHASE AND SALE AGREEMENT BY AND AMONG CHESAPEAKE EXPLORATION, L.L.C., CHESAPEAKE APPALACHIA, L.L.C., CHK UTICA, L.L.C., MC MINERAL COMPANY, L.L.C., CHESAPEAKE LAND DEVELOPMENT COMPANY, L.L.C., CHESAPEAKE ENERGY MARKETING, L.L.C., CHESAPEAKE ROYALTY, L.L.C., MIDCON COMPRESSION, L.L.C. AND EAP OHIO, LLC

#### **CERTIFICATION**

#### I, Robert D. Lawler, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q of Chesapeake Energy Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - (b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - (c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation: and
  - (d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - (a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

October 30, 2018 By: <u>/s/ ROBERT D. LAWLER</u>

Robert D. Lawler

President and Chief Executive Officer

#### CERTIFICATION

- I, Domenic J. Dell'Osso, Jr., certify that:
  - 1. I have reviewed this Quarterly Report on Form 10-Q of Chesapeake Energy Corporation;
  - 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
  - 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
  - 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
    - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
    - (b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
    - (c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
    - (d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
  - 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
    - (a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
    - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

October 30, 2018 By:

/s/ DOMENIC J. DELL'OSSO, JR.

Domenic J. Dell'Osso, Jr.

Executive Vice President and Chief Financial Officer

### CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Chesapeake Energy Corporation (the "Company") on Form 10-Q for the period ended September 30, 2018 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Robert D. Lawler, President and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

October 30, 2018

By: <u>/s/ ROBERT D. LAWLER</u>

Robert D. Lawler

President and Chief Executive Officer

### CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of Chesapeake Energy Corporation (the "Company") on Form 10-Q for the quarter ended September 30, 2018 as filed with the Securities and Exchange Commission on the date hereof (the "Report"), I, Domenic J. Dell'Osso, Jr., Executive Vice President and Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. § 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002, that:

- 1. The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- 2. The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

October 30, 2018

By: <u>/s/ DOMENIC J. DELL'OSSO, JR.</u>

Domenic J. Dell'Osso, Jr. Executive Vice President and Chief Financial Officer